

## OPERATOR ALGEBRAS AND ABSTRACT CLASSIFICATION

## MARTINO LUPINI

### A DISSERTATION SUBMITTED TO THE FACULTY OF GRADUATE STUDIES IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

### GRADUATE PROGRAM IN MATHEMATICS AND STATISTICS YORK UNIVERSITY TORONTO, ONTARIO

March 2015

©Martino Lupini, 2015

# Abstract

This dissertation is dedicated to the study of operator spaces, operator algebras, and their automorphisms using methods from logic, particularly descriptive set theory and model theory. The material is divided into three main themes. The first one concerns the notion of Polish groupoids and functorial complexity. Such a study is motivated by the fact that the categories of Elliott-classifiable algebras, Elliott invariants, abelian separable C\*-algebras, and arbitrary separable C\*-algebras have the same complexity according to the usual notion of Borel complexity. The goal is to provide a functorial refinement of Borel complexity, able to capture the complexity of classifying the objects in a functorial way. Our main result is that functorial Borel complexity provides a finer distinction of the complexity of functorial classification problems.

The second main theme concerns the classification problem for automorphisms of C<sup>\*</sup>algebras from the perspective of Borel complexity theory. Our results show that, for any non-elementary simple separable C<sup>\*</sup>-algebra, the problem of classifying its automorphisms up to unitary equivalence transcends countable structures. Furthermore we prove that in the unital case the relation of unitary equivalence obeys the following dichotomy: it is either smooth, when the algebra has continuous trace, or not classifiable by countable structures.

The last theme concerns applications of model theory to the study and construction of interesting operator spaces and operator systems. Specifically we show that the Gurarij operator space introduced by Oikhberg can be characterized as the Fraïssé limit of the class of finite-dimensional 1-exact operator spaces. This proves that the Gurarij operator space is unique, homogeneous, and universal among separable 1-exact operator spaces. Moreover we prove that, while being 1-exact, the Gurarij operator space does not embed into any exact C\*-algebra. Furthermore the ternary ring of operators generated by the Gurarij operator space is canonical, and does not depend on the concrete representation chosen. We also construct the operator system analog of the Gurarij operator space, and prove that it has analogous properties.

# Contents

	Abstract						
	Table of contents						
	$\operatorname{List}$	of Fig	gures	vi			
	Intr	oducti	on	1			
1	Poli	sh gro	upoids and functorial Borel complexity	6			
	1.1	Descri	ptive set theory and Polish groupoids	9			
		1.1.1	Descriptive set theory	9			
		1.1.2	Locally Polish spaces	11			
		1.1.3	The Effros fibred space	12			
		1.1.4	Analytic and Borel groupoids	12			
		1.1.5	Polish groupoids and Polish groupoid actions	14			
		1.1.6	Some examples of Borel groupoids	16			
		1.1.7	The action groupoid	18			
		1.1.8	Functorial reducibility	19			
		1.1.9	Category preserving maps	20			
		1.1.10	Vaught transforms	20			
		1.1.11	Borel orbits	26			
	1.2	Effros'	theorem and the Glimm-Effros dichotomy	27			
		1.2.1	Effros' theorem	27			
		1.2.2	A Polish topology on quotient spaces	29			
		1.2.3	$G_{\delta}$ orbits	30			
		1.2.4	The Glimm-Effros dichotomy	32			
	1.3	Better	topologies	34			

		1.3.1	Polishability of Borel $G$ -spaces $\ldots \ldots \ldots \ldots \ldots \ldots \ldots 3$
		1.3.2	Finer topologies for Polish $G$ -spaces $\ldots \ldots \ldots \ldots \ldots \ldots 3$
1	.4	Borel	orbit equivalence relations 3
		1.4.1	A Borel selector for cosets
		1.4.2	Borel orbit equivalence relations 4
1	.5	Univer	rsal actions
1	.6	Count	able Borel groupoids    4
		1.6.1	Actions of inverse semigroups on Polish spaces
		1.6.2	Étale Polish groupoids
		1.6.3	The groupoid of germs
		1.6.4	Regularity of the groupoid of germs 4
		1.6.5	Étale groupoids as groupoids of germs
		1.6.6	Borel bisections
		1.6.7	A Polish topology on countable Borel groupoids 5
		1.6.8	Treeable Borel groupoids 5
1	.7	Functo	brial Borel complexity and treeable equivalence relations
		1.7.1	The lifting property 5
		1.7.2	The cocycle property
		1.7.3	Free actions of treeable groupoids
		1.7.4	Characterizing treeable equivalence relations
τ	J <b>ni</b> t	tary e	quivalence of automorphisms of C*-algebras 6
2	.1	Some	background notions on C*-algebras $\ldots \ldots \ldots$
2	.2	Noncla	assifiability criteria
2	.3	The ca	ase of algebras with outer derivations
2	.4	The ca	ase of algebras with only inner derivations
2	.5	A dich	notomy for derivations
Г	ſhe	Gura	rij operator space 9
3	.1	Backg	round material
		3.1.1	Approximate isometries
		3.1.2	Languages and structures
		3.1.3	Fraïssé classes and limits
		3.1.4	Operator spaces
		3.1.5	$M_n$ -spaces

		3.2.1	Amalgamation property	106
		3.2.2	The Fraïssé metric space	107
		3.2.3	The Fraïssé limit	111
	3.3	The ne	oncommutative Gurarij space	113
		3.3.1	MIN and MAX spaces	113
		3.3.2	Exact and 1-exact operator spaces	114
		3.3.3	Amalgamation of 1-exact operator spaces	115
		3.3.4	The Fraïssé metric space	119
		3.3.5	The noncommutative Gurarij space as a Fraïssé limit	121
4	The	Gura	rij operator system	124
	4.1	Prelim	inary notions	125
		4.1.1	Operator systems	125
		4.1.2	Exactness	127
		4.1.3	Pushouts of operator systems	128
		4.1.4	$M_n$ -systems	130
	4.2	Finite-	-dimensional 1-exact operator systems	132
		4.2.1	Amalgamation of 1-exact operator systems	132
		4.2.2	Embeddings of 1-exact operator systems	137
		4.2.3	The Fraïssé metric space	138
	4.3	The G	urarij operator system	140
		4.3.1	The limit of the class $\mathcal{E}_1^{sy}$	140
		4.3.2	Existence and uniqueness of the Gurarij operator system $\ldots$ .	142
		4.3.3	An explicit construction of $\mathbb{GS}$	143
		4.3.4	Unital operator spaces	144
		4.3.5	A unital function system of almost universal disposition $\ldots$ .	146
		4.3.6	The C*-envelope of the Gurarij operator system $\ldots \ldots \ldots \ldots$	147
		4.3.7	The triple envelope of the Gurarij operator space	149

# List of Figures

1	66
---	----

# Introduction

One of the main revolution in physics in the 20th Century is with no doubt the introduction of quantum mechanics, aiming at explaining the physical phenomena at extremely small scale. In this framework the classical laws of physics are replaced by their quantized version, obtained—roughly speaking—by replacing functions with operators on the Hilbert space. The study of algebras of operators on the Hilbert space was initiated by Murray and von Neumann in the 1930s, aiming at providing rigorous mathematical foundation to quantum mechanics. Murray and von Neumann focused on the case of algebras that are closed in the weak operator topology (von Neumann algebras, in the modern terminology), which are the quantized analog of measure spaces. The topological side of this study was initiated by the abstract characterization due to Gelfand and Neumark of algebras of operators that are closed in the norm topology [48]. These can be regarded as the quantum analog of compact Hausdorff spaces. The fact that quantized spaces have no actual points correspond to the physical principle that it is meaningless to speak about points in the phase space of a quantum particle. Since the seminal results of Murray-von Neumann and Gelfand-Neumark, the study of operator algebras has expanded enormously, finding many new applications and connections to a variety of branches of mathematics and physics.

In the last decade many long-standing problems in operator algebras have been settled using methods originally developed within the field of mathematical logic. These problems include the existence of outer automorphisms of the Calkin algebra [32] or the existence of nonisomorphic ultrapowers or relative commutants of a given II<sub>1</sub> factors [35]. These breakthroughs have been the starting point for a new line of research in operator algebras that uses tools and ideas coming from logic and set theory. Generally speaking, such a line of research can be divided into three main areas: set theory and forcing, descriptive set theory and Borel complexity, and model theory and continuous logic.

Descriptive set theory is, broadly speaking, the study of definable subsets of Polish spaces. A generous notion of definability for subsets of Polish spaces and maps between them is being Borel-measurable. The idea of studying abstractly the existence of classifying maps that are definable in this sense was first investigated in [43]. Such a study led to what is now called Borel complexity theory, providing a general framework where the complexity of classification problems in mathematics can be measured and compared. Many strong tools have been developed, most notably Hjorth's theory of turbulence [59], to distinguish between the complexity of classification problems, and refute the possibility of a satisfactory classification by means of a certain kind of invariants. This is especially valuable for fields, such as the theory of C\*-algebras, were classification plays a prominent role.

Originating from the seminal classification results of UHF and AF C\*-algebras due to Glimm [50] and Elliott-Bratteli [27, 16], the Elliott classification program is an ambitious project aiming at a full classification of all simple, separable, nuclear, unital C<sup>\*</sup>-algebras by their Elliott invariant. This is a functorial invariant consisting of K-theoretical information (K-groups), measure-theoretical information (trace simplex), and their interaction (canonical pairing between  $K_0$  and traces). The possible scope of such a program has been recently limited by counterexamples due to Rørdam and Toms [122, 131, 132]. These examples have shown that additional regularity properties are needed to ensure Elliott-classifiability. Three such properties ( $\mathcal{Z}$ -absorption, finite nuclear dimension, and strict comparison) of very different nature (algebraic, topological, and cohomological) have been recently shown to be equivalent for a large class of simple, separable, nuclear, unital C<sup>\*</sup>-algebras, partially confirming a conjecture due to Toms and Winter; see [15] and references therein. The classification program can then be recast for the class of *well behaved* (in any of the conjecturally equivalent acceptations above) simple, separable, unital, nuclear C<sup>\*</sup>-algebras. The study of the Elliott classification program from the perspective of Borel complexity theory has been initiated in [38]. As observed in  $[34, \S3]$ , by combining results from [30, 99, 45, 125, 137]one can conclude that the following classes of objects have the same complexity:

- separable C\*-algebras,
- well-behaved simple, unital, nuclear, separable C\*-algebras,
- abelian C\*-algebras,
- Elliott invariants.

However the functorial nature of the classification of C\*-algebras is not taken into account in the analysis above. This observation motivated us to introduce the notion of functorial Borel complexity, aiming at capturing the complexity of classifying the object of a category in an explicit and functorial way. Such a notion can be seen as a refinement and generalization of the usual notion of Borel complexity, since it reduces to the latter one when the morphisms are ignored, and only the isomorphism relations are considered. For simplicity we restricted ourselves to the case of groupoids, i.e. categories where every arrow is invertible. Furthermore we considered groupoids endowed with a canonical Polish topology, making composition and inversion of arrows continuous (Polish groupoids). It turns out that most of the fundamental results about the descriptive set theory of Polish group actions can be generalized to the case of Polish groupoids, as we show in Chapter 1; see also [89]. As an application we prove that the notion of functorial Borel complexity provides a finer distinction between functorial classification problems than the usual notion of Borel complexity. Precisely, for any nontreeable countable Borel equivalence E relation there are Polish groupoids with G and H both having E as associated orbit equivalence relation, such that G and H have distinct functorial Borel complexity. Such a property in fact characterizes the nontreeable Borel equivalence relations among all the countable Borel equivalence relations: for groupoids with (essentially) treeable orbit equivalence relation Borel complexity and functorial Borel complexity coincide.

In Chapter 2 we apply the theory of Borel complexity, and primarily Hjorth's theory of turbulence, to the study of automorphisms of C\*-algebras. A prominent role in the theory of C\*-algebras, as a source of both examples and applications, is played by the theory of C\*-dynamical systems and crossed products. In its simplest form, a C\*-dynamical system is a pair  $(A, \alpha)$  where A is a C\*-algebra and  $\alpha$  is an automorphism of A, i.e. a function  $\alpha : A \to A$  preserving all the C\*-algebra structure. This can be regarded as a quantized analog of a dynamical system, which is a homeomorphic transformation of a compact Hausdorff space. Two automorphisms  $\alpha, \alpha'$  of A are conjugate if  $\alpha'$  equals  $\beta \circ \alpha \circ \beta^{-1}$  for some automorphism  $\beta$  of A. This can be seen as a quantum analog of the usual notion of conjugacy of topological dynamical systems. In joint work with David Kerr and N. Christopher Phillips we have shown that the classification problem for automorphisms up to conjugacy is intractable for any Elliott-classifiable C\*-algebra [76]. The main technical tool used in the proof is Hjorth's theory of turbulence [59], providing dynamical conditions on a Polish group action ensuring that the corresponding orbit equivalence relation is not classifiable with countable structures as invariants.

The above result suggests that other quantum analogs of conjugacy should be considered to obtain a satisfactory classification in the noncommutative case. Any unitary element uof a C\*-algebra A induces an automorphisms of A defined by  $x \mapsto uxu^*$ . Automorphisms of this form—called inner—can be thought as "rotations" of the space on which A acts, and should be regarded as trivial. It is therefore natural, in classifying the automorphisms of A, to identify automorphisms that are unitarily equivalent, i.e. agree modulo an inner automorphism.

Our main result on the subject, presented in Chapter 2 and also contained in [91], shows that the classification problem of automorphisms up to unitary equivalence is intractable for any simple separable C\*-algebra. Furthermore, we prove that the relation of unitary equivalence of automorphisms of separable unital C\*-algebra obeys the following dichotomy: it is either smooth (when the algebra has continuous trace), or not classifiable by countable structures. Such a result is reminiscent of the analogous behaviour of the relation of unitary equivalence of representations of locally compact groups and C\*-algebras [58, 33, 75].

Combining conjugacy and unitarily equivalence one obtains the notion of cocycle conjugacy. Precisely, two automorphisms are cocycle conjugate if one is unitarily equivalent to a conjugate of the other. Satisfactory classification results of automorphisms of  $\mathcal{O}_2$  and other C\*-algebras up to cocycle conjugacy have been obtained in [61, 62] under suitable freeness conditions. Little is known in general about the complexity of the relation of cocycle conjugacy. In joint work with Eusebio Gardella I have shown that, in the case of the Cuntz algebra  $\mathcal{O}_2$ , the relation of cocycle conjugacy is a complete analytic set [47].

The last topic that we consider in this thesis concerns applications of model theory to operator spaces and operator systems. The theory of operator spaces can be considered as a noncommutative generalization of Banach space theory. It provides the natural framework where important notions for C\*-algebras such as nuclearity and exactness can be defined and studied. A noncommutative analog of the Gurarij Banach space has been introduced by Oikhberg in [104]. In Chapter 3 we prove that such a space can be realized as the Fraïssé limit of the class of finite-dimensional 1-exact operator spaces. As a consequence we deduce that the Gurarij operator space NG is unique, homogeneous, and universal for separable 1-exact operator spaces. These results are also presented in [90]. Moreover we prove that the ternary ring of operators  $\mathcal{T}(NG)$  generated by NG does not depend on the concrete realization of NG as a subspace of B(H), and is not exact. This implies that NG does not embed into any exact C\*-algebra.

In Chapter 4 we consider the analogous construction in the operator systems category, yielding the Gurarij operator system  $\mathbb{GS}$ ; see also [92]. The Gurarij operator system is unique, homogeneous, and universal for separable 1-exact operator systems. Furthermore the C\*-algebra  $C^*(\mathbb{GS})$  generated by  $\mathbb{GS}$  does not depend from the concrete realization of  $\mathbb{GS}$  as a unital subspace of B(H), and it is not exact. This implies that  $\mathbb{GS}$  does not unitally embed into any exact unital C\*-algebra.

In joint work with Isaac Goldbring [51] we have shown that the first order theory of  $\mathbb{NG}$  is the model-completion of the theory of operator spaces, it has a unique separable 1-exact model, and any two separable models of  $\mathbb{NG}$  are *n*-isometric for every  $n \in \mathbb{N}$ . However  $\mathbb{NG}$  is not separably categorical, and in fact it has a continuum of pairwise non completely isometric models, and no atomic model. The analogous properties for the Gurarij operator systems are also established.

### Chapter 1

# Polish groupoids and functorial Borel complexity

Classification of mathematical structures is one of the main components of modern mathematics. It is safe to say that most results in mathematics can be described as providing an explicit classification of a class of mathematical objects by a certain type of invariants.

In the last 25 years the notion of constructive classification has been given a rigorous formulation in the framework of invariant complexity theory. In this context a classification problem is regarded as an equivalence relation on a standard Borel space (virtually all classification problems in mathematics fit into this category). The concept of constructive classification is formalized by the notion of *Borel reduction*. A Borel reduction from an equivalence relation E on X to an equivalence relation E' on X' is a *Borel* function f:  $X \to X'$  with the property that, for every  $x, y \in X$ ,

xEy if and only if f(x)E'f(y).

In other words f is a Borel assignment of complete invariants for E that are equivalence classes of E'. The existence of such a function can be interpreted as saying that classifying the objects of X' up to E' is at least as complicated as classifying the objects of X up to E. This offers a notion of comparison between the complexity of different classification problems.

Several natural equivalence relation can then be used as benchmarks to measure the complexity of classification problems. Perhaps the most obvious such a benchmark is the relation of equality  $=_{\mathbb{R}}$  of real numbers. This gives origin to the basic dichotomy smooth vs. non-smooth: an equivalence relation is *smooth* if it is Borel reducible to  $=_{\mathbb{R}}$ . (The real

numbers can here be replaced by any other uncountable standard Borel space.) Beyond smoothness the next fundamental benchmark is classifiability by countable structures. Here the test is Borel reducibility to the relation of isomorphism within some class of countable first order structures, such as (ordered) groups, rings, etc. Equivalently one can consider orbit equivalence relation associated with Borel actions of the Polish group  $S_{\infty}$  of permutations of  $\mathbb{N}$ . Replacing  $S_{\infty}$  with an arbitrary Polish group yields the notion of equivalence relation classifiable by orbits of a Polish group action.

This framework allows one to build a hierarchy between different classification problems. Many efforts have been dedicated to the attempt to draw a picture as complete as possible of classification problems in mathematics and their relative complexity. To this purpose powerful tools such as Hjorth's theory of turbulence [59] have been developed in order to disprove the existence of Borel reduction between given equivalence relations, and to distinguish between the complexity of different classification problems. This can be interpreted as a way to formally exclude the possibility of a full classification of a certain class of objects by means of a given type of invariants. For example the relation of isomorphism of simple separable C<sup>\*</sup>-algebras has been shown to transcend countable structures in [38]; see also [125]. Similar results have been obtained for several other equivalence relations, such as affine homeomorphism of Choquet simplexes [38], conjugacy of unitary operators on the infinite dimensional separable Hilbert space [74], conjugacy of ergodic measure-preserving transformations of the Lebesgue space [41], conjugacy of homeomorphisms of the unit square [59], conjugacy of irreducible representations of non type I groups [58] or C\*-algebras [33, 75], conjugacy and unitary equivalence of automorphisms of classifiable simple separable C<sup>\*</sup>algebras [76, 91], isometry of separable Banach spaces [99] and complete order isomorphism of separable operator systems. Furthermore the relations of isomorphism and Lipschitz isomorphisms of separable Banach spaces, topological isomorphism of (abelian) Polish groups, uniform homeomorphism of complete separable metric spaces [40], and the relation of completely bounded isomorphism of separable operator spaces [4] have been shown to be not classifiable by the orbits of a Polish group action (and in fact to have maximal complexity among analytic equivalence relations). An exhaustive introduction to invariant complexity theory can be found in [44].

Considering how helpful the theory of Borel complexity has been so far in giving us a clear understanding of the relative complexity of classification problems in mathematics, it seems natural to look at refinements to the notion of Borel reducibility, that can in some situations better capture the notion of explicit classification from the practice of mathematics. Such a line of research has been suggested in [34], where the results of the present chapter

has been announced. This is the case for example when the classification problem under consideration concerns a category. In this case it is natural to ask to the classifying map to be *functorial*, and to assign invariants not only to the objects of the category, but also to the *morphisms*. This is precisely what happens in many explicit examples of classification results in mathematics. In fact in many such examples the consideration of invariants of morphisms is essential to the proof. This is particularly the case in the Elliott classification program of simple C\*-algebras, starting from Elliott's seminal paper of AF algebras [27]. Motivated by similar considerations, Elliott has suggested in [29] an abstract approach to classification by functors. In this chapter we bring Elliott's theory of functorial classification within the framework of Borel complexity theory. For simplicity we consider only categories where every arrow is invertible, called groupoids. Such categories will be assumed to have a global Borel structure that is at least analytic, and makes the set of objects (identified with their identity arrows) a standard Borel space. In the particular case when between any two objects there is at most one arrow (principal groupoids) these are precisely the analytic equivalence relations. One can then consider the natural constructibility requirement for classifying functors, which is being Borel with respect to the given Borel structures. This gives rise to the notion of functorial Borel complexity, which in the particular case of principal groupoids is the usual notion of Borel complexity.

In this chapter we study such a notion of functorial Borel complexity for groupoids, focusing on the case of Polish groupoids. These are the groupoids where the Borel structure is induced by a topology that makes composition and inversion of arrows continuous and open, and has a basis of open sets which are Polish in the relative topology. These include all Polish groups, groupoids associated with Polish group actions, and locally compact groupoids [106, Definition 2.2.2]. The latter ones include the holonomy groupoids of foliations and the tangent groupoids of manifolds [106, Chapter 2], the groupoids of row-finite directed graphs [84], the localization groupoids of actions of countable inverse semigroups [106, Chapter 4]. The main results of the present chapter assert that, for Polish groupoids with essentially countable equivalence relation, the existence of a Borel reducibility between the groupoids is equivalent to the Borel reducibility of the corresponding orbit equivalence relations. On the other hand for every countable equivalence relation E that is not treeable there are two Polish groupoid with orbit equivalence relation E that have distinct functorial Borel complexity; see Section 1.7. This shows that Borel reducibility of groupoids provides a finer notion of complexity than the usual Borel reducibility of equivalence relations. Having a finer notion of complexity is valuable, because it allows one to further distinguish between the complexity of problems that, in the usual framework, turn out to have the same complexity. An example of this phenomenon occurs in the classification problem for C\*-algebras, where it turns out [38, 30, 125] that classifying arbitrary separable C\*-algebras is as difficult as classifying the restricted class of C\*-algebras that are considered to be well behaved (precisely the amenable simple C\*-algebras, or even more restrictively the simple C\*-algebras that can be obtained as direct limits of interval algebras).

In order to prove the above mentioned characterization of essentially treeable equivalence relations we will generalize some fundamental results of the theory of actions of Polish groups to actions of Polish groupoids, answering a question of Ramsay from [120]. These include the Becker-Kechris results on Polishability of Borel G-spaces [7, Chapter 5], existence of universal Borel G-spaces [7, Section 2.6], and characterization of Borel G-spaces with Borel orbit equivalence relation [7, Chapter 7]. The fundamental technique employed is a generalization of the Vaught transform [134] from actions of Polish groups to actions of Polish groupoids.

This chapter is organized as follows: In Section 1.1 we recall some background notions, introduce the notation to be used in the rest, and state the basic properties of the Vaught transform for actions of Polish groupoids. In Section 1.2 we generalize the local version of Effros' theorem from Polish group actions to actions of Polish groupoids, and infer the Glimm-Effros dichotomy for Polish groupoids and Borel reducibility, refining results from [119]. Section 1.3 contains the proof of the Polishability result for Borel G-spaces, showing that any Borel G-space is isomorphic to a Polish G-space, where G is a Polish groupoid. A characterization for Borel G-spaces with Borel orbit equivalence relation is obtained as a consequence in Section 1.4. Section 1.5 contains the construction of a universal Borel G-space for a given Polish groupoid G, generalizing [7, Section 2.6]. Section 1.6 considers countable Borel groupoids, i.e. analytic groupoids with only countably many arrows with a given source. It is shown that every such a groupoid has a Polish groupoid structure compatible with its Borel structure. In particular all results about Polish groupoids apply to countable Borel groupoids. Finally in Section 1.7 the above mentioned characterization of essentially treeable equivalence relations in terms of Borel reducibility is proved.

#### 1.1 Descriptive set theory and Polish groupoids

#### 1.1.1 Descriptive set theory

A *Polish space* is a separable and completely metrizable topological space. Equivalently a topological space is Polish if it is  $T_1$ , regular, second countable, and *strong Choquet* [73,

Theorem 8.18]. A subspace of a Polish space is Polish with respect to the subspace topology if and only if it is a  $G_{\delta}$  [73, Theorem 3.11].

A standard Borel space is a space endowed with a  $\sigma$ -algebra which is the  $\sigma$ -algebra of Borel sets with respect to some Polish topology. An *analytic space* is a space endowed with a countably generated  $\sigma$ -algebra which is the image of a standard Borel space under a Borel function. A subset of a standard Borel space is analytic if it is an analytic space with the relative standard Borel structure. A subset of a standard Borel space is *co-analytic* if its complement is analytic. It is well known that for a subset of a standard Borel space it is equivalent being Borel and being both analytic and co-analytic [73, Theorem 14.7]. If X, Yare standard Borel space and A is a subset of  $X \times Y$ , then for  $x \in X$  the section

$$\{y \in Y : (x, y) \in A\}$$

is denoted by  $A_x$ . The projection of A onto the first coordinate is

$$\{x \in X : A_x \neq \emptyset\},\$$

while the *co-projection* of A is

$$\{x \in X : A_x = Y\}.$$

The projection of an analytic set is analytic, while the co-projection of a co-analytic set is co-analytic.

If X is a Polish space, then the space of closed subsets of X is denoted by F(X). The *Effros Borel structure* on F(X) is the  $\sigma$ -algebra generated by the sets

$$\{F \in F(X) : F \cap U \neq \varnothing\}$$

for  $U \subset X$  open. This makes F(X) a standard Borel space [73, Section 12.C].

Recall that a subset A of a Polish space X has the *Baire property* if there is an open subset U of X such that the symmetric difference  $A \triangle U$  is meager [73, Definition 8.21]. It follows from [73, Corollary 29.14] that any analytic subset of X has the Baire property.

A topological space X is a *Baire space* if every nonempty open subset of X is not meager. Every completely metrizable topological space is a Baire space; see [73, Theorem 8.4].

If X, Y are standard Borel spaces, then we say that Y is *fibred* over X if there is a Borel surjection  $p: Y \to X$ . If  $x \in X$ , then the inverse image of x under p is called the x-fiber of

Y and denoted by  $Y_x$ . If  $Y_0, Y_1$  are fibred over X, then the fibred product

$$Y_0 * Y_1 = \{(y_0, y_1) : p_0(y_0) = p_1(y_1)\}$$

is naturally fibred over X. Similarly if  $(Y_n)_{n \in \mathbb{N}}$  is a sequence of Borel spaces fibred over X we define

$$\bigstar_{n\in\mathbb{N}} Y_n = \{(y_n)_{n\in\mathbb{N}} : p(y_n) = p(y_m) \text{ for } n, m \in \mathbb{N}\}$$

which is again fibred over X. A Borel fibred map from  $Y_0$  to  $Y_1$  is a Borel function  $\varphi : Y_0 \to Y_1$  which sends fibers to fibers, i.e.  $p_1 \circ \varphi = p_0$ .

If E is an equivalence relation on a standard Borel space X, then a subset T of X is a *transversal* for E if it intersects any class of E in exactly one point. A *selector* for E is a Borel function  $\sigma : X \to X$  such that  $\sigma(x)Ex$  for every  $x \in X$  and  $\sigma(x) = \sigma(y)$  whenever xEy.

#### 1.1.2 Locally Polish spaces

**Definition 1.1.1.** A *locally Polish space* is a topological space with a countable basis of open sets which are Polish spaces in the relative topology.

By [73, Theorem 8.18] a locally Polish space is  $T_1$ , second countable, and strong Choquet. Moreover it is a Polish space if and only if it is regular. It follows from [73, Lemma 3.11] that a  $G_{\delta}$  subspace of a locally Polish space is locally Polish.

Suppose that X is a locally Polish space. Denote by F(X) the space of closed subsets of X. The Effros Borel structure on F(X) is the  $\sigma$ -algebra generated by the sets of the form

$$\{F: F \cap U \neq \emptyset\}$$

for  $U \subset X$  open. It has been proved by Anush Tserunyan that the Effros Borel structure on F(X) is standard.

One can deduce from this that the Borel  $\sigma$ -algebra of X is standard. In fact the function

$$\begin{array}{rccc} X & \to & F(X) \\ x & \mapsto & \{x\} \end{array}$$

is clearly a Borel isomorphism onto the set  $F_1(X)$  of closed subsets of F(X) containing exactly one element. It is therefore enough to show that  $F_1(X)$  is a Borel subset of F(X). Fix a countable basis  $\mathcal{A}$  of open Polish subsets of X. Suppose also that for every  $U \in \mathcal{A}$  it is fixed a compatible complete metric  $d_U$  on U. Observe that  $F_1(X)$  contains precisely the closed subsets F of X such that  $F \cap U \neq \emptyset$  for some  $U \in \mathcal{A}$  and for every  $U \in \mathcal{A}$  such that  $F \cap U \neq \emptyset$  and for every  $n \in \mathbb{N}$  there is  $W \in \mathcal{A}$  such that  $cl(W) \subset U$ ,  $diam_U(cl(\overline{W})) < 2^{-n}$ and  $F \cap (X \setminus cl(W)) = \emptyset$ , where  $diam_U(cl(\overline{W}))$  is the diameter of  $\overline{W}$  with respect to the metric  $d_U$ . This shows that  $F_1(X)$  is a Borel subset of X.

#### 1.1.3 The Effros fibred space

Suppose that Z is a locally Polish space, X is a Polish space, and  $p: Z \to X$  is a continuous open surjection. For  $x \in X$  denote by  $Z_x$  the inverse image of x under p. Define  $F^*(Z)$ to be the space of *nonempty* subsets of Z endowed with the Effros Borel structure. Define  $F^*(Z, X)$  to be the Borel subset of closed subsets of Z contained in  $Z_x$  for some  $x \in X$ . The Borel function from  $F^*(Z, X)$  onto X assigning to an element F of  $F^*(Z, X)$  the unique  $x \in X$  such that  $F \subset Z_x$  endows  $F^*(Z, X)$  with the structure of fibred Borel space. The obvious embedding of  $F^*(Z_x)$  into  $F^*(Z, X)$  is a Borel isomorphism onto the x-fiber of  $F^*(Z, X)$ .

Consider for every  $x \in X$  a copy  $\emptyset_x$  of the empty set. Define F(Z, X) to be the disjoint union of  $F^*(Z, X)$  with  $\{\emptyset_x : x \in X\}$ , which is again a standard Borel space fibred over Xin the obvious way. Moreover the x-fiber of F(Z, X) is now naturally isomorphic to the space F(Z) of (possibly empty) subsets of  $Z_x$ . We will call F(Z, X) the (standard) *Effros* fibred space of the fibration  $p: Z \to X$ .

#### 1.1.4 Analytic and Borel groupoids

A groupoid G is a small category where every arrow is invertible. The set of objects of G is denoted by  $G^0$ . We will regard  $G^0$  as a subset of G, by identifying an object with its identity arrow. Denote by  $G^2$  the (closed) set of pairs of composable arrows

$$G^{2} = \{(\gamma, \rho) : s(\gamma) = r(\rho)\}.$$

If A, B are subsets of G, then AB stands for the set

$$\{\gamma \rho : (\gamma, \rho) \in (A \times B) \cap G^2\}.$$

In particular if  $Y \subset X$  then

$$YB = \{\gamma \in B : r(\gamma) \in Y\}$$

and

$$BY = \{ \gamma \in B : s(\gamma) \in Y \} \,.$$

We write xB for  $\{x\}B = r^{-1}[\{x\}] \cap B$  and Bx for  $B\{x\} = s^{-1}[\{x\}] \cap B$ . If A is a set of objects, then the *restriction*  $G_{|A}$  of G to A (this is called "contraction" in [95, 118]) is the groupoid

$$\{\gamma \in G : s(\gamma) \in A, r(\gamma) \in A\}$$

with set of objects A and operations inherited from G.

To every groupoid G one can associate the orbit equivalence relation  $E_G$  on  $G^0$  defined by  $(x, y) \in E_G$  if and only if there is  $\gamma \in G$  such that  $s(\gamma) = x$  and  $r(\gamma) = y$ . The function

$$\begin{array}{rccc} G & \to & E_G \\ \gamma & \mapsto & (r(\gamma), s(\gamma)) \end{array}$$

is a continuous surjection. We say that a groupoid is *principal* when such a map is injective. Thus a principal groupoid is just an equivalence relation on its set of objects. Conversely any equivalence relation can be regarded as a principal groupoid.

The notion of *functor* between groupoids is the usual notion from category theory. Thus a functor from G to H is a function from G to H such that, for every  $\gamma \in G$  and  $(\rho_0, \rho_1) \in G^2$ the following holds:

- $F(s(\gamma)) = s(F(\gamma));$
- $F(r(\gamma)) = r(F(\gamma));$
- $F(\gamma^{-1}) = F(\gamma)^{-1};$
- $F(\rho_0 \rho_1) = F(\rho_0)F(\rho_1).$

When E and E' are principal groupoids, then functors from E to E' are in 1:1 correspondence with *homomorphisms* from E to E' in the sense of [44, Definition 10.1.3].

**Definition 1.1.2.** An analytic groupoid is a groupoid endowed with an analytic Borel structure making composition and inversion of arrows Borel and such that the set of objects and, for every object x, the set of elements with source x, are standard Borel spaces with respect to the induced Borel structure. A *(standard) Borel groupoid* is a groupoid endowed with a standard Borel structure making composition and inversion of arrows Borel, and such that the set of objects is a Borel subset.

It is immediate to verify that principal analytic groupoids are precisely analytic equivalence relations on standard Borel spaces. Similarly principal Borel groupoids are precisely the Borel equivalence relations on standard Borel spaces. A functor between analytic groupoids is Borel if it is Borel as a function with respect to the given Borel structures.

#### 1.1.5 Polish groupoids and Polish groupoid actions

**Definition 1.1.3.** A *topological groupoid* is a groupoid endowed with a topology making composition and inversion of arrows continuous.

It is not difficult to see that for a topological groupoid the following conditions are equivalent:

- 1. Composition of arrows is open;
- 2. The source map is open;
- 3. The range map is open.

(See [121, Exercise I.1.8].)

**Definition 1.1.4.** A *Polish groupoid* is a groupoid endowed with a locally Polish topology such that

- 1. composition and inversion of arrows are continuous and open,
- 2. the set  $G^0$  of objects is a Polish space with the subspace topology,
- 3. for every  $x \in G^0$  the sets Gx and xG are Polish spaces with the subspace topology.

Polish groupoids have been introduced in [119] with the extra assumption that the topology be regular or, equivalently, globally Polish. It is nonetheless noticed in [119, page 362] that one can safely dispense of this additional assumption, without invalidating the results proved therein.

Suppose that G is a Polish groupoid, and X is a Polish space. A continuous action of G on X is given by a continuous function  $p: X \to G^0$  called *anchor map* together with a continuous function  $(g, x) \mapsto gx$  from

$$G \ltimes X = \{(\gamma, x) : p(x) = s(\gamma)\}$$

to X such that, for all  $\gamma, \rho \in G$  and  $x \in X$ 

- 1.  $\gamma(\rho x) = (\gamma \rho)x$ ,
- 2.  $p(\gamma x) = r(\gamma)$ , and
- 3. p(x)x = x.

In such a case we say that X is a Polish G-space. Similarly if X is a standard Borel space, then a Borel action of G on X is given by a Borel map  $p: X \to G^0$  together with a Borel map

$$\begin{array}{rccc} G \ltimes X & \to & X \\ (\gamma, x) & \to & \gamma x \end{array}$$

satisfying the same conditions as above. In this case X will be called a Borel G-space.

Clearly any Polish groupoid acts continuously on its space of objects  $G^0$  by setting p(x) = x and  $(\gamma, x) \mapsto r(\gamma)$ . This will be called the *standard* action of G on  $G^0$ .

Most of the usual notions for actions of groups, such as orbits, or invariant sets, can be generalized in the obvious way to actions of groupoids. If X is a G-space, and  $x \in X$ , then its orbit  $\{\gamma x : s(\gamma) = p(x)\}$  is denoted by [x]. The orbit equivalence relation  $E_G^X$  on X is defined by  $x E_G^X y$  iff [x] = [y]. If A is a subset of X, then its saturation

$$\{\gamma a : a \in A, \gamma \in Gp(a)\}$$

is denoted by [A]. An action is called *free* if  $\gamma x = \rho x$  implies  $\gamma = \rho$  for any  $x \in X$  and  $\gamma, \rho \in Gp(x)$ .

Suppose that G is a Polish groupoid, and X is a Borel G-space. If  $x, y \in G^0$  are in the same orbit define the stabilizer

$$G_x = \{ \gamma \in G : s(\gamma) = p(x) \text{ and } \gamma x = x \}$$

of x, and

$$G_{x,y} = \{ \gamma \in G : s(\gamma) = p(x) \text{ and } \gamma x = y \}.$$

Observe that by [73, Theorem 9.17]  $G_x$  is a closed subgroup of p(x)Gp(x). Therefore  $G_{x,y}$  is also closed, since  $G_{x,y} = G_{x,x}\rho$  for any  $\rho$  such that  $s(\rho) = p(x)$  and  $\rho x = y$ .

Suppose that X and Y are Borel G-spaces with anchor maps  $p_X$  and  $p_Y$ . A Borel fibred map from X to Y is a Borel function  $\varphi : X \to Y$  such that  $p_Y \circ \varphi = p_X$ . A Borel fibred

map from X to Y is G-equivariant if

$$\varphi(\gamma x) = \gamma \varphi(x)$$

for  $x \in X$  and  $\gamma \in Gp(x)$ . A Borel G-embedding from X to Y is an injective G-equivariant Borel fibred map from X to Y. Finally a Borel G-isomorphism from X to Y is a Borel G-embedding which is also onto.

#### 1.1.6 Some examples of Borel groupoids

In this subsection we show how several natural categories of interest can be endowed (after a suitable parametrization) with the structure of Borel groupoid.

Let us first consider the category of *complete separable metric spaces*, having surjective isometries as morphisms. This can be endowed with the structure of Borel groupoid in the following way. Denote by  $\mathbb{U}$  the Urysohn universal metric space. (A survey about  $\mathbb{U}$  and its remarkable properties can be found in [100].) Let  $F(\mathbb{U})$  be the Borel space of closed subsets of  $\mathbb{U}$  endowed with the Effros Borel structure. By universality of the Urysohn space,  $F(\mathbb{U})$  contains an isometric copy of any separable metric space. Moreover any surjective isometry between closed subsets of  $\mathbb{U}$  can be identified with its graph, which is a closed subset of  $\mathbb{U} \times \mathbb{U}$ . The set **CMS** of such graphs is easily seen to be a Borel subset of  $F(\mathbb{U})$ . Moreover a standard computation shows that composition and inversion of arrows are Borel functions in **CMS**. This shows that **CMS** is a Borel groupoid that can be seen as a parametrization of the category of metric spaces with surjective isometries as arrows.

More generally one can look at the category of separable  $\mathcal{L}$ -structures in some signature  $\mathcal{L}$  of continuous logic. (A complete introduction to continuous logic is [9].) One can identify any  $\mathcal{L}$ -structure with an  $\mathcal{L}$ -structure having as support a closed subset of  $\mathbb{U}$ . In such case the interpretation of a function symbol f can be seen as a closed subset of  $\mathbb{U}^{|f|+1}$  where |f|denotes the arity of f. The interpretation of a relation symbol B can be seen as a closed subset of  $\mathbb{U}^{|R|} \times \mathbb{R}$  where again |R| denotes the arity of B. (Here distances and relations are allowed to attain value in the whole real line.) The set  $Mod(\mathcal{L})$  of such structures can be verified to be a Borel subset of

$$F(\mathbb{U}) \times \prod_{f} F(\mathbb{U}) \times \prod_{B} F(\mathbb{U})$$

where f and B range over the function and relation symbols of  $\mathcal{L}$ . Similar parametrizations of the space of  $\mathcal{L}$ -structures can be found in [30] and [10]. As before the space  $Mod(\mathcal{L})$  of isomorphisms between  $\mathcal{L}$ -structures (identified with their graph) is a Borel subset of  $F(\mathbb{U})$ , and composition and inversion of arrows are Borel maps. Thus one can regard  $Mod(\mathcal{L})$ as the Borel groupoid of  $\mathcal{L}$ -structures. In the particular case when one considers *discrete* structures then one can replace the Urysohn space with  $\mathbb{N}$ .

As a particular case of separable structures in a given signature one can consider *separable C\*-algebras.* (The book [11] is a complete reference for the theory of operator algebras.) The complexity of the classification problem for separable C\*-algebras has recently attracted considerable interest; see [38, 37, 30, 125]. Particularly important classes for the classification program are *nuclear* and *exact* C\*-algebras; see [11, Section IV.3]. Separable exact C\*-algebras are precisely the closed self-adjoint subalgebras of the Cuntz algebra  $\mathcal{O}_2$  [78]. Thus the Borel groupoid C\*Exact of closed subalgebras of  $\mathcal{O}_2$ —where a \*-isomorphism between closed subalgebras is identified with its graph—can be regarded as a parametrization for the category of exact C\*-algebras having \*-isomorphisms as arrows. The category of (simple, unital) nuclear C\*-algebras can be regarded as the restriction of C\*Exact to the Borel set of (simple, unital) self-adjoint subalgebras of  $\mathcal{O}_2$ ; see [38, Section 7].

We now look at the category of *Polish groups* with continuous group isomorphisms as arrows. Denote by  $Iso(\mathbb{U})$  the group of isometries of the Urysohn space endowed with the topology of pointwise convergence. Recall that  $Iso(\mathbb{U})$  is a universal Polish group [133], i.e. it contains any other Polish group as closed subgroup. The space  $SG(Iso(\mathbb{U}))$  of closed subgroups of  $Iso(\mathbb{U})$  endowed with the Effros Borel structure can be regarded as the standard Borel space of Polish groups. Moreover a continuous isomorphism between closed subgroups of  $\operatorname{Iso}(\mathbb{U})$  can be identified with its graph, which is a closed subgroup of  $\operatorname{Iso}(\mathbb{U}) \times \operatorname{Iso}(\mathbb{U})$ . It is not difficult to check that the set **PG** of such closed subgroups of  $Iso(\mathbb{U}) \times Iso(\mathbb{U})$  is a Borel subset of the space  $SG(Iso(\mathbb{U}) \times Iso(\mathbb{U}))$  of closed subgroups of  $Iso(\mathbb{U}) \times Iso(\mathbb{U})$  endowed with the Effros Borel structure. (Fix a countable neighborhood basis  $\mathcal{N}$  of the identity in Iso( $\mathbb{U}$ ), and observe that a closed subgroup H of  $\operatorname{Iso}(\mathbb{U}) \times \operatorname{Iso}(\mathbb{U})$  is in **PG** if and only if  $\forall U \in \mathcal{N}$  $\exists V \in \mathcal{N}$  such that  $H \cap (U \times (\operatorname{Iso}(\mathbb{U}) \setminus cl(V))) = \emptyset$  and  $H \cap ((\operatorname{Iso}(\mathbb{U}) \setminus cl(V)) \times U) = \emptyset$ . Moreover a standard calculation shows that composition and inversion of arrows are Borel functions in **PG**. (For composition of arrows, observe that if as before  $\mathcal{N}$  is a countable basis of neighborhoods of the identity in  $Iso(\mathbb{U})$ , D is a dense subset of  $Iso(\mathbb{U})$ , A and B are open subsets of Iso(U), and  $\varphi, \psi \in \mathbf{PG}$ , then  $(\varphi \circ \psi) \cap (A \times B) \neq \emptyset$  if and only if there are  $U, V \in \mathbb{PG}$  $\mathcal{N}$  and  $g, h \in D$  with  $cl(V)^2h \subset B$  and  $Ug \subset A$  such that  $\psi \cap (U^2 \times (\operatorname{Iso}(\mathbb{U}) \setminus cl(V))) = \emptyset$ ,  $\varphi \cap (A \times Ug) \neq \emptyset$ , and  $\psi \cap (Ug \times Vh) \neq \emptyset$ .) This shows that **PG** is a Borel groupoid that can be seen as a parametrization of the category of Polish groups with continuous isomorphisms as arrows.

A similar discussion applies to the category of separable Banach spaces with linear (not necessarily isometric) isomorphisms as arrows. In this case one considers a universal separable Banach space, such as C[0,1]. One then looks at the standard Borel space of closed subspaces of C[0,1] as set of objects, and the set of closed subspaces of  $C[0,1] \oplus C[0,1]$  that code a linear isomorphism between closed subspaces of C[0,1] as set of arrows. The proof that these sets are Borel with respect to the Effros Borel structure is analogous to the case of Polish groups.

#### 1.1.7 The action groupoid

Suppose that G is a Polish groupoid, and X is a Polish G-space. Consider the groupoid

$$G \ltimes X = \{(\gamma, x) \in G \times X : s(\gamma) = p(x)\},\$$

where composition and inversion of arrows are defined by

$$(\rho, \gamma x) (\gamma, x) = (\rho \gamma, x)$$

and

$$(\gamma, x)^{-1} = \left(\gamma^{-1}, \gamma x\right).$$

The set of objects of  $G \ltimes X$  is

$$G^0 \ltimes X = \{(a, x) \in G^0 \times X : p(x) = a\}.$$

Endow  $G \ltimes X$  with the subspace topology from  $G \times X$ . Observe that the function

$$\begin{array}{rccc} X & \to & G^0 \ltimes X \\ x & \mapsto & (p(x), x) \end{array}$$

is a homeomorphism from X to the set of objects of  $G \ltimes X$ . We can therefore identify the latter with X. Under this identification the source of  $(\gamma, x)$  is x and the range is  $\gamma x$ . We claim that  $G \ltimes X$  is a Polish groupoid, called the *action groupoid* associated with the Polish G-space X. Clearly the topology is locally Polish, and composition and inversion of arrows are continuous. We need to show that the source map is open. Suppose that V is an open subset of G, U is an open subset of X, and W is the open subset

$$\{(\gamma, x) : \gamma \in V, x \in U\}$$

of  $G \ltimes X$ . Suppose that W is nonempty and pick  $(\gamma_0, x_0) \in W$ . Thus  $x_0 \in U$  and  $p(x_0) = s(\gamma_0) \in s[V]$ . Therefore there is an open subset  $U_0$  of U containing  $x_0$  such that  $p[U_0] \subset s[V]$ . We claim now that  $U_0$  is contained in the image of W under the source map. In fact if  $x \in U_0$  then  $p(x) = s(\gamma)$  for some  $\gamma \in V$  and therefore x is the source of the arrow  $(\gamma, x)$  in W. This concludes the proof of the fact that  $G \ltimes X$  is a Polish groupoid. To summarize we can state the following proposition.

**Proposition 1.1.5.** Suppose that G is a Polish groupoid, and X is a Polish G-space. The action groupoid  $G \ltimes X$  as defined above is a Polish groupoid. Moreover the map

$$\begin{array}{rcl} X & \to & (G \ltimes X)^0 \\ x & \mapsto & (p(x), x) \end{array}$$

is a homeomorphism such that, for every  $x, x' \in X$ ,

$$xE_G^X x'$$
 iff  $(p(x), x) E_{G \ltimes X} (p(x'), x')$ .

#### 1.1.8 Functorial reducibility

**Definition 1.1.6.** Suppose that G and H are analytic groupoids. A Borel reduction from G to H is a Borel functor F from G to H such that  $xGy \neq \emptyset$  whenever  $F(x)HF(y) \neq \emptyset$ .

Equivalently a Borel functor F from G to H is a Borel reduction from G to H when the function

$$\begin{array}{rccc} G^0 & \to & H^0 \\ x & \mapsto & F(x) \end{array}$$

is a Borel reduction from  $E_G$  to  $E_H$  in the sense of [44, Definition 5.1.1].

**Definition 1.1.7.** Suppose that G and H are analytic groupoids. We say that G is Borel reducible to H—in formulas  $G \leq_B H$ —if there is a Borel reduction from G to H.

The notion of bireducibility is defined accordingly.

**Definition 1.1.8.** Suppose that G and H are analytic groupoids. We say that G is *Borel bireducible* to H—in formulas  $G \sim_B H$ —if G is Borel reducible to H and vice versa.

When E and E' are principal analytic groupoids, then the Borel reductions from E to E' are in 1:1 correspondence with Borel reductions from E to E' in the usual sense of Borel

complexity theory; see [44, Definition 5.1.1]. In particular Definition 1.1.7 generalizes the notion of Borel reducibility from analytic equivalence relations to analytic groupoids.

Similarly as in the case of reducibility for equivalence relations, one can impose further requirements on the reduction map. If G and H are analytic groupoid, we say that G is injectively Borel reducible to H—in formulas  $G \sqsubseteq_B H$  if there is an *injective* Borel reduction from G to H. When G and H are Polish groupoid, one can also insist that the reduction be continuous rather than Borel. One then obtains the notion of continuous reducibility  $\leq_c$  and continuous injective reducibility  $\sqsubseteq_c$ .

Definition 1.1.7 provides a natural notion of comparison between analytic groupoids. This allows one to build a hierarchy of complexity of analytic groupoids, that includes the usual hierarchy of Borel equivalence relations. The *functorial Borel complexity* of an analytic groupoid will denote the position of the given groupoid in such a hierarchy.

#### 1.1.9 Category preserving maps

According to [101, Definition A.2] a continuous map  $f: X \to Y$  between Polish spaces is category preserving if for any comeager subset C of Y the inverse image  $f^{-1}[C]$  of C under f is a comeager subset of X. It is not difficult to see that any continuous open map is category preserving [101, Proposition A.3].

Category-preserving maps satisfy a suitable version of the classical Kuratowski-Ulam theorem for coordinate projections. We will state the particular case of this result for continuous open maps in the following lemma, which is Theorem A.1 in [101].

**Lemma 1.1.9.** Suppose that X is second countable space, Y is a Baire space, and  $f: X \to Y$  is an open continuous map such that  $f^{-1}\{y\}$  is a Baire space for every  $y \in Y$ . If  $A \subset X$  has the Baire property, then the following statements are equivalent:

- 1. A is comeager;
- 2.  $\forall^* y \in Y, A \cap f^{-1}\{y\}$  is comeager in  $f^{-1}\{y\}$ .

#### 1.1.10 Vaught transforms

Suppose in the following that G is a Polish groupoid,

$$\mathcal{A} = \{U_n : n \in \mathbb{N}\}$$

is a basis of Polish open subsets of G, and X is a Borel G-space.

**Definition 1.1.10.** For  $A \subset X$  and  $V \subset G$ , define the Vaught transforms

$$A^{\Delta V} = \{ x \in X : Vp(x) \neq \emptyset \text{ and } \exists^* \gamma \in Vp(x), \, \gamma x \in A \}$$

and

$$A^{*V} = \{x \in X : Vp(x) \neq \emptyset \text{ and } \forall^* \gamma \in Vp(x), \, \gamma x \in A\}.$$

In the particular case when G is a Polish group, and X is a Borel G-space, this definition coincide with the usual Vaught transform; cf. [44, Definition 3.2.2].

**Lemma 1.1.11.** Assume that B and  $A_n$  for  $n \in \mathbb{N}$  are subsets of X. If V is an open subset of G, then the following hold:

- 1.  $B^{\Delta G}$  and  $B^{*G}$  are invariant subsets of X;
- 2.  $(\bigcap_n A_n)^{*V} = \bigcap_n A_n^{*V};$
- 3.  $(\bigcup_n A_n)^{\triangle V} = \bigcup_n A_n^{\triangle V};$
- 4.  $p^{-1}[s[V]]$  is the disjoint union of  $(X \setminus B)^{*V}$  and  $B^{\triangle V}$ ;
- 5. If B is analytic, then

$$B^{\triangle V} = \bigcup \left\{ B^{*U} : V \supset U \in \mathcal{A} \right\} \quad and \quad B^{*V} = \bigcap \left\{ B^{\triangle U} : V \supset U \in \mathcal{A} \right\}.$$

Lemma 1.1.11 is elementary and can be proved similarly as [44, Proposition 3.2.5].

**Lemma 1.1.12.** Suppose that  $B \subset X$  is analytic, and  $U \subset G$  is open. If  $x \in X$  and  $\gamma \in Gp(x)$ , then the following statements are equivalent:

- 1.  $\gamma x \in B^{\Delta U}$ ;
- 2.  $x \in B^{*V}$  for some  $V \in \mathcal{A}$  such that  $V\gamma^{-1} \subset Ur(\gamma)$ ;
- 3.  $x \in B^{\Delta V}$  for some  $V \in \mathcal{A}$  such that  $V\gamma^{-1} \subset Ur(\gamma)$ ;
- 4. there are  $V, W \in \mathcal{A}$  such that  $VW^{-1} \subset U, \gamma \in W$ , and  $x \in B^{\Delta V}$ .
- (1) $\Rightarrow$ (2) By hypothesis  $Ur(\gamma) \neq \emptyset$  and  $\exists^* \rho \in Ur(\gamma)$  such that  $\rho \gamma x \in B$ . Therefore  $U\gamma \neq \emptyset$ and  $\exists^* \rho \in U\gamma$  such that  $\rho x \in B$ . Since B is analytic and the action is Borel, the set

$$\{\rho \in U\gamma : \rho x \in B\}$$

is analytic and in particular it has the Baire property. It follows that there is  $V \in \mathcal{A}$  such that  $Vp(x) \neq \emptyset$ ,  $Vp(x) \subset U\gamma$ , and  $\forall^* \rho \in V$ ,  $\rho x \in B$ . Observe that  $V\gamma^{-1} \subset Ur(\gamma)$ .

 $(2) \Rightarrow (3)$  Obvious.

- (3) $\Rightarrow$ (1) Observe that  $\varnothing \neq Vp(x) \subset U\gamma$ . Thus  $U\gamma \neq \varnothing$  and  $\exists^* \rho \in U\gamma$  such that  $\rho x \in B$ . Thus  $Up(\gamma z) \neq \varnothing$  and  $\exists^* \rho \in Up(\gamma x), \ \rho\gamma x \in B$ . This shows that  $\gamma x \in B^{\triangle U}$ .
- (2) $\Rightarrow$ (4) Pick  $v \in Vp(x)$  and observe that  $v\gamma^{-1} \in Ur(\gamma)$ . Therefore there are  $W, V_0 \in \mathcal{A}$ such that  $v \in V_0 \subset V, \gamma \in W$ , and  $V_0W^{-1} \subset U$ . Moreover since  $x \in B^{*V}$  and  $V_0 \subset V$ we have that  $x \in B^{*V_0}$ .

 $(4) \Rightarrow (2)$  Obvious.

If A is a subset of  $G \ltimes X$  and  $x \in X$ , then  $A_x$  denotes the x-fiber

$$\{\gamma \in G : (\gamma, x) \in A\}$$

of A. The following lemma is inspired by the Montgomery-Novikov theorem; see [73, Theorem 16.1].

**Lemma 1.1.13.** If A is a Borel subset of  $G \ltimes X$  and  $V \subset G$  is open, then

$$\{x \in X : Vp(x) \neq \emptyset \text{ and } A_x \text{ is nonmeager in } Vp(x)\}$$

is Borel. The same conclusion holds if one replaces "nonmeager" with "comeager" or "meager".

*Proof.* Define  $\mathcal{E}$  to be the class subsets of subsets A of  $G \ltimes X$  such that

$${x \in X : Vp(x) \neq \emptyset \text{ and } A_x \text{ is nonmeager in } Vp(x)}$$

is Borel for every nonempty open subset V of G. We claim that:

1.  $\mathcal{E}$  contains the sets of the form

$$U \ltimes B = \{(\rho, x) \in G \ltimes X : x \in B, \, \rho \in U\}$$

for  $B \subset X$  Borel and  $U \subset G$  open;

- 2.  $\mathcal{E}$  is closed by taking countable unions;
- 3.  $\mathcal{E}$  is closed by taking complements.

In fact:

1. If  $A = U \ltimes B$  where  $B \subset X$  is Borel and  $U \subset G$  is open then for every nonempty open set V

$$\{x \in X : Vp(x) \neq \emptyset \text{ and } A_x \text{ is nonmeager in } Vp(x)\}\$$
  
=  $B \cap p^{-1} [s [U \cap V]];$ 

is Borel.

2. If  $A = \bigcup_n A_n$  then for every nonempty open set

$$\{x \in X : Vp(x) \neq \emptyset \text{ and } A_x \text{ is nonmeager in } Vp(x)\}\$$
  
= 
$$\bigcup_{n \in \mathbb{N}} \{x \in X : Vp(x) \neq \emptyset \text{ and } (A_n)_x \text{ is nonmeager in } Vp(x)\};\$$

3. If  $A \subset G \ltimes X$  then for every nonempty open set V

$$\{x \in X : Vp(x) \neq \emptyset \text{ and } ((G \ltimes X) \setminus A)_x \text{ is nonmeager in } V\}$$

$$= \{x \in X : Vp(x) \neq \emptyset \text{ and } A_x \text{ is not comeager in } V\}$$

$$= \bigcup_{U_n \subset V} \{x \in X : U_n p(x) \neq \emptyset \text{ and } A_x \text{ is meager in } U_n\}$$

$$= \bigcup_{U_n \subset V} \left(p^{-1}s\left[U_n\right] \setminus \{x \in X : U_n p(x) \neq \emptyset \text{ and } A_x \text{ is nonmeager in } U_n\}\right).$$

A similar argument shows that the same conclusion holds after replacing "nonmeager" with "meager" or "comeager".  $\hfill \Box$ 

**Lemma 1.1.14.** If  $A \subset X$  is Borel and  $V \subset G$  is open then  $A^{\triangle V}$  and  $A^{*V}$  are Borel.

Proof. Consider the subset

$$\widetilde{A} = \{(\rho, x) \in G \ltimes X : \rho x \in A\}$$

and observe that A is a Borel subset of  $G \ltimes X$  such that

$$A^{\triangle V} = \left\{ x \in X : Vp(x) \neq \emptyset \text{ and } \widetilde{A}_x \text{ is nonmeger in } Vp(x) \right\}$$

and

$$A^{*V} = \left\{ x \in X : Vp(x) \neq \emptyset \text{ and } \widetilde{A}_x \text{ is comeager in } Vp(x) \right\}.$$

The conclusion now follows from Lemma 1.1.13.

**Lemma 1.1.15.** Assume that X is a Polish G-space. If  $B \subset X$ ,  $U \subset G$  is open, and  $\alpha \in \omega_1$ , then the following hold:

- 1. If B is open, then  $B^{\Delta U}$  is open;
- 2. If B is  $\Sigma^0_{\alpha}$ , then  $B^{\Delta U}$  is  $\Sigma^0_{\alpha}$  relatively to  $p^{-1}s[U]$ ;
- 3. If B is  $\Pi^0_{\alpha}$ , then  $B^{*U}$  is  $\Pi^0_{\alpha}$  relatively to  $p^{-1}s[U]$ .

Proof. The proof is analogous to the corresponding one for group actions; see Theorem 3.2.9 of [44]. Suppose that B is open, and pick  $x \in B^{\triangle U}$ . Thus  $Up(x) \neq \emptyset$  and  $\exists^* \rho \in Up(x)$  such that  $\rho x \in B$ . Pick  $U_0 \subset U$  open such that  $x \in B^{*U_0}$  and  $\rho \in U_0p(x)$  such that  $\rho x \in B$ . Since B is open and the action is continuous there are open subsets W and V containing x and  $\rho$  such that  $V \subset U_0$ ,  $VW \subset B$ , and  $p[W] \subset s[V]$ . We claim that  $W \subset B^{\triangle U}$ . In fact if  $w \in W$  then  $Vp(w) \neq \emptyset$ . Moreover since  $VW \subset B$ ,  $\exists^* \rho \in Vp(w)$  such that  $\rho w \in B$ . This concludes the proof that  $B^{\triangle U}$  is open. The other statements follow via (2),(3), and (4) of Lemma 1.1.11.

Using the Vaught transform it is easy to see that, if X is a Borel G-space, then the orbit equivalence relation  $E_G^X$  is idealistic. (This is well known when G is a Polish group; cf. [44, Proposition 5.4.10].) Recall that an equivalence relation E on a standard Borel space X is *idealistic* if there is a map  $[x]_E \mapsto I_{[x]_E}$  assigning to each equivalence class  $[x]_E$  of E and ideal  $I_{[x]_E}$  of subsets of  $[x]_E$  such that  $[x]_E \notin I_{[x]_E}$ , and for every Borel subset A of  $X \times X$ the set  $A_I$  defined by  $x \in A_I$  iff  $\{y \in [x]_E : (x, y) \in A\} \in I_{[x]_E}$  is Borel; see [44, Definition 5.4.9].

**Proposition 1.1.16.** If X is a Borel G-space, then the orbit equivalence relation  $E_G^X$  is idealistic.

*Proof.* Pick  $x \in X$  and denote by C the orbit of x. Define the ideal  $I_C$  of subsets of C by  $S \in I_C$  iff  $\forall^* \rho \in Gp(x), r(\rho) \notin S$ . Observe that this does not depend from the choice of x.

In fact suppose that  $y \in C$  and hence  $y = \gamma x$  for some  $\gamma \in Gp(x)$ . Assume moreover that  $S \subset C$  is such that  $\forall^* \rho \in Gp(x), r(\rho) \notin S$ . Consider the homeomorphism  $\Phi$  from Gp(x) to Gp(y) given by  $\rho \mapsto \rho \gamma$ . It is apparent that

$$\Phi\left[\{\rho \in Gp(x) : r(\rho) \notin S\}\right] = \left\{\rho \in Gp(y) : r(\rho) \notin S\right\}.$$

This shows that  $\forall^* \rho \in Gp(y), r(\rho) \notin S$ , and hence the definition of  $I_C$  is does not depend from the choice of  $x \in C$ . Clearly  $C \notin I_C$  since Gp(x) is a Baire space. It is not difficult to verify that  $I_C$  is a  $\sigma$ -ideal. Suppose that  $A \subset X \times X$  is Borel, and consider the set  $A_I$ defined by  $x \in A_I$  iff  $\{y \in [x] : (x, y) \in A\} \in I_{[x]}$ . Observe that  $x \in A_I$  iff  $\forall^* \rho \in Gp(x),$  $(x, r(\rho)) \notin A$ . Consider

$$X * X = \{(x, y) \in X \times X : p(x) = p(y)\}$$

and the action of G on X \* X defined by p(x, y) = p(x) = p(y) and  $\gamma(x, y) = (x, \gamma y)$  for  $\gamma \in Gp(x, y)$ . Observe that  $x \in A_I$  if and only if

$$(x,x) \in \left( \left( X * X \right) \setminus A \right)^{*G}$$

This shows that  $A_I$  is Borel by Lemma 1.1.14.

Let us denote as customary by  $E_1$  the tail equivalence relation for sequences in [0, 1]. If E is an idealistic Borel equivalence relation, then  $E_1$  is not Borel reducible to E by [71, Theorem 4.1]. Therefore we obtain from Proposition 1.1.16 the following corollary:

**Corollary 1.1.17.** If X is a Borel G-space with Borel orbit equivalence relations, then the orbit equivalence relation  $E_1$  is not Borel reducible to  $E_G^X$ .

Corollary 1.1.17 holds more generally for arbitrary Borel G-spaces, with not necessarily Borel orbit equivalence relations. This was shown by the present author in collaboration with Samuel Coskey, George Elliott, and Ilijas Farah by adapting the proof of [59, Chapter 8].

An equivalence relation E on a standard Borel space E is *smooth* if it is Borel reducible to the relation of equality in some Polish space [44, Definition 5.4.1]. By [44, Theorem 5.4.11] an equivalence relation has a Borel selector precisely when it is smooth and idealistic. Therefore the following corollary follows immediately from Proposition 1.1.16.

**Corollary 1.1.18.** If X is a Polish G-space such that  $E_G^X$  is smooth, then  $E_G^X$  has a Borel selector.

**Corollary 1.1.19.** If G and H are Polish groupoids such that  $E_G$  and  $E_H$  are smooth, then  $G \leq_B H$  if and only if  $E_G \leq E_H$ .

#### 1.1.11 Borel orbits

We now observe that, if G is a Polish groupoid, then the orbits of any Polish G-space are Borel.

**Proposition 1.1.20.** If G is a Polish groupoid, and X is a Polish G-space, then the orbit equivalence relation  $E_G^X$  is analytic and has Borel classes.

Proof. By Proposition 1.1.5 we can consider without loss of generality the case of the standard action of G on its set of objects  $G^0$ . Fix  $x \in G^0$  and consider the right action of xGx on Gx by composition. Observe that xGx is a Polish group, and Gx is a right Polish xGx-space with closed orbits. Therefore by [44, Proposition 3.4.6] the corresponding orbit equivalence relation  $E_{xGx}^{Gx}$  has a Borel transversal T. The orbit [x] is the image of T under the range map r. Since r is 1:1 on T, it follows that [x] is Borel by [73, Theorem 15.1]. Observe now that the orbit equivalence relation  $E_G$  is the image of the standard Borel space G under the Borel function  $\gamma \mapsto (r(\gamma), s(\gamma))$ . This shows that  $E_G$  is analytic.  $\Box$ 

Similarly as in the case of Polish group actions, a uniform bound on the complexity of the orbits in the Borel hierarchy entails Borelness of the orbit equivalence relation.

**Theorem 1.1.21.** Suppose that G is Polish groupoid, and X is a Polish G-space. The orbit equivalence relation  $E_G^X$  is Borel if and only if there is  $\alpha \in \omega_1$  such that every orbit is  $\Pi_{\alpha}^0$ 

*Proof.* One direction is obvious. For the other one consider for  $\alpha \in \omega_1$  the relation  $E_{\alpha}$  of X defined by

$$(x,y) \in E_c$$

iff for every *G*-invariant  $\Pi^0_{\alpha}$  set  $W \subset X$  we have that  $x \in W$  iff  $y \in W$ . If every orbit is  $\Pi^0_{\alpha}$  then  $E^X_G = E_{\alpha}$ . It is thus enough to prove that  $E_{\alpha}$  is co-analytic for every  $\alpha \in \omega_1$ . Consider a universal  $\Pi^0_{\alpha}$  subset *U* of  $\mathbb{N}^{\mathbb{N}} \times X$ . Define the action of *G* on  $\mathbb{N}^{\mathbb{N}} \times X$  by setting p(a,b) = p(b) and  $\gamma(a,b) = (a,\gamma b)$ . Define now

$$T = U^{*G}$$

and observe that T is  $\Pi^0_{\alpha}$  since U is  $\Pi^0_{\alpha}$ . Denote by  $T_a$  the section

$$\{b \in X : (a, b) \in T\}$$

for  $a \in \mathbb{N}^{\mathbb{N}}$ . We have that

$$b \in T_a \quad \Leftrightarrow \quad (a, b) \in T$$
$$\Leftrightarrow \quad \forall^* \gamma \in Gp(b), \ (a, \gamma b) \in U$$
$$\Leftrightarrow \quad \forall^* \gamma \in Gp(b), \gamma b \in U_a$$
$$\Leftrightarrow \quad b \in (U_a)^{*G}.$$

This shows that  $T_a$  is a *G*-invariant  $\Pi^0_{\alpha}$  subset of *X* for every  $\alpha \in \mathbb{N}^{\mathbb{N}}$ . Conversely if *A* is a *G*-invariant  $\Pi^0_{\alpha}$  set then  $A = U_a$  for some  $a \in \mathbb{N}^{\mathbb{N}}$  and hence

$$A = A^{*G} = (U_a)^{*G} = T_a.$$

This shows that  $\{T_a : a \in \mathbb{N}^{\mathbb{N}}\}$  is the collection of all invariant  $\Pi^0_{\alpha}$  sets. It follows that  $(x, y) \in E_{\alpha}$  iff  $\forall a \in \mathbb{N}^{\mathbb{N}}, (a, x) \in T$ . Therefore  $E_{\alpha}$  is co-analytic.

Theorem 1.1.21 was proved for Polish group actions in [126, Sections 3.6 and 3.7].

#### 1.2 Effros' theorem and the Glimm-Effros dichotomy

#### 1.2.1 Effros' theorem

**Lemma 1.2.1.** Suppose that G is a Polish groupoid. Consider the standard action of G on  $G^0$ , and the corresponding Vaught transform. If  $A \subset G^0$  is meager, then  $A^{\Delta G}$  is meager.

*Proof.* The source map  $r: G \to G^0$  is open and, in particular, category preserving; see Subsection 1.1.9. Thus  $r^{-1}[A]$  is a meager subset of G. Therefore, since the source map sis also open, by Lemma 1.1.9 the set of  $x \in X$  such that  $Gx \cap r^{-1}[A]$  is meager. This set is by definition  $A^{\Delta G}$ .

**Theorem 1.2.2.** Suppose that G is a Polish groupoid, X is a Polish G-space, and  $x \in X$ . Denote by [x] the orbit of x. The following statements are equivalent:

- 1. [x] is a  $G_{\delta}$  subset of X;
- 2. [x] is a Baire space;
- 3. [x] is nonmeaser in itself.

Proof. By Proposition 1.1.5 we can assume without loss of generality that  $X = G^0$  and  $G \cap G^0$  is the standard action. The only nontrivial implication is  $3 \Rightarrow 1$ . After replacing G with the restriction of G to the closure of [x], we can assume that [x] is dense in  $G^0$  and hence nonmeager in  $G^0$ . By Proposition 1.1.20, [x] is a Borel subset of  $G^0$  and in particular it has the Baire property. Therefore by [73, Proposition 8.23] the orbit [x] is the union of a meager set M and a  $G_{\delta}$  set U. One can conclude that  $[x] = U^{*G}$  arguing as in [126, Proposition 4.4]. Clearly [x] is the union of  $U^{*G}$  and  $M^{\Delta G}$ . By Lemma 1.2.1,  $M^{\Delta G}$  is meager and hence, since [x] is nonmeager,  $M^{\Delta G} = \emptyset$ . Therefore  $[x] = U^{*G}$  is  $G_{\delta}$  by Lemma 1.1.15.

Theorem 2.1 of [119] asserts that it is equivalent for the conditions in Theorem 1.2.2 to hold for all points of X.

Suppose that G is a Polish groupoid, X is a Polish G-space, and  $x \in G^0$ . The fiber Gp(x) is a Polish space, and the stabilizer  $G_x$  of x is a Polish group acting from the right by composition on Gp(x). One can then consider the quotient space  $Gp(x)/G_x$  and the quotient map  $\pi_x : Gp(x) \to Gp(x)/G_x$ , which is clearly continuous and open. When  $G \curvearrowright G^0$  is the standard action of G on its set of objects and  $x \in G^0$ , then the stabilizer  $G_x$  is just xGx.

It is not difficult to see that the proof of [119, Theorem 3.2] can be adapted to the context where G is a not necessarily regular Polish groupoid, as observed in [119, page 362]. The following lemma can then be obtained as an immediate consequence.

**Lemma 1.2.3** (Ramsay). Suppose that G is a Polish groupoid, and  $x \in G^0$ . If the orbit [x] of x is  $G_{\delta}$ , then the map  $\phi_x : G_x/xG_x \to [x]$  defined by  $\phi_x(\pi(\gamma)) = r(\gamma)$  is a homeomorphism.

**Corollary 1.2.4.** Suppose that G is a Polish groupoid, X is a Polish G-space, and  $x \in X$ . If the orbit [x] of x is  $G_{\delta}$ , then the map  $\phi_x : Gp(x)/G_x \to [x]$  defined by  $\phi_x(\pi(\gamma)) = \gamma x$  is a homeomorphism.

*Proof.* Consider the action groupoid  $G \ltimes X$ , and let us identify X with the space of objects of  $G \ltimes X$  as in Proposition 1.1.5. Consider the map  $\psi$  defined by

$$\begin{array}{rcl} Gp(x) & \to & (G \ltimes X) \, x \\ \\ \gamma & \mapsto & (\gamma, x). \end{array}$$

Observe that  $\psi$  is a continuous map with continuous inverse

$$\begin{array}{rccc} (G \ltimes X) \, x & \to & Gp(x) \\ (\gamma, x) & \mapsto & \gamma. \end{array}$$

Moreover the image of  $G_x$  under  $\psi$  is precisely  $x (G \ltimes X) x$ . The proof is then concluded by invoking Lemma 1.2.3.

#### 1.2.2 A Polish topology on quotient spaces

Suppose in this subsection that G is a Polish groupoid, which is moreover regular. Equivalently the topology of G is (globally) Polish. The following lemma is proved in [119, page 362].

**Lemma 1.2.5** (Ramsay). Suppose that G is a regular Polish groupoid. If U is an open subset of G containing the set of objects  $G^0$ , then there is an open subset V of G containing the set of objects  $G^0$  such that  $VV \subset U$ .

Fix  $x \in G^0$ . If V is a neighborhood of  $G^0$  in G, define the set

$$A_{V,x} = \left\{ (\rho, \gamma) \in Gx \times Gx : \rho \gamma^{-1} \in V \right\}.$$

Observe that, if  $\gamma \in Gx$ , then the collection of open subsets of the form  $V\gamma$ , where V is an open neighborhood of  $r(\gamma)$  in G, is a basis of neighborhoods of  $\gamma$  in Gx. It follows from this observation and Lemma 1.2.5 that the collection

 $\mathcal{U}_x = \left\{ A_{V,x} : V \text{ is a neighborhood of } G^0 \text{ in } G \right\}$ 

generates a uniformity compatible with the topology of Gx.

Suppose now that H is a closed subgroup of xGx, and consider the right action of H on Gx by translation. Denote by  $\pi$  the quotient map  $Gx \to Gx/H$ . Observe that  $\pi$  is continuous and open. If V is a neighborhood of  $G^0$  in G define

$$A_{V,x,H} = \left\{ (\pi(\gamma), \pi(\rho)) \in Gx/H \times Gx/H : \rho h \gamma^{-1} \in V \text{ for some } h \in xGx \right\}.$$

As before the collection

$$\mathcal{U}_{x,H} = \{A_{V,x,H} : V \text{ is a neighborhood of } G^0 \text{ in } G\}$$

generates a uniformity compatible with the topology of Gx/H.

**Proposition 1.2.6.** The quotient Gx/H is a Polish space.

*Proof.* The topology on Gx/H is induced by a countably generated uniformity, and hence it is metrizable. Since the quotient map  $\pi : Gx \to Gx/H$  is continuous and open, it follows from [44, Theorem 2.2.9] that Gx/H is Polish.

**Proposition 1.2.7.** Suppose that G is a regular Polish groupoid, and  $x \in G^0$ . Denote by  $\pi$  the quotient map

$$\pi: Gx \to Gx/xGx.$$

The following statements are equivalent:

- 1. The orbit [x] of x is a  $G_{\delta}$  subset of  $G^0$ ;
- 2. The map  $\phi_x : Gx/xGx \to [x]$  defined by  $\phi_x(\pi(\gamma)) = r(\gamma)$  is a homeomorphism.

*Proof.* The quotient space  $G_x/xG_x$  is Polish by Proposition 1.2.6. Therefore if  $\phi_x$  is a homeomorphism, then [x] is Polish, and hence a  $G_{\delta}$  subset of  $G^0$  by [73, Theorem 3.11]. The converse implication follows from Lemma 1.2.3.

#### **1.2.3** $G_{\delta}$ orbits

**Lemma 1.2.8.** Suppose that G is a Polish groupoid, and  $(U_n)_{n \in \mathbb{N}}$  is an enumeration of a basis of nonempty open subsets of  $G^0$ . If G has a dense orbit, then every element of  $\bigcap_n [U_n]$  has dense orbit.

The proof of Lemma 1.2.8 is immediate. Recall that  $[U_n]$  denotes the *G*-saturation  $r [s^{-1} [U_n]]$  of  $U_n$ .

**Lemma 1.2.9.** Suppose that G is a Polish groupoid. Define the equivalence relation  $\overline{E}$  on  $G^0$  by  $(x, y) \in \overline{E}$  iff the orbits of x and y have the same closure. The equivalence relation  $\overline{E}$  is  $G_{\delta}$  and contains  $E_G$ .

*Proof.* Suppose that  $(U_n)_{n \in \mathbb{N}}$  is an enumeration of a countable open basis of  $G^0$ . We have that  $(x, y) \in \overline{E}$  if and only if  $\forall n \in \mathbb{N}, x \in [U_n]$  iff  $y \in [U_n]$ . It follows that  $\overline{E}$  is  $G_{\delta}$ .  $\Box$ 

**Lemma 1.2.10.** Suppose that G is a Polish groupoid such that every orbit of G is  $G_{\delta}$ . If  $x, y \in G^0$  are such that  $[x] \neq [y]$  and  $[y] \cap \overline{[x]} \neq \emptyset$  then  $\overline{[y]} \cap [x] = \emptyset$ . Equivalently the quotient space  $G^0 / E_G$  is  $T_0$
Proof. After replacing G with the restriction of G to  $\overline{[x]}$  we can assume that  $\overline{[y]} \subset \overline{[x]} = X$ . Denote by  $(U_n)_{n \in \mathbb{N}}$  an enumeration of a basis of nonempty open subsets of  $G^0$ . By Lemma 1.2.8, every element of  $\bigcap_n [U_n]$  has dense orbit. Since  $[x] \cap [y] = \emptyset$ , [y] is not dense in X (otherwise it would be comeager and it would intersect [x]). It follows that, for some  $n \in \mathbb{N}, y \notin [U_n]$  and hence  $[y] \cap U_n = \emptyset$ . This shows that  $\overline{[y]} \subset X \setminus U_n$ . On the other hand  $U_n$  is invariant dense open and [x] is comeager, hence  $[x] \subset U_n$ . This shows that  $\overline{[y]} \cap [x] = \emptyset$ .

**Lemma 1.2.11.** Suppose that G is a Polish groupoid, and X is a Polish G-space. If every orbit is  $G_{\delta}$ , then  $E_G^X$  is smooth.

*Proof.* By Proposition 1.1.5 we can assume that  $X = G^0$  and  $G \curvearrowright G^0$  is the standard action. Observe that if  $x, y \in G^0$ , then [x] = [y] if and only if [x] and [y] have the same closure. This shows that the map  $x \mapsto \overline{[x]}$  from  $G^0$  to the space  $F(G^0)$  of closed subsets of x endowed with the Effros Borel structure is a reduction from  $E_G^X$  to equality in  $G^0$ . It remains to observe that such a map is Borel. In fact if U is an open subset of  $G^0$ , then

$$\left\{x \in G^0 : \overline{[x]} \cap U \neq \varnothing\right\} = [U] = r\left[s^{-1}\left[U\right]\right]$$

is open.

**Proposition 1.2.12.** Suppose that G is a Polish groupoid, and X is a Polish G-space. The following statements are equivalent:

- 1. Every orbit is  $G_{\delta}$ ;
- 2. The orbit equivalence relation  $E_G^X$  is  $G_{\delta}$ ;
- 3. The quotient space  $X/E_G^X$  is  $T_0$ .
- 4. The quotient topology generates the quotient Borel structure

*Proof.* In view of Proposition 1.1.5 we can assume without loss of generality that  $X = G^0$  and  $G \curvearrowright G^0$  is the standard action.

(1) $\Rightarrow$ (2) Consider the equivalence relation  $\overline{E}$  defined as in 1.2.9. Suppose that  $x, y \in X$  are such that  $(x, y) \in \overline{E}$ . It follows that [x] and [y] are both dense subsets of  $Y = \overline{[x]} = \overline{[y]}$ . Since both the orbit of x and y are  $G_{\delta}$ , [x] and [y] are comeager subsets of Y. It follows that they are not disjoint, and hence [x] = [y]. This shows that  $E_G = \overline{F}$  and in particular  $E_G$  is  $G_{\delta}$ .

 $(2) \Rightarrow (1)$  Obvious.

- $(2) \Rightarrow (3)$  Follows from Lemma 1.2.10.
- (3) $\Rightarrow$ (1) Since the quotient map  $\pi : X \to X/E_G$  is continuous and open,  $X/E_G$  has a countable basis  $\{U_n : n \in \mathbb{N}\}$ . If  $x \in X$  then

$$[x] = \bigcap \left\{ \pi^{-1} \left[ U_n \right] : n \in \mathbb{N}, \, \pi(x) \in U_n \right\}$$

This shows that [x] is  $G_{\delta}$ .

- (3)⇒(4) The Borel structure generated by the quotient topology is separating and countably generated. By [94, Theorem 4.2] it must coincide with the quotient Borel structure.
- (4) $\Rightarrow$ (3) Observe that the orbits are Borel. Therefore the quotient Borel structure is separating and hence the quotient topology separates points, i.e. it is  $T_0$ .

The equivalence of the conditions in Proposition 1.2.12 has been proved in [119, Theorem 2.1] under the additional assumption that the orbit equivalence relation is  $F_{\sigma}$ .

#### 1.2.4 The Glimm-Effros dichotomy

Denote by  $E_0$  the orbit equivalence relation on  $2^{\mathbb{N}}$  defined by  $(x, y) \in E_0$  iff x(n) = y(n)for all but finitely many  $n \in \mathbb{N}$ . Observe that  $E_0$  can be regarded as the (principal) Polish groupoid associated with the free action of  $\bigoplus_{n \in \mathbb{N}} \mathbb{Z}/2\mathbb{Z}$  on  $\prod_{n \in \mathbb{N}} \mathbb{Z}/2\mathbb{Z}$  by translation. The proof of the following result is contained in [119, Section 4]. An exposition of the proof in the case of Polish group actions can be found in [44, Theorem 6.2.1].

**Proposition 1.2.13.** Suppose that G is a Polish groupoid. If  $E_G$  is dense and meager in  $G^0 \times G^0$ , then  $E_0 \sqsubseteq_c G$ .

Recall that  $E_0 \sqsubseteq_c G$  means that there is an injective continuous functor  $F : E_0 \to G$ such that the restriction of F to the set of objects is a Borel reduction from  $E_0$  to  $E_G$ ; see Subsection 1.1.8. One can then obtain the following consequences:

**Proposition 1.2.14.** Suppose that G is a Polish groupoid. If G has no  $G_{\delta}$  orbits, then  $E_0 \sqsubseteq_c G$ .

*Proof.* After replacing G with the restriction of G to a class of the equivalence relation E defined as in Lemma 1.2.9, we can assume that every orbit is dense. By Theorem 1.2.2 every orbit is meager. It follows from Lemma 1.1.9 that  $E_G$  is meager. One can now apply Proposition 1.2.13.

**Theorem 1.2.15.** Suppose that G is a Polish groupoid. If every  $G_{\delta}$  orbit is  $F_{\sigma}$ , then either  $E_G$  is  $G_{\delta}$  or  $E_0 \sqsubseteq_c G$ .

Proof. Suppose that  $E_0 \not\subseteq_c G$ . In particular for every  $G_{\delta}$  subspace Y of  $G^0$ ,  $E_0 \not\subseteq_c G_{|Y}$ . Denote by  $\overline{E}$  the equivalence relation defined as in Lemma 1.2.9. If  $y \in X$  define  $Y = [y]_{\overline{E}}$ and observe that Y is an  $E_G$ -invariant  $G_{\delta}$  subset of  $G^0$ . Moreover every  $E_G$ -orbit contained in Y is dense in Y. Since  $E_0$  is not continuously reducible to G, by Proposition 1.2.14 there is  $z \in Y$  such that  $[z]_{E_G}$  is a dense  $G_{\delta}$  subset of Y. In particular  $[z]_{E_G}$  is  $G_{\delta}$  subset of  $G^0$ . Therefore by assumption also  $Y \setminus [z]_G$  is  $G_{\delta}$ . Since every orbit of Y is dense,  $Y \setminus [z]_{E_G}$  must be empty and  $[z]_{E_G} = Y = [y]_{\overline{E}}$  is  $G_{\delta}$ . This shows that every orbit of G is  $G_{\delta}$  and hence  $E_G$  is  $G_{\delta}$  by Proposition 1.2.12.

**Corollary 1.2.16.** Suppose that G and H are Polish groupoids such that every  $G_{\delta}$  orbit is  $F_{\sigma}$ . If  $E_G$  and  $E_H$  are Borel reducible to  $E_0$ , then  $G \leq H$  if and only if  $E_G \leq E_H$ .

*Proof.* Suppose that  $E_G \leq E_H$ . If  $E_H$  is smooth then the conclusion follows from Corollary 1.1.19. If  $E_G$  is not smooth then  $G \sim_B H \sim_B E_0$  by Theorem 1.2.15; see Definition 1.1.8.

We can combine Proposition 1.2.12 with Theorem 1.2.15 to get the following result. It was obtained in [119] under the additional assumption that the orbit equivalence relation  $E_G$  is  $F_{\sigma}$ . The notion of nonatomic and ergodic Borel measure with respect to an equivalence relation can be found in [44, Definition 6.1.5].

**Theorem 1.2.17.** Suppose that G is a Polish groupoid such that every  $G_{\delta}$  orbit is  $F_{\sigma}$ . The following statements are equivalent:

- 1. There is an orbit which is not  $G_{\delta}$ ;
- 2.  $E_0 \sqsubseteq_c G;$
- 3.  $E_0 \leq_B E_G;$
- 4. There is an  $E_G$ -nonatomic  $E_G$ -ergodic Borel probability measure on  $G^0$ ;
- 5.  $E_G$  is not smooth;

- 6.  $E_G$  is not  $G_{\delta}$ ;
- 7. Some orbit is not open in its closure.

Proof. The implication  $(1) \Rightarrow (2)$  follows from Theorem 1.2.15. The implication  $(2) \Rightarrow (3)$  is obvious. For  $(3) \Rightarrow (4)$ , observe that if  $f : 2^{\mathbb{N}} \to X$  is a Borel reduction from  $E_0$  to  $E_G$ ,  $\mu$  is the product measure on  $2^{\mathbb{N}}$ , and  $\nu$  is the push-forward of  $\mu$  under f, then  $\nu$  is an  $E_G$ -nonatomic and  $E_G$ -ergodic Borel probability measure on  $G^0$ . The implication  $(4) \Rightarrow (5)$ follows from [44, Proposition 6.1.6]. By Lemma 1.2.11 (5) implies 6). The implication  $(6) \Rightarrow (1)$  is contained in Proposition 1.2.12. Since a set that is open in its closure is  $G_{\delta}$ , the implication  $(1) \Rightarrow (7)$  is obvious. Let us show that  $(7) \Rightarrow (1)$ . Suppose that every orbit is  $G_{\delta}$ , and fix  $x \in G^0$ . After replacing G with its restriction to the closure of the orbit of x, we can assume that x has dense orbit. Therefore [x] is a dense  $G_{\delta}$  in x. Since [x] is by assumption also  $F_{\sigma}$ ,  $[x] = \bigcup_n F_n$  where the  $F_n$ 's are closed in X. Being [x] nonmeager in X, there is an open subset U of X contained in  $F_n$  for some  $n \in \mathbb{N}$ . Hence [x] = [U] is open.

## **1.3** Better topologies

#### **1.3.1** Polishability of Borel *G*-spaces

**Theorem 1.3.1.** Suppose that G is a Polish groupoid. Every Borel G-space is Borel Gisomorphic to a Polish G-space. Equivalently if X is a Borel G-space, then there is a Polish topology compatible with the Borel structure of G that makes the action of G on X continuous.

Theorem 1.3.1 answers a question of Ramsay from [120]. The rest of this subsection is dedicated to the proof of Theorem 1.3.1. The analogous statement for actions of Polish groups is proved in a similar way in [7, Theorem 5.2.1]. Fix a countable basis  $\mathcal{A}$  of Polish open subsets of G. Suppose that X is a Polish G-space. We want to define a topology t on X such that

- 1. t is Polish,
- 2. the action  $G \curvearrowright (X, t)$  is continuous, i.e. the anchor map  $p: X \to G^0$  is continuous and

$$\begin{array}{rcccc} G \ltimes X & \to & X \\ (\gamma, x) & \mapsto & \gamma x \end{array}$$

is continuous,

3. t generates the Borel structure of X.

By Lemma 1.1.14 and [7, 5.1.3 and 5.1.4] there exists a countable Boolean algebra  $\mathcal{B}$  of Borel subsets of X satisfying the following conditions:

- For all  $B \in \mathcal{B}$  and  $U \in \mathcal{A}, B^{\Delta U} \in \mathcal{B}$ ;
- The topology t' generated by the basis  $\mathcal{B}$  is Polish.

Observe that the identity function from X with its original Borel structure to (X, t') is Borel measurable, and hence a Borel isomorphism by [73, Theorem 15.1]. It follows that t'generates the Borel structure of X. Define S to be the set

$$\left\{B^{\triangle U}: B \in \mathcal{B}, \, U \in \mathcal{A}\right\},\,$$

and t to be the topology on X having S as subbasis.

**Claim.** The action  $G \curvearrowright (X, t)$  is continuous.

*Proof.* If  $V \in \mathcal{A}$  then

$$p^{-1}\left[s\left[V\right]\right] = X^{\Delta V} \in \mathcal{S}.$$

This shows that  $p: X \to G^0$  is *t*-continuous. Let us now show that the map  $G \ltimes X \to X$  is *t*-continuous. Suppose that  $B \in \mathcal{B}, U \in \mathcal{A}$ , and  $(\gamma_0, x_0) \in G \ltimes X$  is such that  $\gamma_0 x_0 \in B^{\triangle U}$ . By Lemma 1.1.12 there are  $W, V \in \mathcal{A}$  such that  $VW^{-1} \subset U, x_0 \in B^{\triangle V}$ , and  $\gamma_0 \in W$ . We claim that  $\gamma x \in B^{\triangle U}$  for every  $x \in B^{\triangle V}$  and  $\gamma \in W$ . Fix  $x \in B^{\triangle V}$  and  $\gamma \in W$  and observe that  $V\gamma^{-1} \subset VW^{-1} \subset U$  and hence it follows from Lemma 1.1.12 that  $\gamma x \in B^{\triangle U}$ . This shows that the action is continuous.

**Claim.** The space (X, t) is  $T_1$ .

Proof. Pick distinct points x, y of X. If  $p(x) \neq p(y)$  then there are disjoint  $V, W \in \mathcal{A}$  such that  $p(x) \in V$  and  $p(y) \in W$ . Thus  $p^{-1}[V]$  and  $p^{-1}[W]$  are open sets separating x and y. Suppose that p(x) = p(y). Consider the function  $f: Gp(x) \to X \times X$  defined by

$$f(\gamma) = (\gamma x, \gamma y)$$

Observe that f is Borel when  $X \times X$  is endowed with the  $t' \times t'$  topology. By [73, Theorem 8.38] there is a dense  $G_{\delta}$  subset Q of Gx such that the restriction of f to Q is  $(t' \times t')$ -continuous. Let  $\gamma_0 \in Q$ . Since  $\mathcal{B}$  is a basis for the Polish topology t' on X there are disjoint

elements B, C of  $\mathcal{B}$  such that  $\gamma_0 x \in B$  and  $\gamma_0 y \in C$ . Since f is  $(t' \times t')$ -continuous on Q there is  $U \in \mathcal{A}$  such that  $Up(x) \neq \emptyset$  and

$$f\left[Up(x)\cap Q\right]\subset B\times C.$$

Thus  $\forall^* \gamma \in Up(x), \ \gamma x \in B$  and  $\gamma y \in C$ . This shows that

$$x \in B^{\Delta U}, y \in C^{\Delta U}, y \notin B^{\Delta U}$$
, and  $x \notin C^{\Delta U}$ .

This concludes the proof that (X, t) is  $T_1$ .

**Claim.** The space (X, t) is regular.

Proof. Suppose that  $B \in \mathcal{B}$  and  $U \in \mathcal{A}$ . Pick  $x_0 \in B^{\Delta U}$ . It is enough to show that there is a *t*-open subset N of  $B^{\Delta U}$  containing  $x_0$  such that the *t*-closure of N is contained in  $B^{\Delta V}$ . Since  $x_0 \in B^{\Delta U}$  by Lemma 1.1.12 there are  $W_1, V_1 \in \mathcal{A}$  such that  $V_1 W_1^{-1} \cup V_1 \subset U$ ,  $p(x_0) \in W_1$ , and  $x_0 \in B^{\Delta V_1}$ . Since  $x_0 \in B^{\Delta V_1}$  again by Lemma 1.1.12 there are  $V_2, W_2 \in \mathcal{A}$ such that  $V_2 W_2^{-1} \subset V_1, p(x_0) \in W_2$ , and  $x_0 \in B^{\Delta V_2}$ . Define  $W \in \mathcal{A}$  such that

$$p(x_0) \in W \subset W_1^{-1} \cap W_2.$$

Consider

$$N = B^{\Delta V_2} \cap p^{-1}s\left[W\right]$$

and observe that N is a t-open subset of X containing  $x_0$ . We claim that the closure of N is contained in  $B^{\Delta U}$ . Define  $F = (B^{\Delta V_2})^{*W}$  and observe that F is relatively closed in  $p^{-1}s[W]$  by Lemma 1.1.11(4). We claim that

$$N \subset F \subset B^{\Delta U}.$$

Suppose that  $x \in N$ . If  $\gamma \in Wp(x)$  we have that

$$V_2\gamma^{-1} \subset V_2W^{-1} \subset V_2W_2^{-1} \subset V_1$$

Therefore  $\gamma x \in B^{\Delta V_1}$ . Being this true for every  $\gamma \in Wp(x)$ ,  $x \in (B^{\Delta V_1})^{*W} = F$ . Suppose now that  $x \in F$  and pick  $\gamma \in Wp(x)$  such that  $\gamma x \in B^{\Delta V_1}$ . We thus have

$$V_1 \gamma \subset V_1 W \subset V_1 W_1^{-1} \subset U$$

which implies by Lemma 1.1.12 that  $x = \gamma^{-1}(\gamma x) \in B^{\Delta U}$ . This concludes the proof that  $N \subset F \subset B^{\Delta U}$ . We will now show that the *t*-closure of N is contained in  $B^{\Delta U}$ . It is enough to show that if  $x \notin B^{\Delta U}$  then there is a *t*-open neighborhood of x disjoint from N. This is clear if  $p(x) \notin s[W]$ . Suppose now that  $p(x) \in s[W]$ . Since F is relatively closed in  $p^{-1}s[W]$  and

$$N \subset F \subset B^{\triangle V} \cap p^{-1}s\left[W\right]$$

we have that  $p^{-1}s[W] \setminus F$  is an open subset of X containing x and disjoint from N. This concludes the proof that the closure of N is contained in  $B^{\Delta V}$ . We have thus found an open neighborhood N of x whose closure is contained in  $B^{\Delta V}$ . This concludes the proof that (X, t) is regular.

**Claim.** The space (X, t) is strong Choquet.

*Proof.* Define  $\mathcal{C}$  to be the (countable) set of nonempty finite intersections of elements of  $\mathcal{S}$ and observe that  $\mathcal{C}$  is a basis for (X, t). Fix a well ordering E of the countable set  $\mathcal{C} \times \mathcal{B} \times \mathcal{A}$ . Let d' be a complete metric on X compatible with the Polish topology t'. We want to define a strategy for Player II in the strong Choquet game; see [73, Section 8.D]. Suppose that Player I plays t-open sets  $N_i$  for  $i \in \mathbb{N}$  and  $x_i \in N_i$ . At the *i*-th turn Player II will choose an element  $(M_i, B_i, U_i)$  of  $\mathcal{C} \times \mathcal{B} \times \mathcal{A}$  in such a way that the following properties hold:

1.  $x_i \in M_i$ ;

- 2. The *t*-closure of  $M_i$  is contained in  $N_i$ ;
- 3. The closure of  $U_{i+1}$  in  $U_0$  is contained in  $U_i$ ;
- 4. The t'-closure of  $B_{i+1}$  is contained in  $B_i$ ;
- 5. The d'-diameter of  $B_i$  is less than  $2^{-i}$ ;
- 6. The  $d_{U_0}$ -diameter of  $U_i$  is less than  $2^{-i}$  for  $i \ge 1$ , where  $d_{U_0}$  is a compatible complete metric on  $U_0$ ;
- 7.  $M_i \subset B_i^{\Delta U_i}$ .

Player II strategy is the following: At the *i*-th turn pick the *E*-least tuple  $(M_i, B_i, U_i)$ in  $\mathcal{C} \times \mathcal{B} \times \mathcal{A}$  satisfying properties (1)–(7). We need to show that the set of such tuples is nonempty. Observe that  $x_i \in N_i \subset M_{i-1} \subset B_{i-1}^{\Delta U_{i-1}}$ . Thus  $U_{i-1}p(x_i) \neq \emptyset$  and  $\exists^* \gamma \in$  $U_{i-1}p(x_i)$  such that  $\gamma x_i \in B_{i-1}$ . Since  $\mathcal{B}$  is a basis for (X, t') and  $\mathcal{A}$  is a basis for G we can find  $B_i$  and  $U_i$  such that

- (3)-(6) hold.
- $U_i p(x_i) \neq \emptyset$ , and
- $\exists^* \gamma \in U_i p(x_i)$  such that  $\gamma x_i \in B_i$ .

Consider  $M = B_i^{\Delta U_i} \cap N_i$  and observe that M is a *t*-open set containing  $x_i$ . Since (X, t) is regular there is  $M_i \in \mathcal{C}$  such that  $x_i \in M_i$  and the closure of  $M_i$  is contained in  $N_i \cap B_i^{\Delta U_i}$ . This ensures that (1), (2), (7) are satisfied. We now show that this gives a winning strategy for Player II. For every  $i \in \mathbb{N}$  we have that  $x_i \in M_i \subset B_i^{\Delta U_i}$  and hence there is  $\gamma_i \in U_i p(x_i)$  such that  $\gamma_i x_i = y_i \in B_i$ . Define  $\gamma$  to be the limit of the sequence  $(\gamma_i)_{i \in \mathbb{N}}$  in  $U_0$  and y to be the t'-limit of the sequence  $(y_i)_{i \in \mathbb{N}}$  in Y. Observe that

$$p(y) = \lim_{i \to i} p(y_i) = \lim_{i \to i} r(\gamma_i) = r(\gamma).$$

Define  $x = \gamma^{-1}y \in X$  and observe that x is the t-limit of the sequence  $(x_i)_{i \in \mathbb{N}}$ . Fix  $i \in \mathbb{N}$ . For j > i we have that  $x_j \in N_j \subset M_i$  and hence x is contained in the t-closure of  $M_i$ , which is in turn contained in  $N_i$ . This shows that  $x \in \bigcap_{i \in \mathbb{N}} N_i$ , concluding the proof that Player II has a winning strategy in the strong Choquet game in (X, t).

The proof of Theorem 1.3.1 is finished recalling that a regular  $T_1$  strong Choquet space is Polish [73, Theorem 8.18].

#### **1.3.2** Finer topologies for Polish *G*-spaces

**Theorem 1.3.2.** Suppose that G is a Polish groupoid, and  $(X, \tau)$  is a Polish G-space. Assume that  $V \subset G$  is an open Polish subset,  $P \subset X$  is  $\Sigma^0_{\alpha}$  for some  $\alpha \in \omega_1$ , and  $Q = P^{\Delta V}$ . There is a topology t on X such that:

- 1. t is a Polish topology;
- 2. t is finer that  $\tau$ ;
- 3. Q is t-open,
- 4. The action of G on (X, t) is continuous;
- 5. t has a countable basis  $\mathcal{B}$  such that for every  $B \in \mathcal{B}$  there is  $n \in \mathbb{N}$  such that B is  $\Sigma^{0}_{\alpha+n}$  with respect to  $\tau$ .

The analogous statement for actions of Polish groups is proved in a similar way in [7, Theorem 5.1.8]. Let  $\mathcal{A}$  be a countable basis of Polish open subsets for G containing V and let  $\mathcal{D}$  be a countable basis for  $(X, \tau)$ . By Lemma 1.1.15 and [7, 5.1.3, 5.1.4] there is a countable Boolean algebra  $\mathcal{B}$  of subsets of X satisfying the following:

- 1. For  $B \in \mathcal{B}$  and  $U \in \mathcal{A}$ ,  $B^{\Delta V} \in \mathcal{A}$ ;
- 2.  $P \in \mathcal{B};$
- 3.  $\mathcal{D} \subset \mathcal{B};$
- 4.  $\mathcal{B}$  is a basis for a Polish topology t';
- 5. For every  $B \in \mathcal{B}$ , there is  $n \in \mathbb{N}$  such that B is  $\Sigma_{\alpha+n}$  with respect to  $\tau$ .

Define

$$\mathcal{S} = \left\{ B^{\triangle V} : B \in \mathcal{B}, V \in \mathcal{A} \right\},$$

and

$$\mathcal{S}^* = \mathcal{S} \cup \mathcal{D}.$$

Consider the topology t on X having  $S^*$  as a subbasis. We claim that t is a Polish topology finer that  $\tau$  and coarser that t' making the action continuous. Clearly t is finer that  $\tau$  and in particular  $p: (X, t) \to G^0$  is continuous. The proof that the action is continuous and that t is a Polish topology is analogous to the proof of Theorem 1.3.1. The following corollary can be obtained from Theorem 1.3.2 together with [7, Subsection 5.1.3].

**Corollary 1.3.3.** Suppose that G is a Polish groupoid, and  $(X, \tau)$  is a Polish G-space. If  $\mathcal{J}$  is a countable collection of G-invariant Borel subsets of X, then there is a Polish topology t on X finer than  $\tau$  and making the action continuous such that all elements of  $\mathcal{J}$  are t-clopen.

## **1.4** Borel orbit equivalence relations

#### 1.4.1 A Borel selector for cosets

Suppose that G is a Polish groupoid. Denote by F(G) the standard Borel space of closed subsets of G endowed with the Effros Borel structure. A similar proof as [73, Theorem 12.13] shows that there is a Borel function

$$\sigma: F(G) \setminus \{\emptyset\} \to G$$

such that  $\sigma(A) \in A$  for every nonempty closed subset A of G. Denote by S(G) the Borel space of closed subgroupoids of G. This is the Borel subset of F(G) containing the closed subsets H of G such that for  $\gamma, \rho \in H$ ,  $\gamma^{-1} \in H$  and if  $r(\gamma) = s(\rho)$  then  $\rho\gamma \in H$ . If  $H \in S(G)$  denote by  $\sim_H$  the equivalence relation on G defined by  $\gamma \sim_H \rho$  iff  $\gamma = \rho h$  for some  $h \in H$  or, equivalently,  $\gamma H = \rho H$ .

**Proposition 1.4.1.** The relation  $\sim$  on  $G \times S(G)$  defined by  $(\gamma, H) \sim (\gamma', H')$  iff H = H'and  $\gamma H = \gamma' H'$  has a Borel transversal T.

*Proof.* Define the map f from  $S(G) \times G$  to F(G) by  $f(\gamma, H) = \gamma H$ . We claim that f is Borel. Let us show that if U is an open subset of G then the set

$$A_U = \{(\gamma, H) \in G \times S(G) : \gamma H \cap U \neq \emptyset\}$$

is Borel. Since the set

$$\left\{ (\rho, \gamma, H) \in G \times G \times S(G) : \gamma^{-1} \rho \in H \text{ and } \rho \in U \right\}$$

is Borel, its projection  $A_U$  on the last two coordinates is analytic. We want to show that  $A_U$  is co-analytic. Fix a countable basis of Polish open sets  $\{U_n : n \in \mathbb{N}\}$  for G. Observe that  $(H, \gamma) \in A_U$  if and only if there is  $n \in \mathbb{N}$  such that  $\gamma U_n \subset U$  and  $U_n \cap H \neq \emptyset$ . It is now enough to show that  $\{\gamma \in G : \gamma U_n \subset U\}$  is co-analytic. This follows from the fact that

$$\{(\gamma, \rho) \in G \times U_n : \text{either } r(\rho) \neq s(\gamma) \text{ or } r(\rho) = s(\gamma) \text{ and } \gamma \rho \in U\}$$

is a Borel set and it co-projection on the first coordinate is  $\{\gamma \in G : U_n \gamma \subset U\}$ . If now  $\sigma : F(G) \setminus \{\emptyset\} \to G$  is a Borel map such that  $\sigma(A) \in A$  for every nonempty closed subset A of G, define  $g(\gamma, H) = ((\sigma \circ f)(\gamma, H), H)$ . Observe that g is a Borel selector for  $\sim$ . Therefore the set

$$T = \{(\gamma, H) : g(\gamma, H) = (\gamma, H)\}$$

is a Borel transversal for  $\sim$ .

**Corollary 1.4.2.** If G is a Polish groupoid, and X is a Borel G-space, then the orbits are Borel subsets of X.

*Proof.* Observe that the stabilizer  $G_x$  is a closed subgroup of p(x)Gp(x) by [73, Theorem 9.17]. Consider a Borel transversal  $T_x$  for the equivalence relation  $\sim_{G_x}$ . The function  $\gamma \mapsto \gamma x$  from  $T \cap Gx$  to X is a 1:1 Borel function from  $T_x$  onto the orbit of x. This shows that the orbit of x is Borel by [73, Theorem 15.1].

#### **1.4.2** Borel orbit equivalence relations

Suppose that G is a Polish groupoid, and X is a Polish G-space. If  $x \in X$  then Lemma 1.1.14 and [7, 5.1.3 and 5.1.4] show that there is a sequence  $(B_{x,n})_{n\in\mathbb{N}}$  of Borel subsets of X such that  $[x] = B_{x,0}$  and

$$\mathcal{B}(x) = \{B_{x,n} : n \in \mathbb{N}\}$$

is a Boolean algebra that is a basis for a topology t(x) on X making the action continuous, and such that  $B^{\Delta U} \in \mathcal{B}(x)$  whenever  $B \in \mathcal{B}(x)$  and  $U \in \mathcal{A}$ . It is implicit in the proof of Lemma 1.1.14 and [7, 5.1.3, 5.1.4] that, under the additional assumption that the orbit equivalence relation  $E_G^X$  is Borel, the dependence of the sequence  $(\mathcal{B}_{x,n})_{n\in\mathbb{N}}$  from x is Borel, i.e. the relation

$$\mathcal{B}(y,x,n) \Leftrightarrow y \in \mathcal{B}_{x,n}$$

is Borel. This concludes the proof of the following lemma; see also [7, Lemma 7.1.3].

**Lemma 1.4.3.** Suppose that G is a Polish groupoid, and X is a Polish G-space. Assume that  $\mathcal{A}$  is a countable basis of Polish open subsets of G. If the orbit equivalence relation  $E_G^X$  is Borel, then there is a Borel subset  $\mathcal{B}$  of  $X \times X \times \mathbb{N}$  such that, letting

$$B_{x,n} = \{ y : (y, x, n) \in \mathcal{B} \}$$

and

$$\mathcal{B}(x) = \{B_{x,n} : n \in \mathbb{N}\},\$$

for every  $x \in X$  the following hold:

- 1.  $[x] = B_{x,0};$
- 2.  $B^{\Delta U} \in \mathcal{B}(x)$  for every  $B \in \mathcal{B}(x)$ , and  $U \in \mathcal{A}$ ;
- 3.  $\mathcal{B}(x)$  is a Boolean algebra;
- 4.  $\mathcal{B}(x)$  is a basis for a Polish topology t(x) on X making X a Polish G-space.

The following result provides a characterization of the Borel G-spaces with Borel orbit equivalence relation. The analogous result for Polish group actions is [7, Theorem 7.1.2].

**Theorem 1.4.4.** Suppose that G is a Polish groupoid, and X is a Borel G-space. The following statements are equivalent

1. The function

$$\begin{array}{rccc} X & \to & F(G) \\ x & \mapsto & G_x \end{array}$$

is Borel;

2. The function

$$\begin{array}{rccc} X \times X & \to & F(G) \\ (x,y) & \mapsto & G_{x,y} \end{array}$$

is Borel;

3. The orbit equivalence relation  $E_G^X$  is Borel.

Recall that, for  $x, y \in G^0$ ,  $G_x$  denotes the (closed) stabilizer

$$\{\gamma \in Gp(x) : \gamma x = x\}$$

and while  $G_{x,y}$  is the set

$$\{\gamma \in Gp(x) : \gamma x = y\};$$

see Subsection 1.1.5.

Proof. By Theorem 1.3.1 we can assume without loss of generality that X is a Polish Gspace. Fix a countable basis  $\mathcal{A} = \{U_n : n \in \mathbb{N}\}$  of nonempty Polish open subsets of G. Denote also by  $T \subset G \times S(G)$  a Borel transversal for the relation  $(\gamma, H) \sim (\gamma', H')$  iff H = H' and  $\gamma H = \gamma H'$  as in Proposition 1.4.1.

 $(1) \Rightarrow (2)$  Fix a nonempty open subset U of G. It is enough to show that the set

$$\{(x,y)\in X\times X: U\cap G_{x,y}\neq\varnothing\}$$

is co-analytic. Observe that  $G_{x,y} \cap U \neq \emptyset$  if and only if there is a unique  $\gamma \in G$  such that  $s(\gamma) = p(x), (\gamma, G_x) \in T$ , and  $\gamma G_x \cap U \neq \emptyset$ . Moreover  $\gamma G_x \cap U \neq \emptyset$  if and only if there is  $n \in \mathbb{N}$  such that  $\gamma U_n \subset U$  and  $U_n \cap G_x \neq \emptyset$ . Fix  $n \in \mathbb{N}$  and recall that by the proof of Proposition 1.4.1 { $\gamma \in G : \gamma U_n \subset U$ } is co-analytic. This concludes the proof that

$$\{(x, y) \in X \times X : U \cap G_{x, y} \neq \emptyset\}$$

is co-analytic.

 $(2) \Rightarrow (1)$  Obvious.

- (1) $\Rightarrow$ (3) Observe that  $(x, y) \in E_G$  if and only if there is a unique  $\gamma \in T$  such that  $(\gamma, G_x) \in T$  and  $r(\gamma) = y$ .
- (3) $\Rightarrow$ (1) Suppose that  $\mathcal{B}$ ,  $\mathcal{B}(x)$ , t(x), and  $B_{x,n}$  for  $x \in X$  and  $n \in \mathbb{N}$  are defined as in Lemma 1.4.3. Observe that the orbit  $[x] = B_{x,0}$  is open in t(x). It follows from Lemma 1.2.3 that the map

$$\begin{array}{rcl} Gp(x) \, / G_x & \to & [x] \\ & \gamma G_x & \mapsto & \gamma x \end{array}$$

is a t(x)-homeomorphism. We want to show that for every  $U \in \mathcal{A}$  the set

$$\{x \in X : G_x \cap U \neq \emptyset\}$$

is Borel. It is enough to show that for every basic nonempty U, V the set

$$\{x \in X : UG_x \cap V \neq \emptyset\}$$

is co-analytic. We claim that  $UG_x \cap V \neq \emptyset$  iff  $\exists U_m \subset U$  such that  $\forall B \in \mathcal{B}(x)$ ,  $x \in B^{\Delta U_m}$  implies  $x \in B^{\Delta V}$ . In fact suppose that  $UG_x \cap V \neq \emptyset$  and pick m such that  $U_m \subset U$  and  $U_m \gamma \subset V$  for some  $\gamma \in G_x$ . If  $x \in B^{\Delta U_m}$  then  $x = \gamma^{-1}x \in B^{\Delta U_m \gamma}$  and hence  $x \in B^{\Delta V}$  by Lemma 1.1.12. Conversely suppose that  $UG_x \cap V = \emptyset$  and hence

$$\{\gamma x : \gamma \in U\} \cap \{\gamma x : \gamma \in V\} = \emptyset.$$

Fix  $m \in \mathbb{N}$ . Since the map

$$\begin{array}{rcl} Gp(x) \, / G_x & \to & [x] \\ & \gamma G_x & \mapsto & \gamma x \end{array}$$

is a t(x)-homeomorphism, the set  $\{\gamma x : \gamma \in U_m\}$  is open in [x]. Thus there is  $B \in \mathcal{B}(x)$  such that

$$x \in B \subset \{\gamma x : \gamma \in U_m\}.$$

Moreover

$$\{\gamma \in Gp(x) : \gamma x \in B\}$$

is an open subset of Gp(x). Therefore there is  $k \in \mathbb{N}$  such that  $U_k \subset U_m$  and

$$U_k p(x) \subset \{\gamma \in Gp(x) : \gamma x \in B\}.$$

In particular  $x \in B^{\Delta U_k}$  but  $x \notin B^{\Delta V}$ .

# 1.5 Universal actions

Suppose that G is a Polish groupoid. The space G is fibred over the space of objects  $G^0$  via the source map  $r: G \to G^0$ . One can then consider the corresponding Effros fibred space  $F(G, G^0)$  of closed subsets of G contained in Gx for some  $x \in G^0$ ; see Subsection 1.1.3. Recall that  $F(G, G^0)$  is a standard Borel space fibred over  $G^0$  via the Borel map assigning x to a closed nonempty subset F of xG. Moreover  $F(G, G^0)$  has naturally the structure of Borel G-space given by the map

$$(\gamma, F) \mapsto \gamma F$$

for  $F \subset s(\gamma)G$ , where

$$\gamma F = \left\{ \gamma \rho : \rho \in F \right\}.$$

Similarly the fibred product

$$\underset{n\in\mathbb{N}}{\bigstar} F(G,G^0) = \left\{ (F_n)_{n\in\mathbb{N}} \in F(G,G)^{\mathbb{N}} : \exists x \in G^0 \ \forall n \in \mathbb{N}, \ F_n \subset xG \right\}$$

is naturally a Borel G-space with respect to the coordinate-wise action of G

$$\gamma(F_n)_{n\in\mathbb{N}} = (\gamma F_n)_{n\in\mathbb{N}}.$$

We want to show that the Borel G-space  $\bigstar_{n\in\mathbb{N}} F(G, G^0)$  is a universal Borel G-space. This means that if X is any Borel G-space, then there is a Borel G-embedding  $\varphi : X \to \bigstar_{n\in\mathbb{N}} F(G, G^0)$ ; see Subsection 1.1.5.

The following lemma is well known. A proof is included for convenience of the reader.

**Lemma 1.5.1.** If X is a Polish space,  $A \subset X$ , and E(A) is the set of  $x \in X$  such that for every neighborhood V of x,  $V \cap A$  is not meager, then E(A) is closed in X. Moreover A has the Baire property iff  $A \triangle E(A)$  is meager.

*Proof.* Clearly E(A) is closed, and if  $A \triangle E(A)$  is meager then A has the Baire property. Observe that if  $A, B \subset X$  are such that  $A \triangle B$  is meager, then E(A) = E(B). If A has the Baire property, there is an open subset U of X such that  $A \triangle U$  is meager. Thus E(A) = E(U) is equal to the closure  $\overline{U}$  of U. It follows that

$$A \bigtriangleup E(A) \subset (A \bigtriangleup U) \cup \left(\overline{U} \backslash U\right)$$

is meager.

Suppose that X is a Borel G-space. In view of Theorem 1.3.1 we can assume without loss of generality that X is in fact a Polish G-space. Fix a countable open basis  $\mathcal{B} = \{B_n : n \in \mathbb{N}\}$ of nonempty open subsets of X. Assume further that  $\mathcal{A}$  is a countable basis of Polish open subsets of G. Define for  $n \in \mathbb{N}$  the fibred Borel map  $\varphi_n : X \to F(G, G^0)$  by setting

$$\varphi_n(x) = (E\left(\{\gamma \in Gp(x) : \gamma x \in B_n\}\right))^{-1}$$

Define the Borel fibred map  $\varphi: X \to \bigstar_{n \in \mathbb{N}} F(G, G^0)$  by  $\varphi(x) = (\varphi_n(x))_{n \in \mathbb{N}}$ .

**Claim.**  $\varphi$  is Borel measurable

It is enough to show that  $\varphi_n$  is Borel measurable for every  $n \in \mathbb{N}$ . Suppose that  $V \in \mathcal{A}$ . We want to show that the set of  $x \in X$  such that

$$E\left(\{\gamma \in Gp(x) : \gamma x \in B_n\}\right) \cap V \neq \emptyset$$

is Borel. Observe that

$$E\left(\{\gamma \in Gp(x) : \gamma x \in B_n\}\right) \cap V \neq \emptyset$$

if and only if  $\exists W \in \mathcal{A}$  such that  $W \subset V$  and

$$\{\gamma \in Gp(x) : \gamma x \in B_n\}$$

is comeager in Wp(x). The set of such elements x of X is Borel by Lemma 1.1.13.

**Claim.**  $\varphi$  is G-equivariant, i.e.  $\varphi(\gamma x) = \gamma \varphi(x)$  for  $(\gamma, x) \in G \ltimes X$ 

It is enough to show that  $\varphi_n(\gamma x) = \gamma \varphi_n(x)$  for  $n \in \mathbb{N}$ . Observe that

$$\varphi_n(\gamma x) = \left(E\left(\{\rho \in Gr(\gamma) : \rho \gamma x \in B_n\}\right)\right)^{-1}$$

and

$$\varphi_n(x) = \left(E\left(\{\tau \in Gp(x) : \tau x \in B_n\}\right)\right)^{-1}$$

We thus have to prove that

$$\left(E\left(\{\rho \in Gr(\gamma) : \rho\gamma x \in B_n\}\right)\right)^{-1} = \gamma \left(E\left(\{\tau \in Gp(x) : \tau x \in B_n\}\right)\right)^{-1}$$

or equivalently

$$E\left(\left\{\rho \in Gr(\gamma) : \rho\gamma x \in B_n\right\}\right) = E\left(\left\{\tau \in Gp(x) : \tau x \in B_n\right\}\right)\gamma^{-1}$$

Since  $\tau \mapsto \tau \gamma^{-1}$  is a homeomorphism from Gp(x) to  $Gr(\gamma)$  we have that

$$E\left(\left\{\tau \in Gp(x) : \tau x \in B_n\right\}\right)\gamma^{-1} = E\left(\left\{\tau \in Gp(x) : \tau x \in B_n\right\}\gamma^{-1}\right)$$
$$= E\left(\left\{\rho \in Gr(\gamma) : \rho\gamma x \in B_n\right\}\right)$$

**Claim.**  $\varphi$  is injective

Assume that  $x, y \in X$  are such that  $\varphi(x) = \varphi(y)$ . Thus p(x) = p(y) and for every  $n \in \mathbb{N}$ 

$$\{\gamma \in Gp(x) : \gamma x \in B_n\} \bigtriangleup \{\gamma \in Gp(y) : \gamma y \in B_n\}$$

is meager. Thus  $\forall^* \gamma \in Gp(x), \forall n \in \mathbb{N}, \gamma x \in B_n$  iff  $\gamma y \in B_n$ . Thus for some  $\gamma \in Gp(x)$ ,  $\gamma x \in B_n$  iff  $\gamma y \in B_n$  for all  $n \in \mathbb{N}$ . This implies that  $\gamma x = \gamma y$  and hence x = y.

# 1.6 Countable Borel groupoids

#### 1.6.1 Actions of inverse semigroups on Polish spaces

An *inverse semigroup* is a semigroup T such that every  $t \in T$  has a semigroup-theoretic inverse  $t^* \in T$ . This means that  $t^*$  is the unique element of T such that

$$tt^*t = t$$
 and  $t^*tt^* = t^*$ .

If T is an inverse semigroup, then the set E(T) of idempotent elements is a commutative subsemigroup of T, and hence a semilattice; see [106, Proposition 2.1.1]. In particular E(T)has a natural order defined by

$$e \leq f$$
 iff  $ef = fe = e$ .

Observe that for every  $t \in T$  the elements  $tt^*$  and  $t^*t$  are idempotent.

Suppose that X is a Polish space. The semigroup  $\mathcal{H}(X)$  of partial homeomorphisms between open subsets of X is clearly an inverse semigroup.

**Definition 1.6.1.** An action  $\theta: T \curvearrowright X$  of a countable inverse semigroup T on the Polish space X is a semigroup homomorphism  $\theta: t \mapsto \theta_t$  from T to  $\mathcal{H}(X)$ .

Observe that a semigroup homomorphism between inverse semigroups automatically preserves inverses; see [106, Proposition 2.1.1].

#### 1.6.2 Étale Polish groupoids

Suppose that G is a Polish groupoid. A subset u of G is a bisection if the source and range maps restricted to u are injective. A bisection of G is open if it is an open subset of G. It is not difficult to verify that the following conditions are equivalent:

- 1. The source and range maps of G are local homeomorphisms from G to  $G^0$ ;
- 2. Composition of arrows in G is a local homeomorphism from  $G^2$  to G;
- 3. G has a countable basis of open bisections;
- 4. G has a countable inverse semigroup of open bisections that is basis for the topology of G;
- 5.  $G^0$  is an open subset of G.

When these conditions are satisfied, G is called étale Polish groupoid. If G is an étale Polish groupoid, then in particular for every  $x \in G^0$  the fiber Gx is a countable discrete subset of G.

#### 1.6.3 The groupoid of germs

Suppose that  $\theta : T \curvearrowright X$  is an action of a countable inverse semigroup on a Polish space. We want to associate to such an action an étale Polish groupoid  $\mathcal{G}(\theta, T, X)$  that contains all the information about the action. This construction can be found in [31] in the case when X is locally compact.

If  $e \in E(T)$  denote by  $D_e$  the domain of  $\theta_e$ . Observe that the domain of  $\theta_t$  is  $D_{t^*t}$  and the range of  $\theta_t$  is  $D_{tt^*}$ . Define  $\Omega$  to be the subset of  $T \times X$  of pairs (u, x) such that  $x \in D_{u^*u}$ . Consider the equivalence relation  $\sim$  on  $\Omega$  defined by  $(u, x) \sim (v, y)$  iff x = y and for some  $e \in E(S)$ , ue = ve and  $x \in D_e$ . The equivalence class [u, x] of (u, x) is called the germ of u at x. Observe that if e witnesses that  $(u, x) \sim (v, x)$  then, after replacing e with  $u^*uv^*ve$  we can assume that  $e \leq u^*u$  and  $e \leq v^*v$ . It can be verified as in [31, Proposition 4.7] that if (u, x) and (v, y) are in  $\Omega$  and  $x = \theta_v(y)$  then  $(uv, y) \in \Omega$ . Moreover the germ [uv, y] of uv at y depends only on [x, s] and [y, t].

One can then define the groupoid  $\mathcal{G}(\theta, T, X) = \Omega / \sim$  of germs of the action  $S \curvearrowright X$ obtained by setting

- $\mathcal{G}(\theta, T, X)^2 = \{([u, x], [v, y]) : \theta_v(y) = x\},\$
- [u, x] [v, y] = [uv, y], and
- $[u, x]^{-1} = [\theta_{u^*}, \theta_u(x)].$

Observe that the map  $x \mapsto [e, x]$  from X to G, where e is any element of E(S) such that  $x \in D_e$ , is a well-defined bijection from X to the set of objects  $G^0$  of G. Identifying X with  $G^0$  we have that the source and range maps s and r are defined by

$$s[u,x] = x$$

and

$$r\left[u,x\right] = \theta_u(x).$$

We now define the topology of  $\mathcal{G}(\theta, T, X)$ . For  $u \in T$  and  $U \subset D_{u^*u}$  open define

$$\Theta(u, U) = \{ [u, x] \in G : x \in U \}$$

It can be verified as in [31, Proposition 4.14, Proposition 4.15, Corollary 4.16, Proposition 4.17, and Proposition 4.18] that the following hold:

- 1.  $\mathcal{G}(\theta, T, X)$  is an étale Polish groupoid;
- 2. the map  $x \mapsto [e, x]$  where e is any element of E(S) such that  $x \in D_s$ , is a homeomorphism from X onto the space of objects of  $\mathcal{G}(\theta, T, x)$ ;
- 3. if  $u \in T$  and  $U \subset D_{u^*u}$  then  $\Theta(u, U)$  is an open bisection of U, and the map  $x \mapsto [u, x]$  is a homeomorphism from U onto  $\theta(u, U)$ ;
- 4. if  $\mathcal{A}$  is a basis for the topology of X, then the collection

$$\{\theta (u, A \cap D_{u^*u}) : u \in S, A \in \mathcal{A}\}$$

is a basis of open bisections for  $\mathcal{G}(\theta, T, X)$ .

#### **1.6.4** Regularity of the groupoid of germs

The groupoid of germs  $\mathcal{G}(\theta, T, X)$  for an action  $\theta : T \curvearrowright X$  is in general not Hausdorff, even when X is locally compact. Here we isolate a condition that ensures that  $\mathcal{G}(\theta, T, X)$ is regular.

Define the order  $\leq$  on T by setting  $u \leq v$  iff  $u = vu^*u$ . Observe that this extends the order of E(T). Moreover if  $u \leq v$  then

$$u^*u = v^*vu^*u^*v^*v = v^*vu^*u$$

and hence  $u^*u \leq v^*v$ . We say that T is a *semilattice* if it is a semilattice with respect to the order  $\leq$  just defined, i.e. for every pair u, v of elements of T there is a largest element  $u \wedge v$  below both u and v.

**Proposition 1.6.2.** Suppose that T is a semilattice. If there is a subset C of T such that:

- 1. for every  $u \in T$  and  $x \in D_{u^*u}$  there is  $c \in C$  such that  $x \in D_{(u \wedge c)^*(u \wedge c)}$ , and
- 2. for every distinct  $c, d \in C$ ,  $\Theta(c, D_{c^*c}) \cap \Theta(d, D_{d^*d}) = \emptyset$ ,

then the groupoid of germs  $\mathcal{G}(\theta, T, X)$  is regular.

Proof. Suppose that [u, x] is an element of  $\mathcal{G}(\theta, T, X)$ , and W is an open neighborhood of [u, x] in  $\mathcal{G}(\theta, T, X)$ . There are an open subset U of X contained in  $D_{u^*u}$  such that  $[u, x] \in \Theta(u, U) \subset W$ . Pick  $c \in C$  such that  $x \in D_{(u \wedge c)^*(u \wedge c)}$ , and an open neighborhood V of x with  $\overline{V}$  contained in  $U \cap D_{(u \wedge c)^*(u \wedge c)}$ . We claim that  $\Theta(u \wedge c, V)$  is an open neighborhood of [u, x] whose closure is contained in W. To show this it is enough to show that  $\Theta(u \wedge c, \overline{V})$  is closed in  $\mathcal{G}(\theta, T, X)$ . Pick  $[v, y] \in \mathcal{G}(\theta, T, X) \setminus \Theta(u \wedge c, \overline{V})$ . If  $y \notin \overline{V}$  then clearly there is an open neighborhood of [t, y] disjoint from  $\Theta(u \wedge c, \overline{V})$ . Suppose that  $y \in \overline{V}$ . Pick  $d \in C$  such that  $y \in D_{(u \wedge d)^*(u \wedge d)}$ . In such case we have that

$$\Theta\left(u\wedge d, D_{(u\wedge d)^*(u\wedge d)}\right)$$

is an open neighborhood of y disjoint from  $\Theta(u \wedge c, \overline{V})$ . This concludes the proof.

#### 1.6.5 Étale groupoids as groupoids of germs

Suppose that G is an étale Polish groupoid, and  $\Sigma$  is a countable inverse semigroup of open bisections of G. One can define the standard action of  $\Sigma$  on  $G^0$  by setting  $D_e = e$  for every  $e \in E(\Sigma)$ , and  $\theta_u : D_{u^*u} \to D_{uu^*}$  by

$$\theta_u(x) = r(ux),$$

where ux is the only element of u with source x. The same proof as [31, Proposition 5.4] shows the following fact:

**Proposition 1.6.3.** Suppose that  $\Sigma$  is a countable inverse semigroup of open bisections of G such that  $\bigcup \Sigma = G$  and for every  $u, v \in \Sigma$ ,  $u \cap v$  is the union of the elements of  $\Sigma$ contained in  $u \cap v$ . Consider the standard action  $\theta : \Sigma \cap G^0$ . The map from  $\mathcal{G}(\theta, \Sigma, X)$ to G assigning to the germ [u, x] of u at x the unique element of u with source x is well defined, and it is an isomorphism of étale Polish groupoids.

In particular every étale Polish groupoid is isomorphic to the groupoid of germs of an action of an inverse semigroup on a Polish space.

#### **1.6.6** Borel bisections

We will say that a (standard) Borel groupoid is countable if for every  $x \in G^0$ , the set  $Gx = s^{-1}[\{x\}]$  is countable. Observe that the countable Borel equivalence relations are exactly the principal countable Borel groupoids.

Suppose that G is a countable Borel groupoid. Observe that the set  $\mathcal{S}(G)$  of Borel bisections of G is an inverse semigroup. The idempotent semilattice E(S) is the Boolean algebra of Borel subsets of  $G^0$ . The order  $\leq$  on  $\mathcal{S}(G)$  as in Subsection 1.6.4 is defined by  $u \leq v$  iff  $u \subset v$ . Therefore (S) is a semilattice with  $u \wedge v = u \cap v$ .

**Lemma 1.6.4.** Suppose that X, Z are standard Borel spaces and  $s : Z \to X$  is a Borel countable-to-one surjection. There is a countable partition  $(P_n)_{n \in \mathbb{N}}$  of Z into Borel subsets such that  $s_{|P_n}$  is 1:1 for every  $n \in \mathbb{N}$ .

*Proof.* It is enough to show that  $Z = \bigcup_n P_n$ , where  $P_n$  are Borel subsets of Z such that  $s_{|P_n}$  is 1:1. After replacing Z with the disjoint union of Z and  $X \times \mathbb{N}$ , and setting s(x, n) = x for  $(x, n) \in X \times \mathbb{N}$ , we can assume that for every  $x \in X$  the inverse image  $s^{-1} \{x\}$  is countably infinite. We want to define a Borel function  $e: X \to Z^{\mathbb{N}}$  such that  $\{e(x)_n : n \in \mathbb{N}\}$  is an

enumeration of  $s^{-1}\{x\}$  for every  $x \in X$ . Consider the Borel subset E of  $X \times Z^{\mathbb{N}}$  defined by

$$(x, (e_n)) \in E \quad \Leftrightarrow \quad (e_n) \text{ is a enumeration of } s^{-1} \{x\}$$
  
 $\Leftrightarrow \quad s(e_n) = x \text{ and } \forall z \in s^{-1} \{x\} \exists n \text{ such that } z = e_n$ 

(Recall that the image of a standard Borel space under a countable-to-one Borel function is Borel; see [73, Exercise 18.15].) We want to find a Borel uniformization of E. For each  $x \in X$  endow  $s^{-1} \{x\}$  with the discrete topology and  $s^{-1} \{x\}^{\mathbb{N}}$  with the product topology. Observe that for  $(e_n) \in s^{-1} \{x\}^{\mathbb{N}}$  we have that  $(e_n) \in E_x$  iff  $\forall z \in s^{-1} \{x\} \exists n \in \mathbb{N}$  such that  $e_n = z$ . Thus  $E_x$  is a dense  $G_{\delta}$  subset of  $s^{-1} \{x\}^{\mathbb{N}}$ . Define the following  $\sigma$ -ideal  $\mathcal{I}_x$  in  $Z^{\mathbb{N}}$ :  $A \in \mathcal{I}_x$  iff  $A \cap E_x$  is meager in  $s^{-1} \{x\}^{\mathbb{N}}$ . Thus  $E_x \notin \mathcal{I}_x$ . In order to conclude that Ehas a Borel uniformization, by [73, Theorem 18.6] it is enough to show that the assignment  $x \mapsto \mathcal{I}_x$  is Borel-on-Borel as in [73, Definition 18.5]. Suppose that Y is a standard Borel space and  $A \subset Y \times X \times Z^{\mathbb{N}}$ . Consider the set

$$\{(y,x) \in Y \times X : A_{y,x} \in \mathcal{I}_x\}$$
  
= 
$$\left\{(y,x) \in Y \times X : A_{y,x} \cap E_x \text{ is meager in } s^{-1} \{x\}^{\mathbb{N}}\right\}$$

Clearly we can assume that  $A \subset Y \times E$ . If  $e : \mathbb{N} \to s^{-1} \{x\}$  is a bijection, then e induces a homeomorphism  $\pi_e : \mathbb{N}^{\mathbb{N}} \to s^{-1} \{x\}^{\mathbb{N}}$ . Therefore for  $(y, x) \in Y \times X$  we have that

$$A_{y,x} \cap E_x \text{ is meager in } s^{-1} \{x\}^{\mathbb{N}} \quad \Leftrightarrow \quad \pi_e^{-1} \left[ A_{y,x} \cap s^{-1} \{x\}^{\mathbb{N}} \right] \text{ is meager}$$
$$\Leftrightarrow \quad \left\{ w \in \mathbb{N}^{\mathbb{N}} : \pi_e(w) \in A_{y,x} \right\} \text{ is meager}.$$

Consider the Borel subset Q of  $Y \times X \times Z^{\mathbb{N}}$  defined by  $(y, x, e) \in Q$  iff  $(x, e) \in E$  and  $\forall n, m \in \mathbb{N}$  if  $n \neq m$  then  $e_n \neq e_m$  and  $\{w \in \mathbb{N}^{\mathbb{N}} : (y, x, e \circ w) \in A_{y,x}\}$  is meager. We have that

$$\begin{array}{rcl} A_{y,x} \in \mathcal{I}_x & \Leftrightarrow & \exists e \text{ such that } (y,x,e) \in Q \\ & \Leftrightarrow & \forall e \forall n \neq m \in \mathbb{N}, \ (x,e) \in E, \text{ and } e_n \neq e_m \ \Rightarrow \ (z,x,e) \in Q. \end{array}$$

This shows that  $\{(y, x) : A_{y,x} \in \mathcal{I}_x\}$  is both analytic and co-analytic, and hence Borel. **Proposition 1.6.5.** If G is a countable Borel groupoid, then there is a countable partition of G into Borel bisections. Moreover for every  $n \in \mathbb{N}$  we have that

$$\left\{x \in G^0 : |Gx| = n\right\}$$

is Borel.

*Proof.* The source map  $s: G \to G^0$  satisfies the hypothesis of Lemma 1.6.4. Therefore one can find a countable partition  $\mathcal{H}$  of G into Borel subsets such that the source map is 1:1 on every element of  $\mathcal{H}$ . Define

$$\mathcal{C} = \left\{ u \cap v^{-1} : u, v \in \mathcal{H} \right\}$$

an observe that C is a countable collection of pairwise disjoint Borel bisections of G. Observe now that for every  $u \in C$ ,

$$\{x \in G^0 : \exists \gamma \in u, x = s(\gamma)\} = s[u] = u^{-1}u$$

is Borel being 1:1 image of a Borel set. Moreover |Gx| = m iff  $\exists u_0, \ldots, u_{m-1} \in \mathcal{C}$  pairwise distinct such that  $x \in u_i u_i^{-1}$  for  $i \in m$  and  $\forall w \in \mathcal{C}$  if  $x \in ww^{-1}$  then  $w = u_i$  for some  $i \in m$ .

Let us say that a Borel bisection u is full if  $uu^{-1} = u^{-1}u = G^0$ . It is clear from Proposition 1.6.5 that if G is a countable Borel groupoid, then there is a partition of G into full Borel bisections.

#### 1.6.7 A Polish topology on countable Borel groupoids

In this subsection we observe that any countable Borel groupoid is Borel isomorphic to a regular zero-dimensional étale Polish groupoid. Suppose that G is a countable Borel groupoid. Pick a countable partition C of G into full Borel bisections and consider the smallest inverse subsemigroup of T with the property that  $u \cap v \in T$  whenever  $u, v \in T$ . Observe that T is countable. By [73, Exercise 13.5] there is a zero-dimensional Polish topology  $\tau^0$  on  $G^0$  generating the Borel structure on  $G^0$  such that  $u^{-1}u$  is clopen for every  $u \in T$ . Consider the standard action  $\theta$  of T on  $(G^0, \tau^0)$  and observe that it satisfies the condition of Proposition 1.6.2. Therefore the associated groupoid of germs  $\mathcal{G}(\theta, T, G^0)$  is an étale zero-dimensional regular Polish groupoid. Arguing as in the proof of [31, Proposition 5.4] one can verify that the function  $\phi$  from G to  $\mathcal{G}(\theta, T, G^0)$  sending  $\gamma$  to  $[c, s(\gamma)]$  where cis the only element of C such that  $\gamma \in C$  is a well defined Borel isomorphism of countable Borel groupoids.

#### **1.6.8** Treeable Borel groupoids

Suppose that G is a countable Borel groupoid. A graphing Q of G is a Borel subset Q of  $G \setminus G^0$  such that  $Q = Q^{-1}$  and  $\bigcup_{n \in \mathbb{N}} Q^n = G$ , where  $Q^0 = G^0$ . Suppose that Q is a graphing of G. Define  $P^*(Q)$  to be the set of finite nonempty sequences  $(\gamma_i)_{i \in n+1}$  in Q such that  $r(\gamma_{i+1}) = s(\gamma_i)$  and  $\gamma_{i+1} \neq \gamma_i^{-1}$  for  $i \in n$ . For  $(\gamma_i)_{i \in n+1}$  in  $P^*(Q)$  one can define

$$\prod_{i\in n+1}\gamma_i$$

to be the product  $\gamma_n \gamma_{1n-2} \cdots \gamma_1 \gamma_0$  in G. We say that Q is a *treeing* if for every  $(\gamma_i)_{i \in n+1} \in P^*(Q)$ ,

$$\prod_{i\in n+1}\gamma_i\notin Q^0$$

or, equivalently, for every  $\gamma \in G \setminus G^0$  there is exactly one element  $(\gamma_i)_{i \in n+1}$  of  $P^*(Q)$  such that  $\prod_{i \in n+1} \gamma_i = \gamma$ . A countable Borel groupoid is *treeable* when it admits a treeing [3, Section 8].

It is not difficult to verify that a principal countable Borel groupoid is treeable precisely when it is treeable as an equivalence relation. A countable group is treeable as groupoid if and only if it is a free group.

In the following if Q is a treeing of G we denote by P(Q) the union of  $P^*(Q)$  and  $\{\emptyset\}$ . In analogy with free groups, if  $(\gamma_n, \ldots, \gamma_0) \in P(Q)$  we say that  $\gamma_n \cdots \gamma_0$  is a reduced word, and that the length  $l(\gamma_n \cdots \gamma_0)$  of  $\gamma_n \cdots \gamma_0$  is n + 1.

**Proposition 1.6.6.** Suppose that G is a countable Borel groupoid. If there is a Borel complete section A for  $E_G$  such that  $G_{|A}$  is treeable, then G is treeable.

Proof. Pick a Borel function  $f: G^0 \to G$  such that f(a) = a for  $a \in A$ , s(f(x)) = x and  $r(f(x)) \in A$  for  $x \in G^0$ . Suppose that  $Q_A$  is a treeing for  $G_{|A}$ . Observe that  $Q_A \cup f[G^0 \setminus A]$  is a treeing for G.

We want to show that Borel subgroupoids of treeable groupoid are treeable. A particular case of this statement is that a subgroup of a countable free group is free, which is the well known *Nielsen-Schreier theorem*. The strategy of our proof will be a Borel version for groupoids of Schreier's proof of the Nielsen-Schreier theorem.

Suppose that G is a treeable groupoid with no elements of order 2, and H is a Borel subgroupoid of G. In the rest of the subsection we will show that H is treeable. Denote by  $\sim_H$  the equivalence relation  $\gamma \sim_H \rho$  iff  $\gamma H = \rho H$ . Suppose that Q is a treeing for

G. Since G has no elements of order 2 we can write  $Q = Q^+ \cup Q^-$  where  $Q^+$  and  $Q^-$  are disjoint and  $Q^+ = (Q^-)^{-1}$ . A Borel transversal U for  $\sim_H$  is Schreier if  $\gamma_n \cdots \gamma_0 \in T$  implies  $\gamma_k \cdots \gamma_0 \in T$  for  $k \in n$ . We want to show that there is a Schreier Borel transversal for H.

Suppose that  $(V_n)_{n \in \mathbb{N}}$  is a partition of  $G \setminus G^0$  into full Borel bisections. If  $\gamma_n \cdots \gamma_0$  and  $\gamma'_m \cdots \gamma'_0$  are reduced words with  $r(\gamma_n) = r(\gamma'_m) = x$ , set

$$\gamma_n \cdots \gamma_0 <_x \gamma'_m \cdots \gamma'_0$$

iff n < m, or n = m and for some  $k \in n$ ,  $\gamma_i = \gamma'_i$  for  $i \in k$  and for some  $N \in \mathbb{N}$ ,  $\gamma_k \in V_N$ while  $\gamma'_k \notin V_n$  for any  $n \leq N$ . Define also

$$x <_x \gamma_n \cdots \gamma_0.$$

Observe that  $\langle x \rangle$  is a Borel order of xG with minimum x, and the function  $x \mapsto \langle x \rangle$  is Borel. Define now for  $\gamma \in G$ ,  $\overline{\gamma}$  to be the  $\langle r(\gamma) \rangle$ -least element of  $\gamma H$ . Thus  $\overline{\gamma} \in \gamma H$  and hence  $\overline{\gamma}^{-1}\gamma \in H$ . Consider  $U = \{\overline{\gamma}^{-1}\gamma : \gamma \in G\}$  and observe that, since x is the  $\langle x \rangle$ -minimum element of xG,  $U \cap H \subset H^0$ . Arguing as in [64, Section 2.3] one can show that U is a Schreier transversal for  $\sim_H$ . Define then

$$A = \left\{ \overline{\gamma u}^{-1} \gamma u : u \in U, \, \gamma \in Q \right\} \subset H.$$

The same proof as Lemma 3 in [64, Section 3.3] shows that  $\bigcup_{n \in \mathbb{N}} A^n = H$ . Define now

$$B = \left\{ \overline{\gamma u}^{-1} \gamma u : u \in U, \, \gamma \in Q^+, \, \text{and} \, \gamma u \notin U \right\}.$$

The same proof as Lemma 4 in [64, Section 3.4] shows that

$$B^{-1} = \left\{ \overline{\gamma u}^{-1} \gamma u : u \in U, \, \gamma \in Q^{-}, \, \text{and} \, \gamma u \notin U \right\},\$$

and  $A \setminus H^0$  is the disjoint union of B and  $B^{-1}$ . Finally one can show that  $A \setminus H^0$  is a treeing for G as in [64, Section 3.6]. The proof is the same as the proof of Theorem 1 in [64, Section 3.6]. The fundamental lemma is the following:

**Lemma 1.6.7.** Suppose that  $b = \overline{u\gamma}^{-1}u\gamma \in A \setminus H^0$  and  $b' = \overline{v\rho}^{-1}v\rho \in A \setminus H^0$ . The product  $\rho v \overline{\gamma u}^{-1}\gamma$  is equal to a reduced word  $\rho w\gamma$  for some  $w \in G$ , unless  $v = \overline{\gamma u}$  and  $\rho = \gamma^{-1}$ , in which case

$$u = \overline{\gamma^{-1} \overline{\gamma u}} = \overline{\rho v}$$

 $b' = b^{-1}$ .

The proof of Lemma 1.6.7 is analogous to the proof of Lemma 5 in [64, Section 3.5].

# 1.7 Functorial Borel complexity and treeable equivalence relations

#### 1.7.1 The lifting property

**Definition 1.7.1.** Suppose that G is a Polish groupoid. We say that G has the *lifting* property if the following holds: For any Polish groupoid H such that  $E_H$  is Borel, and any Borel function  $f: G^0 \to H^0$  such that  $f(x)E_Hf(x')$  whenever  $xE_Gx'$ , there is a Borel functor  $F: G \to H$  that extends f.

**Remark 1.7.2.** If  $E_G$  has the lifting property (as a principal groupoid), then G has the lifting property.

**Proposition 1.7.3.** A treeable countable Borel groupoid with no elements of order 2 has the lifting property.

Proof. Suppose that G is a treeable countable Borel groupoid with no elements of order 2, H is a Polish groupoid such that  $E_H$  is Borel, and  $f: G^0 \to H^0$  is a Borel function such that  $f(x)E_Gf(x')$  whenever  $xE_Gx'$ . Suppose that Q is a treeing for G. Write  $Q = Q^+ \cup Q^-$  where  $Q^+ = (Q^-)^{-1}$  and  $Q^+$  and  $Q^-$  are disjoint. Since  $E_H$  is Borel, then map  $(x, y) \mapsto xHy$ from  $E_G$  to  $F(H) \setminus \{\emptyset\}$  is Borel by Theorem 1.4.4. Fix a Borel map  $\sigma : F(H) \setminus \{\emptyset\} \to H$ such that  $\sigma(A) \in A$  for every  $A \in F(H) \setminus \{\emptyset\}$ . Define

- F(x) = f(x) for  $x \in G^0$ ,
- $F(\gamma) = \sigma \left( f(r(\gamma)) H f(s(\gamma)) \right)$  for  $\gamma \in Q^+$ ,
- $F(\gamma) = F(\gamma^{-1})^{-1}$  for  $\gamma \in (Q^+)^{-1}$ , and
- $F(\gamma_n \cdots \gamma_0) = F(\gamma_n) \cdots F(\gamma_0)$  if  $\gamma_n \cdots \gamma_0 \in G \setminus G^0$  is a reduced word.

It is immediate to check that F is a Borel functor such that  $F_{|G^0} = f$ .

**Proposition 1.7.4.** If G is a Polish groupoid and  $A \subset G^0$  is a Borel complete section for  $E_G$  such that  $G_{|A}$  has the lifting property and there is a Borel map  $\phi : G^0 \to G$  such that  $s(\phi(x)) = x$  and  $r(\phi(x)) \in A$  for every  $x \in G^0$ , then G has the lifting property.

and

Proof. Without loss of generality we can assume that  $\phi(x) = x$  for  $x \in A$ . Define  $y(x) = r(\phi(x))$  for  $x \in G^0$ . Suppose that  $f: G^0 \to H^0$  is a Borel function such that  $f(x)E_Hf(x')$  whenever  $xE_Gx$ . Since  $G_{|A}$  has the lifting property there is a Borel functor  $F: G_{|A} \to H$  such that  $F_{|A} = f_{|A}$ . Define  $h(x) = \sigma(f(y(x))Hf(x))$ . Define now for  $\rho \in G$  such that  $s(\rho) = x$  and  $r(\rho) = y$ 

$$F(\rho) = h(y)^{-1} F(\phi(y)\rho\phi(x)^{-1}) h(x)$$

and observe that F is a Borel functor such that  $F_{|G^0} = f$ .

**Theorem 1.7.5.** Suppose that G is a Polish groupoid. If  $E_G$  is essentially treeable, then  $E_G$  has the lifting property.

*Proof.* Observe that the assignment  $[x]_{E_G} \mapsto I_{[x]_{E_G}}$ , where

$$A \in I_{[x]_{E_G}} \Leftrightarrow \{\gamma \in xG : s(\gamma) \in A\}$$
 is measured by the second sec

is a Borel ccc assignment of  $\sigma$ -ideals in the sense of [72, page 285]; see Subsection 1.1.10. It follows from [72, Theorem 1.5] together with the fact that  $E_G$  is essentially treeable that there is a countable Borel subset A of  $G^0$  meeting every orbit in a countable nonempty set. Thus  $(E_G)_{|A}$  is treeable equivalence relation. In particular by Proposition 1.7.3 the equivalence relation  $(E_G)_{|A}$  has the lifting property. Therefore  $G_{|A}$  has the lifting property. Since  $(E_G)_{|A}$  is countable one can find a Borel map  $p: X \to A$  such that  $(x, p(x)) \in E_G$ for every  $x \in X$  and p(x) = x for  $x \in A$ . It follows from Proposition 1.7.4 that  $E_G$  has the lifting property.

**Corollary 1.7.6.** Suppose that G and H are Polish groupoids. If  $E_G$  is essentially treeable, and  $E_H$  is Borel, then  $G \leq_B H$  if and only if  $E_G \leq E_H$ .

**Proposition 1.7.7.** Suppose that G is a Polish groupoid. If  $E_G$  is essentially countable, then there is an invariant dense  $G_{\delta}$  set  $C \subset G^0$  such that  $(E_G)_{|C}$  is essentially hyperfinite.

*Proof.* By [60, Theorem 6.2] there is a comeager and invariant subset  $C_0$  of  $G^0$  such that  $(E_G)_{|C_0}$  is essentially hyperfinite. Pick a dense  $G_{\delta}$  subset  $C_1$  of  $C_0$  and then define

$$C = \{x \in X : \forall^* \gamma \in Gx, \, \gamma x \in C_1\}.$$

The properties of the Vaught transform together with Lemma 1.1.9 imply that C is an invariant dense  $G_{\delta}$  set contained in  $C_0$ . In particular  $(E_G)_{|C}$  is essentially hyperfinite.  $\Box$ 

**Corollary 1.7.8.** Suppose that G is a Polish groupoid such that  $E_G$  is essentially countable. There is an invariant dense  $G_{\delta}$  subset C of  $G^0$  with the following property: For any Polish groupoid H such that  $E_G \leq_B E_H$  and  $E_H$  is Borel,  $G_{|C} \leq_B H$ .

#### 1.7.2 The cocycle property

**Definition 1.7.9.** An analytic groupoid G has the *cocycle property* if there is a Borel functor  $F: E_G \to G$  such that F(x, x) = x for every  $x \in G^0$ .

It is immediate to verify that a Polish group action  $G \curvearrowright X$  has the cocycle property as defined in [60] if and only if the action groupoid  $G \ltimes X$  has the cocycle property as in Definition 1.7.9. The proof of the following proposition is essentially the same of the proof of the implication (ii) $\Rightarrow$ (iii) in [63, Theorem 3.7], and it is presented for convenience of the reader.

**Proposition 1.7.10.** Suppose that G is a countable Borel groupoid, and X a Borel G-space. If  $G \ltimes X$  has the cocycle property, then there is a free Borel G-space Y such that  $E_G^Y \sim_B E_G^X$ . Moreover if G is treeable then  $E_G^X$  is treeable.

*Proof.* Since  $G \ltimes X$  has the cocycle property there is a Borel functor

$$F: E_G^X \to G$$

such that s(F(x,y)) = p(y) and F(x,y)y = x. Consider the equivalence relation  $\sim$  on  $G \ltimes X$  defined by  $(\gamma, x) \sim (\rho, y)$  iff  $(x, y) \in E_G^X$  and  $\gamma F(x, y) = \rho$ . Clearly  $\sim$  is Borel. We now show that it has a Borel selector. Observe that the range H of F is a Borel subgroupoid of G (since F is countable to one). By Proposition 1.4.1 there is a Borel selector  $t: G \to G$  for the equivalence relation  $\gamma \sim_H \gamma'$  iff  $\gamma H = \gamma' H$ . Observe that if  $(\gamma, x) \sim (\rho, y)$  then  $\gamma H = \rho H$  and hence  $t(\gamma) = t(\rho)$ . Moreover there is a unique element  $x_0$  of X such that  $(t(\gamma), x_0) \sim (\gamma, x)$ . Define  $S(\gamma, x) = (t(\gamma), x_0)$  and observe that S is a Borel selector for the equivalence relation  $\sim$ . Define Y to be the quotient of  $G \ltimes X$  by  $\sim$ . Define now the Borel action of G on Y by  $p[\gamma, x] = r(\gamma)$  and  $\rho[\gamma, x] = [\rho\gamma, x]$  for  $\rho \in Gr(\gamma)$ . It is easy to verify that such an action is free, and  $[\gamma, x] E_G^Y[\rho, y]$  iff  $x E_G^X y$ . Let us now observe that  $E_G^X \sim_B E_G^Y$ . If  $q: X \to G$  is a Borel map such that s(q(x)) = p(x) for every  $x \in X$ , then the map  $x \mapsto [q(x), x]$  is a Borel reduction from  $E_G^X$  to  $E_G^X$ . Suppose finally that G is treeable with treeing Q. We want to show that  $E_G^X$  is treeable. Since  $E_G^X \sim_B E_G^Y$ , it is enough to show that  $E_G^Y$  is treeable. Fix an equivalence class  $[[\gamma, x]]_F$  of  $E_G^Y$ .

the map from  $[[\gamma, x]]_{E_G^Y}$  to Gp(x) defined by  $[\rho, y] \mapsto \rho F(y, x)$  is bijective. One can then consider the treeing

$$\{[\rho, y] \in Y : \rho F(y, x) \in Q\}$$

for  $E_G^Y$ .

Lemma 1.7.11 can be proved similarly as Proposition 1.7.4.

**Lemma 1.7.11.** Suppose that G is a countable Borel groupoid action, and  $A \subset G^0$  is a Borel complete section for  $E_G$ . If there is a Borel function  $\psi : (E_G)_{|A|} \to G$  such that  $s(\psi(x,y)) = y$  and  $r(\psi(x,y)) = x$ , then G has the cocycle property.

**Lemma 1.7.12.** Suppose that G is a countable Borel groupoid. If  $E_G \leq_B G$ , then there is an invariant Borel subset Y of  $G^0$  such that  $G_{|Y}$  has the cocycle property and  $(E_G)_{|Y} \sim_B E_G$ 

Proof. Since  $E_G \leq_B G$  there is a Borel functor  $F : E_G \to G$ . Define  $A \subset G^0$  to be image of X under F. Since  $F_{|X}$  is countable-to-one, A is a Borel subset of  $G^0$ ; see [73, Theorem 18.10]. By [73, Exercise 18.14] there is a Borel function  $g : A \to X$  such that  $(f \circ g)(y) = y$ for every  $y \in A$ . Define now Y to be the union of the orbits of G that meet A. Clearly  $E_G \sim_B E_{G_{|Y}}$ . The map

$$\begin{aligned} (E_G)_{|A} &\to G \\ (x,y) &\mapsto F(g(x),g(y)) \end{aligned}$$

together with Lemma 1.7.11 imply that  $G_{|Y}$  has the cocycle property.

#### 1.7.3 Free actions of treeable groupoids

We want to show that, if G is a treeable groupoid, and  $G \curvearrowright X$  is a free Borel action of G, then the associated orbit equivalence relation is treeable. This will follow from a more general result about  $\mathcal{L}$ -structured equivalence relations.

Suppose that  $\mathcal{L} = \{R_n : n \in \mathbb{N}\}$  is a countable relational language in first order logic, where  $R_n$  has arity  $k_n \in \mathbb{N}$ . Suppose that E is a countable Borel equivalence relation on a standard Borel space X. According to [63, Definition 2.17] the equivalence relation E is  $\mathcal{L}$ -structured if there are Borel relations  $R_n^E \subset X^{k_n}$  such that, for any  $k \in \mathbb{N}$  and  $x_1, \ldots, x_{k_n} \in X, x_1, \ldots, x_{k_n}$  belong to the same E-class whenever  $(x_1, \ldots, x_{k_n}) \in R_n^E$ . In particular every E-class [x] is the universe of an  $\mathcal{L}$ -structure

$$\left\langle \left[x\right], \left(\left[x\right]^{k_n} \cap R_n^E\right)_{n \in \mathbb{N}} \right\rangle.$$

Similarly, if X is a standard Borel space, then standard Borel bundle  $\mathcal{A}$  of countable  $\mathcal{L}$ structures over X is a standard Borel space  $\mathcal{A}$  fibred over X with countable fibers  $(A_x)_{x \in X}$ ,
endowed with Borel subsets  $R_n^{\mathcal{A}} \subset \mathcal{A}^{k_n}$  such that, for any  $k \in \mathbb{N}$  and  $a_1, \ldots, a_{k_n} \in \mathcal{A}$ ,  $a_1, \ldots, a_{k_n}$  belong to the same fiber over X whenever  $(a_1, \ldots, a_{k_n}) \in R_n^{\mathcal{A}}$ . Suppose that G
is a Polish groupoid, and  $\mathcal{A}$  is a standard Borel of  $\mathcal{L}$ -structures over  $G^0$ . A Borel action  $G \curvearrowright \mathcal{A}$  is a Borel map  $(\gamma, a) \mapsto \gamma a$  defined for  $(\gamma, a) \in \mathcal{A} \times G$  such that  $a \in A_{s(\gamma)}$ ,
and  $(a_1, \ldots, a_{k_n}) \in R_n^{\mathcal{A}}$  if and only if  $(\gamma a_1, \ldots, \gamma a_{k_n}) \in R_n^{\mathcal{A}}$  for every  $n \in \mathbb{N}, \gamma \in G$ , and  $a_1, \ldots, a_{k_n} \in A_{s(\gamma)}$ . The proof of the following theorem is very similar to the argument at
the beginning of Section 3.2 in [63], and it is reproduced for convenience of the reader.

**Theorem 1.7.13.** Suppose that G is a Polish groupoid such that there is a standard Borel bundle  $\mathcal{A}$  of countable  $\mathcal{L}$ -structures and a Borel action  $G \curvearrowright \mathcal{A}$  such that the corresponding orbit equivalence relation  $E_G^{\mathcal{A}}$  is smooth, and for every  $a \in \mathcal{A}$  the stabilizer  $G_a = \{\gamma \in Gp_{\mathcal{A}}(a) : \gamma a = a\}$  is a compact subset of G. If X is a standard Borel space, and  $G \curvearrowright X$  is a free Borel action of G on X, then there is an  $\mathcal{L}$ -structured countable Borel equivalence relation E such that  $E \sim_B E_G^X$ , and moreover every class of E is isomorphic to some fiber of  $\mathcal{A}$ .

*Proof.* By Corollary 1.1.18 there is a Borel selector t for  $E_G^{\mathcal{A}}$ . Moreover by Theorem 1.3.1 we can assume without loss of generality that the action  $G \curvearrowright X$  is continuous. Define

$$\mathcal{A} * X = \left\{ (x, a) \in X \times A : a \in A_{p(x)} \right\}$$

and the action  $G \curvearrowright (X * A)$  by  $\gamma(x, a) = (\gamma x, \gamma a)$ . Observe that such an action is free and in particular the associated orbit equivalence relation  $\sim$  is Borel. We now show that  $\sim$  has a Borel selector. Suppose that  $(x, a) \in X * A$ . Observe that if  $(x, a) \sim (x, b)$  then  $aE_G^A b$  and hence t(a) = t(b). Therefore t(a) depends only on the  $\sim$ -class [x, a] of (x, a). Observe that

$$G_{t(a)}x = \{y \in X : (y, t(a)) \in [x, a]\}$$

is a compact subset of X. Denote by  $\sigma : F(X) \setminus \{\emptyset\} \to X$  a Borel function such that  $\sigma(A) \in A$  for every  $A \in F(X) \setminus \{\emptyset\}$ . Define  $S(x,a) = (\sigma(G_{t(a)}x), t(a))$ , and observe that S is a Borel selector for  $\sim$ . Define the standard Borel space  $Y = (X * A) / \sim$  and the countable Borel equivalence relation E on Y by [x, a] E[y, b] iff  $x E_G^X y$ . We now define for every E-class  $C = [[x, a]]_E$  an  $\mathcal{L}$ -structure on C. Fix  $n \in \mathbb{N}$  and suppose that  $[\gamma_i x, a_i]$  for  $i \in k_n$  are elements of C, where  $\gamma_i \in Gp(x)$  for  $i \in k_n$ . Set  $([\gamma_i x, a_i])_{i \in k_n} \in R_n^C$  iff  $(\gamma_i^{-1}a_i)_{i \in k_n} \in R_n^A$ . Using the fact that G acts by  $\mathcal{L}$ -isomorphisms one can verify that

this does not depend on the choice of  $[x, a] \in C$ . Define now  $([x_i, a_i])_{i \in k_n} \in R_n^E$  if and only if  $[x_0, a_0] E[x_1, a_1] E \cdots E[x_{k_n-1}, a_{k_n-1}]$  and  $([x_i, a_i])_{i \in k_n} \in R_n^C$  where  $C = [[x, a]]_E$ . This defines Borel relations  $R_n^E$  on E that make  $E \mathcal{L}$ -structured. Moreover the Borel map  $f : C \to \mathcal{A}$  defined by  $f[\gamma x, a] = \gamma^{-1}a$ , where C is the class of [x, a], shows that the  $\mathcal{L}$ -structure

$$\left\langle C, \left(R_n \cap C^{k_n}\right)_{n \in \mathbb{N}} \right\rangle$$

is isomorphic to  $A_{p(x)}$ . Finally we observe now that  $E \sim E_G^X$ . If  $q: G^0 \to \mathcal{A}$  such that q(x) belongs to the fiber  $A_x$  for every  $x \in G^0$ , then the map  $x \mapsto [x, q(p(x))]$  is a Borel reduction from  $E_G^X$  to E. Conversely the map  $[x, a] \mapsto x^*$  where  $[x^*, a^*] = S([x, a])$  witnesses that E is Borel reducible to  $E_G^X$ .

Let us now consider the particular case of Theorem 1.7.13 when  $\mathcal{L}$  is the language with a single binary relation. Assume further that G is a treeable Borel groupoid. A standard Borel bundle of trees over X is a standard Borel bundle  $(A_x)_{x \in X}$  of countable  $\mathcal{L}$ -structures such that  $A_x$  is a tree for every  $x \in X$ . A treeing of G defines on G a structure of standard Borel bundle of trees over G. Moreover the action of G on itself by left translation is compatible with such a bundle of trees structure, and has a smooth orbit equivalence relation. Therefore by Theorem 1.7.13 E is treeable. This concludes the proof of the following corollary.

**Corollary 1.7.14.** If G is a countable treeable groupoid, and  $G \curvearrowright X$  is a free Borel action, then the orbit equivalence relation  $E_G^X$  is treeable.

#### 1.7.4 Characterizing treeable equivalence relations

Denote by  $\mathbb{F}_{\infty}$  the free countable group on infinitely many generators. The following result subsumes [63, Theorem 3.7].

**Theorem 1.7.15.** Suppose that E is a countable Borel equivalence relation on a standard Borel space X. The following statements are equivalent:

- 1. E is treeable;
- 2. E has the lifting property;
- 3. For every countable Borel groupoid G and Borel action  $G \curvearrowright X$  such that  $E_G^X = E$ , the groupoid  $G \ltimes X$  has the cocycle property;
- 4. For every Borel action  $\mathbb{F}_{\infty} \curvearrowright X$  such that  $E_{\mathbb{F}_{\infty}}^X = E, \ E_{\mathbb{F}_{\infty}}^X \leq_B \mathbb{F}_{\infty} \ltimes X;$

- 5. For every countable Borel groupoid G and Borel action  $G \curvearrowright X$  such that  $E_G^X = E$ , there is a free Borel action  $G \curvearrowright Y$  such that  $E_G^Y \sim E_G^X$ ;
- 6. For every countable Borel groupoid G and Borel action  $G \curvearrowright X$  such that  $E \subset E_G^X$ there is a free Borel action  $G \curvearrowright Y$  such that  $E \sqsubseteq_B E_G^Y$ .
- *Proof.*  $(1) \Rightarrow (2)$  It follows from Proposition 1.7.3.
- (2) $\Rightarrow$ (3) It follows form the fact that if  $E_G^X$  has the lifting property, then  $G \ltimes X$  has the cocycle property.
- $(3) \Rightarrow (4)$  Obvious.
- (4) $\Rightarrow$ (1) It follows from Lemma 1.7.12 and Proposition 1.7.10.
- $(3) \Rightarrow (5)$  This follows from Proposition 1.7.10.
- (5) $\Rightarrow$ (1) Consider an action  $\mathbb{F}_{\infty} \curvearrowright X$  such that  $E = E_{\mathbb{F}_{\infty}}^X$  and then apply (5) and Corollary 1.7.14.
- (2) $\Rightarrow$ (6) Since *E* has the lifting property, there is a Borel function  $F: E \to G$  such that s(F(x,y)) = p(y) and F(x,y)y = x for every  $(x,y) \in E$ . Consider on  $G \ltimes X$  the equivalence relation  $(\gamma, x) \sim (\rho, y)$  iff xEy and  $\rho = \gamma F(x,y)$ . Proceeding as in the proof of Proposition 1.7.10 one can show that  $\sim$  has a Borel selector. Thus the quotient *Y* of  $G \ltimes X$  by  $\sim$  is standard. Define the Borel action  $G \curvearrowright Y$  by  $p[\gamma, x] = r(\gamma)$  and  $\rho[\gamma, x] = [\rho\gamma, x]$ . As in the proof of Proposition 1.7.10 one can show that  $\sum p(x) = p(y)$  and  $p[\gamma, x] = p(\gamma, x]$ . As in the proof of Proposition 1.7.10 one can show that  $p[\gamma, x] = r(\gamma)$  and  $p[\gamma, x] = [\rho\gamma, x]$ . As in the proof of Proposition 1.7.10 one can show that such an action is free. Moreover the map  $x \mapsto [p(x), x]$  is an injective Borel reduction from *E* to  $E_G^Y$ .
- (6)⇒(1) It follows from Corollary 1.7.14 together with the fact that a subrelation of a treeable equivalence relation is treeable [63, Proposition 3.3]; see also Subsection 1.6.8.

The following corollary is a direct consequence of the implication  $(4) \Rightarrow (1)$  in Theorem 1.7.15.

**Corollary 1.7.16.** For any nontreeable equivalence relation E there are countable Borel groupoids G and H that have E as orbit equivalence relation such that G is not Borel reducible to H. Moreover one can take G = E and H to be the action groupoid of an action of  $\mathbb{F}_{\infty}$ .

Corollary 1.7.16 can be interpreted as asserting that functorial Borel complexity provides a finer distinction between the complexity of classification problems in mathematics than the traditional notion of Borel complexity for equivalence relations

# Chapter 2

# Unitary equivalence of automorphisms of C\*-algebras

If A is a separable C\*-algebra, the group Aut(A) of automorphisms of A is a Polish group with respect to the topology of pointwise norm convergence. An automorphism of A is called (multiplier) *inner* if it is induced by the action by conjugation of a unitary element of the multiplier algebra M(A) of A. Inner automorphisms form a Borel normal subgroup Inn(A)of the group of automorphisms of A. The relation of *unitary equivalence* of automorphisms of A is the coset equivalence relation on Aut(A) determined by Inn(A). (The reader can find more background on C<sup>\*</sup>-algebras in Section 2.1.) The main result presented in this chapter asserts that if A does not have continuous trace, then it is not possible to effectively classify the automorphisms of A up to unitary equivalence using countable structures as invariants; in particular this rules out classification by K-theoretic invariants. (The Ktheoretic invariants of  $C^*$ -algebras were shown to be computable by a Borel function in [37, Theorem 3.3]. Even though [37, Theorem 3.3] does not consider the K-theory of \*homomorphisms, it is not difficult to verify that the proof can be adapted to show that the computation of K-theory of \*-homomorphisms is given by a *Borel functor*. The main ingredient of the proof is the fact that one can enumerate in a Borel fashion dense sequences of projections and of unitary elements of the algebra and of all its amplifications [38, Lemma 3.13].) In the course of the proof of the main result we will show that the existence of an outer derivation on a C<sup>\*</sup>-algebra A is equivalent to a seemingly stronger statement, that we will refer to as Property AEP (see Definition 2.3.4), implying in particular the existence of an outer derivable automorphism of A.

In the last decade a number of natural equivalence relations arising in different areas of

mathematics have been shown to be not classifiable by countable structures. For example the type of invariants that appear in the spectral theorem for normal operators transcend countable structures by a result of Kechris and Sofronidis [74, Theorem 1]. The theory of turbulence, developed by Greg Hjorth in the second half of the 1990s, plays a key role in the proof of this and of many other analogous results.

Turbulence is a dynamic condition on a continuous action of a Polish group on a Polish space, implying that the associated orbit equivalence relation is not classifiable by countable structures. Many nonclassifiability results were established directly or indirectly using this criterion. For instance Hjorth showed in [59, Section 4.3] that the orbit equivalence relation of a turbulent Polish group action is Borel reducible to the relation of homeomorphism of compact spaces, which in turn is reducible to the relation of isomorphism of separable simple nuclear unital C\*-algebras by a result of Farah-Toms-Törnquist [38, Corollary 5.2]. As a consequence these equivalence relations are not classifiable by countable structures.

In this chapter, we use Hjorth's theory of turbulence to prove the following theorem. (See Definition 2.1.1 for the notion of continuous trace C\*-algebra.)

# **Theorem 2.0.17.** If A is a separable $C^*$ -algebra that does not have continuous trace, then the automorphisms of A are not classifiable by countable structures up to unitary equivalence.

Theorem 2.0.17 strengthens [111, Theorem 3.1], where the automorphisms of A are shown to be not concretely classifiable under the same assumptions on the C\*-algebra A. We will in fact show that the same conclusion holds even if one only considers the subgroup consisting of approximately inner automorphisms of A, i.e. pointwise limits of inner automorphisms.

A particular implication of Theorem 2.0.17 is that it is not possible to classify the automorphisms of any separable C\*-algebra that does not have continuous trace up to unitary equivalence by Borel-computable K-theoretic invariants. This should be compared with the classification results of (sufficiently outer) automorphisms up to other natural equivalence relations, such as outer conjugacy; see [103, Section 3]. Nakamura showed in [103, Theorem 9] that aperiodic automorphisms of Kirchberg algebras are classified by their KK-classes up to outer conjugacy. Theorem 1.4 of [81] asserts that there is only one outer conjugacy class of uniformly aperiodic automorphisms of UHF algebras. These results were more recently generalized and expanded to classification of actions of  $\mathbb{Z}^2$  and  $\mathbb{Z}^n$  up to outer conjugacy (see [97], [96], [68], and [98]).

Phillips and Raeburn obtained in [113] a cohomological classification of automorphisms of a  $C^*$ -algebra with continuous trace up to unitary equivalence. Such classification implies that if A has continuous trace and the spectrum of A is homotopy equivalent to a compact space, then the normal subgroup Inn(A) of inner automorphisms is closed in Aut(A); see [116, Theorem 0.8]. In particular (cf. [11, Corollary II.6.5.7]) this conclusion holds when A is unital and has continuous trace. It follows from a standard result in descriptive set theory –see [44, Exercise 6.4.4]– that the automorphisms of A are concretely classifiable up to unitary equivalence if and only if Inn(A) is a closed subgroup of Aut(A). Theorem 0.8 of [116] and Theorem 2.0.17 therefore imply the following dichotomy result:

**Theorem 2.0.18.** If A is a separable unital  $C^*$ -algebra, then the following statements are equivalent:

- 1. the automorphisms of A are concretely classifiable up to unitary equivalence;
- 2. the automorphisms of A are classifiable by countable structures up to unitary equivalence;
- 3. A has continuous trace.

More generally the same result holds if A is a separable C\*-algebra with (not necessarily Hausdorff) compact spectrum. Without this hypothesis the implication  $3 \Rightarrow 1$  of Theorem 2.0.18 does not hold, as pointed out in [116, Remark 0.9]. We do not know if the implication  $3 \Rightarrow 2$  holds for a not necessarily unital C\*-algebra A.

In particular Theorem 2.0.18 offers another characterization of unital C\*-algebras that have continuous trace, in addition to the classical Fell-Dixmier spectral condition (see [39], [21]) or the reformulation in terms of central sequences by Akemann and Pedersen; see [2, Theorem 2.4].

The dichotomy in the Borel complexity of the relation of unitary equivalence of automorphisms of a unital C\*-algebra expressed by Theorem 2.0.18 should be compared with the analogous phenomenon concerning the relation of unitary equivalence of irreducible representations of a C\*-algebra A. It is a classical result of Glimm from [49] that such a relation is smooth if and only if A is Type I. It was proved in [75] and, independently, in [33] that the irreducible representations of a C\*-algebra that is not Type I are in fact not classifiable by countable structures up to unitary equivalence.

The strategy of the proof of Theorem 2.0.17, summarized in Figure 1, is the following: We first introduce in Definition 2.3.4 and 2.3.9 properties AEP and AEP<sup>+</sup>, named after Akemann, Elliott, and Pedersen since they can be found *in nuce* in their works [2] and [28]. (The main result of [28] is a characterization of C<sup>\*</sup>-algebras with only inner derivations as



Figure 1:

direct sum of simple C\*-algebras and C\*-algebras with no nontrivial central sequence [28, Theorem 1]. Theorem 2.4 of [2] shows that a C\*-algebra is does not have any nontrivial central sequence if and only if it has continuous trace.) We then show in Proposition 2.3.11 that Property AEP<sup>+</sup> is stronger than Property AEP; moreover by Theorem 2.3.5 Property AEP is equivalent to the existence of an outer derivation, and by Lemma 2.3.3 it implies that the conclusion of Theorem 2.0.17 holds.

This concludes the proof under the assumption that the C\*-algebra A has an outer derivation. We then assume that A does not have continuous trace and has only inner derivations. Using the already mentioned characterization of C\*-algebras with only inner derivations from [28, Theorem 1] and the characterization of continuous trace C\*-algebras in terms of central sequences given in [2, Theorem 2.4], we infer that in this case A has a simple nonelementary direct summand. We then deduce in Proposition 2.4.5 that A contains a central sequence that is not strict-hypercentral. (A similar result was proved by Phillips in the unital case, cf. [112, Theorem 3.6].) The proof is finished by proving that the existence of a central sequence that is not strict-hypercentral implies that the conclusion of Theorem
2.0.17 holds. This is done in Proposition 2.4.6.

This chapter is organized as follows: Section 2.1 contains some background on C<sup>\*</sup>algebras and introduces the notations used in the rest of the chapter; Section 2.2 infers from Hjorth's theory of turbulence a criterion of nonclassifiability by countable structures (Criterion 2.2.3), to be applied in the proof of Theorem 2.0.17; Section 2.3 establishes Theorem 2.0.17 in the case of C<sup>\*</sup>-algebras with outer derivations, while Section 2.4 deals with the case of C<sup>\*</sup>-algebras with only inner derivations; Section 2.5 present a dichotomy result for derivations analogous to Theorem 2.0.18 (Theorem 2.5.1).

## 2.1 Some background notions on C\*-algebras

The multiplier algebra M(A) of a C\*-algebra A is the largest unital C\*-algebra containing A as an essential ideal; see [11, II.7.3]. It can be regarded as the noncommutative analog of the Stone-Čech compactification of a locally compact Hausdorff space. The strict topology on M(A) is the locally convex vector space topology on M(A) generated by the seminorms  $x \mapsto ||ax|| + ||xa||$  for  $a \in A$  [11, II.7.3.11]. A positive contraction  $b_0$  of A is strictly positive if

$$ab_0^{\frac{1}{n}} \to a$$

for every  $a \in A$  [11, II.4.2.1]. If  $b_0$  is any strictly positive contraction in A, then the strict topology on M(A) can be equivalently defined as the locally convex vector space topology on A generated by the single seminorm  $x \mapsto ||b_0x|| + ||xb_0||$ . The multiplier algebra of a separable C\*-algebra A is not norm separable (unless A is unital, in which case M(A)coincides with A). Nonetheless the *strict topology* of M(A) is Polish and induces a Polish group structure on the group U(A) of unitary elements of M(A). If u is a unitary multiplier of A, i.e. an element of U(A), then one can define as before the automorphism Ad(u) of A. An automorphism of A is called *inner* if it is of the form Ad(u) for some unitary multiplier u, and *outer* otherwise. Inner automorphisms of a separable C\*-algebra A form a Borel normal subgroup of Aut(A); see [111, Proposition 2.4]. Two automorphisms  $\alpha$  and  $\beta$  of Aare called *unitarily equivalent* if  $\alpha \circ \beta^{-1}$  is inner or, equivalently,

$$\alpha(x) = \beta(uxu^*)$$

for some unitary multiplier u and every  $x \in A$ . This defines a Borel equivalence relation on Aut(A).

A representation of A on a Hilbert space H is a \*-homomorphism from A to the C\*-

algebra B(H) of bounded linear operators on H; see [11, Definition II.6.1.1]. Two representations  $\pi$ ,  $\pi'$  of A on Hilbert spaces H, H' are *unitarily equivalent* if there is a surjective linear isometry  $U: H \to H'$  such that

$$U\pi(a) = \pi'(a)U$$

for every  $a \in A$ . A representation  $\pi$  of A on a Hilbert space H is called *irreducible* if there is no nontrivial closed subspace of H which is  $\pi(a)$ -invariant for every  $a \in A$ . The *spectrum*  $\hat{A}$  of a separable C\*-algebra A is the space of unitary equivalence classes of irreducible representations of A on a separable Hilbert space [109, Section 4.1]. This is canonically endowed with the *hull-kernel topology*, which is the topology having as open basis the collection of sets of the form

$$\mathcal{O}_I = \left\{ [\pi] \in \hat{A} : I \nsubseteq \operatorname{Ker}(\pi) \right\}$$

for some closed ideal I of A. In general this topology has very poor separation properties, and can even fail to be  $T_0$ . A closed ideal of A is *primitive* if it is the kernel of an irreducible representation of A. A C\*-algebra A is called *primitive* if  $\{0\}$  is a primitive ideal in A, i.e. A has a faithful irreducible representation. The *primitive spectrum*  $\check{A}$  of A is the space of primitive ideals of A endowed with the quotient topology from the canonical surjection

$$\hat{A} \rightarrow \check{A}$$
  
 $[\pi] \mapsto \operatorname{Ker}(\pi).$ 

An element x of a C\*-algebra A is *abelian* if the closure of  $x^*Ax$  in A is a commutative subalgebra.

**Definition 2.1.1.** A separable C\*-algebra A has *continuous trace* if it is generated by abelian elements, and the spectrum  $\hat{A}$  endowed with the hull-kernel topology is a Hausdorff space.

Equivalent reformulations of the notion of continuous trace C\*-algebras can be found in [11, Definition IV.1.4.12 and Proposition IV.1.4.19]. The class of C\*-algebras that do not have continuous trace is fairly large, and in particular includes all C\*-algebras that are not Type I. (A C\*-algebra A is Type I if every nonzero quotient of A contains a nonzero abelian element. Several equivalent characterizations of Type I C\*-algebras are listed in [11, IV.1.5.1].) More information about C\*-algebras with continuous trace can be found in the monograph [117]. In the rest of the chapter, we assume all  $C^*$ -algebras to be norm separable, apart from multiplier algebras and enveloping von Neumann algebras. If A is a C\*-algebra, then the universal representation  $\pi_u$  of A is the direct sum of all cyclic representations of A associated with states of A [109, 3.7.6, 3.7.8]. The enveloping von Neumann algebra of A is the closure of  $\pi_u[A]$  in the strong operator topology. It is a well known theorem (see [109, Proposition 3.7.8]) that the enveloping von Neumann algebra of A is isometrically isomorphic –as a Banach space– to the second dual of A. We will therefore denote in the following by  $A^{**}$  the enveloping von Neumann algebra of A. The  $\sigma$ -weak topology on  $A^{**}$ coincides with the weak\* topology of  $A^{**}$  regarded as the dual Banach space of  $A^*$ . The algebra A can be identified with a  $\sigma$ -weakly dense subalgebra of  $A^{**}$ . Moreover by [109, 3.12.3] we can identify the multiplier algebra M(A) of A with the idealizer of A inside  $A^{**}$ , i.e. the algebra of elements x such that  $xa \in A$  and  $ax \in A$  for every  $a \in A$ . Analogously, the unitization  $\tilde{A}$  of A [11, II.1.2] is identified with the subalgebra of M(A) generated by A and 1.

If x is a normal element of A, i.e. commuting with its adjoint, and f is a complexvalued continuous function defined on the spectrum of x, then f(x) denotes the element of  $\tilde{A}$  obtained from x and f using functional calculus (II.2 of [11] is a complete reference for the basic notions of spectral theory and continuous functional calculus in operator algebras). If x, y are element of a C\*-algebra, then [x, y] denotes their commutator xy - yx; moreover if S is a subset of a C\*-algebra A, then  $S' \cap A$  denotes the relative commutant of S in A; see [11, I.2.5.3]. The set N of natural numbers is supposed not to contain 0. Boldface letters **t** and **s** indicate sequences of real numbers whose n-th terms are  $t_n$  and  $s_n$  respectively. Analogously **x** stands for the sequence  $(x_n)_{n\in\mathbb{N}}$  of elements of a C\*-algebra A.

## 2.2 Nonclassifiability criteria

Recall that a subset A of a Polish space X has the Baire property [73, Definition 8.21] if its symmetric difference with some open set is meager. A function between Polish spaces is Baire measurable [73, Definition 8.37] if the inverse image of any open set has the Baire property. Observe that, in particular, any Borel function is Baire measurable. Suppose that E and R are equivalence relations on Polish spaces X and Y respectively. We say that E is generically R-ergodic if, for every Baire measurable function  $f: X \to Y$  such that f(x)Rf(y) whenever xEy, there is a comeager subset C of X such that f(x)Rf(y) for every  $x, y \in C$  [44, Definition 10.1.4]. Observe that if E is generically R-ergodic and no equivalence class of E is comeager then, in particular, E is not Borel reducible to R. One of the main tools in the study of Borel complexity of equivalence relations is Hjorth's theory of turbulence. A standard reference for this subject is [59]. Turbulence is a dynamical property of a continuous group action of a Polish group G on a Polish space X; see [59, Definition 3.13]. The main result about turbulent actions is the following result of Hjorth (Theorem 3.21 in [59]):

The orbit equivalence relation  $E_G^X$  associated with a turbulent action  $G \curvearrowright X$  of a Polish group G on a Polish space X is generically  $\simeq_{\mathcal{C}}$ -ergodic for every class  $\mathcal{C}$  of countable structures, where  $\simeq_{\mathcal{C}}$  denotes the relation of isomorphism for elements of  $\mathcal{C}$ . Since (by definition of turbulence)  $E_G^X$  has meager equivalence classes, it is in particular not classifiable by countable structures.

This result is valuable because it allows one to obtain several nonclassification results. In order to apply such result it will be useful to first state and prove the following to easy lemmas:

**Lemma 2.2.1.** Suppose that E, F, and R are equivalence relations on Polish spaces X, Y, and Z respectively, and that F is generically R-ergodic. If there is a comeager subset  $\tilde{C}$  of Y and a Baire measurable function  $f: \tilde{C} \to X$  such that:

- f(x)Ef(y) for any  $x, y \in \widetilde{C}$  such that xFy;
- f[C] is comeager in X for every comeager subset C of  $\widetilde{C}$ ;

then the relation E is generically R-ergodic as well.

Proof. Suppose that  $g: X \to Z$  is a Baire measurable function such that g(x)Rg(x') for any  $x, x' \in X$  such that xEx'. The composition  $g \circ f$  is a Baire measurable function from  $\tilde{C}$  to Z such that  $(f \circ g)(y)R(f \circ g)(y')$  for any  $y, y' \in \tilde{C}$  such that yFy'. Since  $\tilde{C}$  is comeager in Y, and F is generically R-ergodic, there is a comeager subset C of  $\tilde{C}$  such that  $(g \circ f)(y)R(g \circ f)(y')$  for every  $y, y' \in C$ . Therefore, f[C] is a comeager subset of X such that g(x)Rg(x') for every  $x, x' \in f[C]$ .

Observe that if f is continuous, open, and onto, then it will automatically satisfy the second condition of Lemma 2.2.1.

**Lemma 2.2.2.** Suppose that E and F are equivalence relations on Polish spaces X and Y respectively, and F is generically  $\simeq_{\mathcal{C}}$ -ergodic for every class  $\mathcal{C}$  of countable structures. If there is a Baire measurable function  $f: Y \to X$  such that

- f(x)Ef(y) whenever xFy, and
- no preimage of an E-class is comeager,

then the relation E is not classifiable by countable structures.

Proof. Suppose by contradiction that there is a class  $\mathcal{C}$  of countable structures and a Borel reduction  $g: X \to \mathcal{C}$  of E to  $\simeq_{\mathcal{C}}$ . The composition  $g \circ f: Y \to \mathcal{C}$  is a Baire measurable function from Y to  $\mathcal{C}$  such that  $(g \circ f)(y) \simeq_{\mathcal{C}} (g \circ f)(y')$  for any  $y, y' \in Y$  such that yFy'. Since F is generically  $\simeq_{\mathcal{C}}$ -ergodic, there is a comeager subset C of Y such that  $(g \circ f)(y) \simeq_{\mathcal{C}} (g \circ f)(y')$  for every  $y, y' \in C$ . Therefore, being g a reduction of E to  $\simeq_{\mathcal{C}}$ , f(y)Ef(y') for every  $y, y' \in C$ . This contradicts our assumptions.  $\Box$ 

Consider  $\mathbb{R}^{\mathbb{N}}$  as a Polish space with the product topology and  $\ell^1$  as a Polish group with its Banach space topology. The fact that the action of  $\ell^1$  on  $\mathbb{R}^{\mathbb{N}}$  by translation is turbulent is a particular case of [59, Proposition 3.25]. It then follows by Hjorth's turbulence theorem that the associated orbit equivalence relation  $E_{\mathbb{R}^{\mathbb{N}}}^{\ell^1}$  is generically  $\simeq_{\mathcal{C}}$ -ergodic for every class  $\mathcal{C}$  of countable structures. It is not difficult to see that the function  $f : (\mathbb{R} \setminus \{0\})^{\mathbb{N}} \to (0, 1)^{\mathbb{N}}$ defined by

$$f(\mathbf{t}) = \left(\frac{|t_n|}{|t_n|+1}\right)_{n \in \mathbb{N}}$$

satisfies both the first (being continuous, open, and onto) and the second condition of Lemma 2.2.1, where:

- F is the relation  $E_{\mathbb{R}^{\mathbb{N}}}^{\ell^1}$  of equivalence modulo  $\ell^1$  of sequences of real numbers;
- E is the relation  $E_{(0,1)^{\mathbb{N}}}^{\ell^1}$  of equivalence modulo  $\ell^1$  of sequences of real numbers between 0 and 1.

It follows that the latter relation is generically  $\simeq_{\mathcal{C}}$  -ergodic for every class  $\mathcal{C}$  of countable structures. Considering the particular case of Lemma 2.2.2 when F is the relation  $E_{(0,1)^{\mathbb{N}}}^{\ell^1}$  one obtains the following nonclassifiability criterion:

**Criterion 2.2.3.** If *E* is an equivalence relation on a Polish space *X* and there is a Baire measurable function  $f: (0,1)^{\mathbb{N}} \to X$  such that:

- $f(\mathbf{x})Ef(\mathbf{y})$  for any  $\mathbf{x}, \mathbf{y} \in (0, 1)^{\mathbb{N}}$  such that  $\mathbf{x} \mathbf{y} \in \ell^1$ ;
- the preimage under f of any E-class is meager;

then the relation E is not classifiable by countable structures.

In order to apply Criterion 2.2.3 we will need the following fact about nonmeager subsets of  $(0, 1)^{\mathbb{N}}$ :

**Lemma 2.2.4.** If X is a nonmeager subset of  $(0,1)^{\mathbb{N}}$ , then there is an uncountable  $Y \subset X$  such that, for every pair of distinct points  $\mathbf{s}, \mathbf{t}$  of Y,  $\|\mathbf{s} - \mathbf{t}\|_{\infty} \geq \frac{1}{4}$ , where

$$\|\mathbf{s}-\mathbf{t}\|_{\infty} = \sup_{n\in\mathbb{N}} |t_n - s_n|.$$

*Proof.* Define for every  $\mathbf{s} \in (0, 1)$ ,

$$K_{\mathbf{s}} = \left\{ \mathbf{t} \in (0,1)^{\mathbb{N}} \mid \left\| \mathbf{t} - \mathbf{s} \right\|_{\infty} \le \frac{1}{4} \right\}.$$

Observe that  $K_{\mathbf{s}}$  is a closed nowhere dense subset of  $(0, 1)^{\mathbb{N}}$ . Consider the class  $\mathcal{A}$  of subsets Y of X with the property that, for every  $\mathbf{s}, \mathbf{t}$  in Y distinct,  $\|\mathbf{s} - \mathbf{t}\| \geq \frac{1}{4}$ . If  $\mathcal{A}$  is partially ordered by inclusion, then it has some maximal element Y by Zorn's lemma. By maximality,

$$X \subset \bigcup_{\mathbf{t} \in Y} \left\{ \mathbf{s} \in (0,1)^{\mathbb{N}} \mid \|\mathbf{t} - \mathbf{s}\|_{\infty} \le \frac{1}{4} \right\}.$$

Since the set X is nonmeaser, Y is uncountable.

## 2.3 The case of algebras with outer derivations

The aim of this section is to show that if a C\*-algebra A has an outer derivation, then the relation of unitary equivalence of approximately inner automorphisms of A is not classifiable by countable structures. In proving this fact we will also show that any such C\*-algebra satisfies a seemingly stronger property, that we will refer to as Property AEP (see Definition 2.3.4).

A derivation of a C\*-algebra A is a linear function

$$\delta:A\to A$$

satisfying the *derivation identity*:

$$\delta(xy) = \delta(x)y + x\delta(y)$$

for  $x, y \in A$ . The derivation identity implies that  $\delta$  is a bounded linear operator on A; see [109, Proposition 8.6.3]. The set  $\Delta(A)$  of derivations of A is a closed subspace of the Banach space B(A) of bounded linear operators on A. A derivation is called a \*-derivation if it is a positive linear operator, i.e. it sends positive elements to positive elements. Any element a of the multiplier algebra of A defines a derivation  $\operatorname{ad}(ia)$  of A, by

$$\operatorname{ad}(ia)(x) = [ia, x].$$

This is a \*-derivation if and only if a is self-adjoint. A derivation of this form is called *inner*, and *outer* otherwise. More generally, if a is an element of the enveloping von Neumann algebra of A that *derives* A, i.e.  $ax - xa \in A$  for any  $x \in A$ , then one can define the (not necessarily inner) derivation ad(ia) of A. Since any derivation is linear combination of \*-derivations (see [109, 8.6.2]), the existence of an outer derivation is equivalent to the existence of an outer \*-derivation. The set  $\Delta_0(A)$  of inner derivations of A is a Borel (not necessarily closed) subspace of  $\Delta(A)$ . The norm on  $\Delta_0(A)$  defined by

$$\|\operatorname{ad}(ia)\|_{\Delta_0(A)} = \inf \{ \|a - z\| \mid z \in \mathcal{Z}(A) \},\$$

where  $\mathcal{Z}(A)$  denotes the *center* of A, makes  $\Delta_0(A)$  a separable Banach space isometrically isomorphic to the quotient of A by  $\mathcal{Z}(A)$ . The inclusion of  $\Delta_0(A)$  in  $\Delta(A)$  is continuous, and the closure  $\overline{\Delta_0(A)}$  of  $\Delta_0(A)$  in  $\Delta(A)$  is a closed separable subspace of  $\Delta(A)$ . If  $\delta$  is a \*-derivation then the exponential  $\exp(\delta)$  of  $\delta$ , regarded as an element of the Banach algebra B(A) of bounded linear operators of A, is an automorphism of A. Automorphisms of this form are called derivable. If  $\delta = \operatorname{ad}(ia)$  is inner then

$$\exp(\delta) = \operatorname{Ad}(\exp(ia))$$

is inner as well. Lemma 2.3.1 provides a partial converse to this statement. (The converse is in fact false in general; see [67, Example 6.1].) For more information on derivations and derivable automorphisms, the reader is referred to [109, Section 8.6].

**Lemma 2.3.1.** Suppose that A is a primitive C\*-algebra. If  $\delta$  is a \*-derivation of A with operator norm strictly smaller than  $2\pi$  such that  $\exp(\delta)$  is inner, then  $\delta$  is inner.

The lemma is proved in [67, Theorem 4.6 and Remark 4.7] under the additional assumption that A is unital. It is not difficult to check that the same proof works without change in the nonunital case.

**Definition 2.3.2.** Suppose that A is a C\*-algebra,  $(a_n)_{n \in \mathbb{N}}$  is a dense sequence in the unit ball of A, and  $\mathbf{x} = (x_n)_{n \in \mathbb{N}}$  is a sequence of pairwise orthogonal positive contractions of A such that for every  $n \in \mathbb{N}$  and  $i \leq n$ ,

$$||[x_n, a_i]|| \le 2^{-n}.$$

Since the  $x_n$ 's are pairwise orthogonal, if **t** is a sequence of real numbers of absolute value at most 1, then the series

$$\sum_{n \in \mathbb{N}} t_n x_n$$

converges in the strong operator topology to a self-adjoint element of  $A^{**}$ . Moreover, the sequence of inner automorphisms

$$\left( \operatorname{Ad} \left( \exp \left( i \sum_{k \le n} t_k x_k \right) \right) \right)_{n \in \mathbb{N}}$$

of A converges -in view of (2.3.2)- to the approximately inner automorphism

$$\alpha_{\mathbf{t}} := \operatorname{Ad}\left(\exp\left(i\sum_{n\in\mathbb{N}}t_nx_n\right)\right).$$

The equivalence relation  $E_{\mathbf{x}}$  on  $(0,1)^{\mathbb{N}}$  is defined by

 $\mathbf{s}E_{\mathbf{x}}\mathbf{t}$  iff  $\alpha_{\mathbf{t}}$  and  $\alpha_{\mathbf{s}}$  are unitarily equivalent.

This equivalence relation is finer than the relation of  $\ell^1$ -equivalence introduced in Section 2.2. In fact if  $\mathbf{s}, \mathbf{t} \in (0, 1)^{\mathbb{N}}$  and  $\mathbf{s} - \mathbf{t} \in \ell^1$ , then the series

$$\sum_{n \in \mathbb{N}} (t_n - s_n) x_n$$

converges in A. It is then easily verified that

$$u := \exp\left(i\sum_{n\in\mathbb{N}} \left(t_n - s_n\right)x_n\right)$$

is a unitary multiplier of A such that

$$\operatorname{Ad}(u) \circ \alpha_{\mathbf{s}} = \alpha_{\mathbf{t}}.$$

Therefore, if the equivalence classes of  $E_{\mathbf{x}}$  are meager, the continuous function

$$(0,1)^{\mathbb{N}} \to \operatorname{Aut}(A)$$
$$\mathbf{t} \mapsto \alpha_{\mathbf{t}}$$

satisfies the hypothesis of Criterion 2.2.3. This concludes the proof of the following lemma:

**Lemma 2.3.3.** Suppose that A is a C\*-algebra. If for some sequence  $\mathbf{x}$  of pairwise orthogonal positive contractions of A satisfying the commutation condition (2.3.2) the equivalence relation  $E_{\mathbf{x}}$  has meager equivalence classes, then the approximately inner automorphisms of A are not classifiable by countable structures.

Lemma 2.3.3 motivates the following definition.

**Definition 2.3.4.** A C\*-algebra A has Property AEP if for every dense sequence  $(a_n)_{n \in \mathbb{N}}$  in the unit ball of A there is a sequence  $\mathbf{x} = (x_n)_{n \in \mathbb{N}}$  of pairwise orthogonal positive contractions of A such that:

- 1.  $||[x_n, a_i]|| < 2^{-n}$  for  $i \in \{1, 2, \dots, n\}$ ;
- 2. the relation  $E_{\mathbf{x}}$  as in Definition 2.3.2 has meager conjugacy classes.

$$a = \sum_{n \in \mathbb{N}} \left( t_n - s_n \right) x_n$$

of  $A^{**}$  derives A. The automorphism Ad( $\exp(ia)$ ) is outer, and hence such is the \*derivation ad(*ia*). The rest of this section is devoted to prove that, conversely, if A has an outer derivation, then A has Property AEP.

**Theorem 2.3.5.** If A is a  $C^*$ -algebra, the following statements are equivalent:

1. A has an outer derivation;

2. A has Property AEP.

The following lemma shows that primitive nonsimple C\*-algebras have Property AEP. The main ingredients of the proof are borrowed from [28, Lemma 6] and [2, Lemma 3.2].

**Lemma 2.3.6.** If A is a primitive nonsimple infinite-dimensional  $C^*$ -algebra, then it has Property AEP.

Proof. Fix a faithful irreducible representation  $\pi : A \to B(H)$ . By [109, Theorem 3.7.7]  $\pi$  extends to a  $\sigma$ -weakly continuous representation  $\pi^{**} : A^{**} \to B(H)$ . Fix a dense sequence  $(a_n)_{n \in \mathbb{N}}$  in the unit ball of A and a strictly positive contraction  $b_0$  of A. (Recall that a positive contraction  $b_0$  of A is strictly positive if

$$ab_0^{\frac{1}{n}} \to a$$

for every  $a \in A$  [11, II.4.2.1].) As in the proof of [2, Lemma 3.2], one can define a sequence  $\mathbf{x} = (x_n)_{n \in \mathbb{N}}$  of pairwise orthogonal projections such that for some  $\varepsilon > 0$  and every  $k, n \in \mathbb{N}$  such that  $k \leq n$ ,

•  $||x_n b_0|| > \varepsilon;$ 

• 
$$||[x_n, a_k]|| < 2^{-n}$$
.

Now suppose by contradiction that the equivalence relation  $E_{\mathbf{x}}$  has a nonmeager equivalence class X. Thus for every  $\mathbf{t}, \mathbf{s} \in X$  the automorphism

$$\alpha_{\mathbf{t},\mathbf{s}} = \operatorname{Ad}\left(\exp\left(i\sum_{n\in\mathbb{N}}(t_n - s_n)x_n\right)\right)$$

is inner. Fix  $\mathbf{s}, \mathbf{t} \in X$ . Observe that  $\alpha_{\mathbf{t},\mathbf{s}}$  is the exponential of the \*-derivation

$$\delta_{\mathbf{t},\mathbf{s}} = \operatorname{ad}\left(i\sum_{n\in\mathbb{N}}(t_n-s_n)x_n\right).$$

By Lemma 2.3.1 the \*-derivation  $\delta_{\mathbf{t},\mathbf{s}}$  is inner. Thus, there is an element  $z_{\mathbf{t},\mathbf{s}}$  of the center of the enveloping von Neumann algebra of A such that

$$\sum_{n\in\mathbb{N}} (t_n - s_n) x_n + z_{\mathbf{t},\mathbf{s}} \in M(A).$$

Recall that  $\pi$  has been extended to a  $\sigma$ -weakly continuous representation  $\pi^{**} : A^{**} \to B(H)$ by [109, Theorem 3.7.7]. The image of a central element of  $A^{**}$  under  $\pi^{**}$  belongs to the relative commutant of  $\pi[A]$  in B(H), which consists only of scalar multiples of the identity by [11, II.6.1.8]. Thus,

$$\pi\left(\sum_{n\in\mathbb{N}}(t_n-s_n)x_n\right)\in\pi^{**}\left[M(A)\right]$$

Hence

$$\pi\left(b_0\sum_{n\in\mathbb{N}}(t_n-s_n)x_n\right)\in\pi[A].$$

By Lemma 2.2.4 one can find an uncountable subset Y of X such that any pair of distinct elements of Y has uniform distance at least  $\frac{1}{4}$ . Fix  $\mathbf{s} \in Y$ . For all  $\mathbf{t}, \mathbf{t}' \in Y$ , there is  $m \in \mathbb{N}$  such that

$$\left|t_m - t'_m\right| \ge \frac{1}{4}$$

Henceforth,

$$\begin{aligned} & \left\| \pi \left( b_0 \left( \sum_{n \in \mathbb{N}} (t_n - s_n) x_n \right) \right) - \pi \left( b_0 \left( \sum_{n \in \mathbb{N}} (t'_n - s_n) x_n \right) \right) \right\| \\ &= \left\| \pi \left( b_0 \sum_{n \in \mathbb{N}} (t_n - t'_n) x_n \right) \right\| \\ &= \left\| b_0 \sum_{n \in \mathbb{N}} (t_n - t'_n) x_n \right\| \\ &\geq \left\| b_0 \sum_{n \in \mathbb{N}} (t_n - t'_n) x_n x_m a_0 \right\| \\ &\geq \left\| t_m - t'_m \right\| \| (x_m b_0)^* (x_m b_0) \| \ge \frac{\varepsilon^2}{4}. \end{aligned}$$

Since Y is uncountable this contradicts the separability of  $\pi[A]$ .

In order to prove Property AEP for all C\*-algebra with outer \*-derivations we need the fact that Property AEP is *liftable*. This means that if a C\*-algebra A has a quotient with property AEP, then A has property AEP. (For an exhaustive introduction to liftable properties the reader is referred to Chapter 8 of [88].)

**Lemma 2.3.7.** If  $\pi : A \to B$  is a surjective \*-homomorphism and B has Property AEP, then A has Property AEP.

Proof. Suppose that  $(a_n)_{n\in\mathbb{N}}$  is a dense sequence in A. Thus,  $(\pi(a_n))_{n\in\mathbb{N}}$  is a dense sequence in B. Pick a sequence  $(y_n)_{n\in\mathbb{N}}$  in B obtained from  $(\pi(a_n))_{n\in\mathbb{N}}$  as in the definition of Property AEP. By [88, Lemma 10.1.12], there is a sequence  $(z_n)_{n\in\mathbb{N}}$  of pairwise orthogonal positive contractions of A such that  $\pi(z_n) = y_n$  for every  $n \in \mathbb{N}$ . Fix an increasing quasicentral approximate unit of Ker $(\pi)$  (cf. [11, Section II.4.3]), i.e. a sequence  $(u_k)_{k\in\mathbb{N}}$  of elements of Ker $(\pi)$  such that:

• 
$$\lim_{k \to +\infty} \|u_k x - x\| = \lim_{k \to +\infty} \|xu_k - x\| = 0$$
 for every  $x \in \operatorname{Ker}(\pi)$ ;

•  $\lim_{k \to +\infty} \|[u_k, a]\| = 0$  for every  $a \in A$ .

For every  $n, i \in \mathbb{N}$  such that  $i \leq n$ , by [11, Proposition II.5.1.1],

$$\lim_{k \to +\infty} \left\| z_n^{\frac{1}{2}} (1 - u_k) z_n^{\frac{1}{2}} a_i - a_i z_n^{\frac{1}{2}} (1 - u_k) z_n^{\frac{1}{2}} \right\| = \lim_{k \to +\infty} \left\| (1 - u_k) (z_n a_i - a_i z_n) \right\|$$
$$= \left\| y_n \pi(a_i) - \pi(a_i) y_n \right\| < 2^{-n}.$$

Thus, there is  $k_n \in \mathbb{N}$  such that, if

$$x_n = z_n^{\frac{1}{2}} (1 - u_{k_n}) z_n^{\frac{1}{2}},$$

then

$$\|x_n a_i - a_i x_n\| < 2^{-n}$$

for every  $i \leq n$ . Observe that  $(x_n)_{n \in \mathbb{N}}$  is a sequence of pairwise orthogonal positive contractions of A. Moreover, if  $E \subset (0,1)^{\mathbb{N}}$  is nonmeaser, consider  $\mathbf{s}, \mathbf{t} \in E$  such that the automorphism

$$\operatorname{Ad}\left(\exp\left(i\sum_{n\in\mathbb{N}}(t_n-s_n)y_n\right)\right)$$

of B is outer. We claim that the automorphism

$$\operatorname{Ad}\left(\exp\left(i\sum_{n\in\mathbb{N}}(t_n-s_n)x_n\right)\right)$$

of A is outer. Suppose that this is not the case. Thus, there is z in the center of the enveloping von Neumann algebra of A such that

$$\exp\left(i\sum_{n\in\mathbb{N}}(t_n-s_n)x_n\right)+z\in U(A).$$

Denoting by  $\pi^{**}: A^{**} \to B^{**}$  the normal extension of  $\pi$  –see [11, III.5.2.10]– one has that

$$\exp\left(i\sum_{n\in\mathbb{N}}(t_n-s_n)y_n\right) + \pi^{**}(z) = \pi^{**}\left(\exp\left(i\sum_{n\in\mathbb{N}}(t_n-s_n)x_n\right) + z\right) \in U(B)$$

by Theorem 4.2 of [1]. Since  $\pi^{**}(z)$  belongs to the center of the enveloping von Neumann algebra of B,

$$\exp\left(i\sum_{n\in\mathbb{N}}(t_n-s_n)y_n\right)+\pi^{**}(z)$$

is a unitary multiplier of B that implements

$$\operatorname{Ad}\left(\exp\left(i\sum_{n\in\mathbb{N}}(t_n-s_n)y_n\right)\right).$$

Hence, the latter automorphism of B is inner, contradicting the assumption.

Liftability of Property AEP allows one to easily bootstrap Property AEP from primitive nonsimple C\*-algebras to C\*-algebra whose primitive spectrum is not  $T_1$ .

**Lemma 2.3.8.** If A is a C\*-algebra whose primitive spectrum  $\check{A}$  is not  $T_1$ , then A has Property AEP.

*Proof.* Since A is not  $T_1$ , by [109, 4.1.4] there is an irreducible representation  $\pi$  of A whose kernel is not a maximal ideal. This implies that the image of A under  $\pi$  is a nonsimple primitive C\*-algebra. By Lemma 2.3.6 the latter C\*-algebra has Property AEP. Therefore, being Property AEP liftable by Lemma 2.3.7, A has Property AEP.

In order to show that a C\*-algebra A has Property AEP, it is sometimes easier to show that it has a stronger property that we will refer to as Property AEP<sup>+</sup>. Property AEP<sup>+</sup> appears, without being explicitly defined, in the proofs of Lemma 17, Theorem 18, and the main Theorem of [28], as well as in the proofs of Lemma 3.5 and Lemma 3.6 of [2].

Recall that a bounded sequence  $(x_n)_{n\in\mathbb{N}}$  of elements of A is called central if for every  $a \in A$ ,

$$\lim_{n \to +\infty} \|[x_n, a]\| = 0.$$

The beginning of Section 2.4 contains a discussion about the notion of central sequence, the related notion of hypercentral sequence, and their basic properties.

**Definition 2.3.9.** A C\*-algebra A has Property AEP<sup>+</sup> if there is a sequence  $(\pi_n)_{n \in \mathbb{N}}$  of irreducible representations of A such that, for some positive contraction  $b_0$  of A and a central sequence  $(x_n)_{n \in \mathbb{N}}$  of pairwise orthogonal positive contractions of A:

• the sequence

$$(\pi_n((x_n-\lambda)b_0))_{n\in\mathbb{N}}$$

does not converge to 0 for any  $\lambda \in \mathbb{C}$ ;

•  $x_m \in \text{Ker}(\pi_n)$  for every pair of distinct natural numbers n, m.

To prove that Property AEP<sup>+</sup> implies Property AEP we will need the following lemma:

**Lemma 2.3.10.** Fix a strictly positive real number  $\eta$ . For every  $\varepsilon > 0$  there is  $\delta > 0$  such that for every  $C^*$ -algebra A and every pair of positive contractions x, b of A such that  $\|b\| \ge \eta$ , if

$$\|(\exp(ix) - \mu)b\| \le \delta$$

for some  $\mu \in \mathbb{C}$  then

 $\|(x-\lambda)b\| \le \varepsilon$ 

for some  $\lambda \in \mathbb{C}$ .

*Proof.* Fix  $\varepsilon > 0$ . Let L be the principal branch of the logarithm. Since L is an analytic function on the open disc of radius 1 centered in 1, there is a polynomial

$$p(Z) = \rho_0 + \rho_1 Z + \ldots + \rho_n Z^n$$

such that

$$\left|p\left(z\right) - L\left(z\right)\right| \le \frac{\varepsilon}{2}$$

for every  $z \in \mathbb{C}$  such that  $|z - 1| \leq \exp(i)$ . In particular for every  $t \in [0, 1]$ 

$$|p(\exp(it)) - t| = |p(\exp(it)) - L(\exp(it))| \le \frac{\varepsilon}{2}.$$

If  $\mu \in \mathbb{C}$  is such that  $|\mu| \leq \frac{2}{\eta}$ , define  $p_{\mu}(Z)$  to be the polynomial in Z obtained from p(Z) by replacing the indeterminate Z by  $Z + \mu$ . Observe that the j -th coefficient of  $p_{\mu}(Z)$  is

$$\rho_j^{\mu} = \sum_{i=j}^n \rho_i \binom{i}{j} \mu^{j-i}$$

for  $0 \le j \le n$ . Finally define

$$C = \sum_{1 \le j \le i \le n} |\rho_i| \binom{i}{j} \left(\frac{3}{\eta}\right)^{j-1} \left(\frac{2}{\eta}\right)^{j-i}$$

and

$$\delta = \min\left\{\frac{\varepsilon}{2C}, 1\right\}.$$

Suppose that A is a C\*-algebra and  $x, b \in A$  are positive contractions such that  $||b|| \ge \eta$ and, for some  $\mu \in \mathbb{C}$ ,

$$\|(\exp(ix) - \mu)b\| \le \delta.$$

Thus,

$$|\mu| \le \frac{2}{\eta}.$$

Moreover

$$\begin{aligned} \|(x - \rho_0^{\mu})b\| &= \|(p(\exp(ix)) - \rho_0^{\mu})b\| + \frac{\varepsilon}{2} \\ &= \left\| \left( \sum_{j=1}^n \rho_j^{\mu}(\exp(ix) - \mu)^j \right) b \right\| + \frac{\varepsilon}{2} \\ &\leq \sum_{j=1}^n \left| \rho_j^{\mu} \right| \|\exp(ix) - \mu\|^{j-1} \,\delta + \frac{\varepsilon}{2} \\ &\leq \sum_{j=1}^n \sum_{i=j}^n |\rho_i| \left( \frac{i}{j} \right) \left( \frac{2}{\eta} \right)^{j-i} \left( \frac{3}{\eta} \right)^{j-1} \delta + \frac{\varepsilon}{2} \\ &\leq C\delta + \frac{\varepsilon}{2} \leq \varepsilon. \end{aligned}$$

This concludes the proof.

We can now prove that Property AEP<sup>+</sup> implies property AEP.

## **Proposition 2.3.11.** If a $C^*$ -algebra A has Property $AEP^+$ , then it has property AEP.

*Proof.* Suppose that  $(\pi_n)_{n\in\mathbb{N}}$  is a sequence of irreducible representations of A,  $b_0$  is a positive contraction of A of norm 1, and  $(x_n)_{n\in\mathbb{N}}$  is a sequence of orthogonal positive elements of A as in the definition of Property AEP<sup>+</sup>. Fix a dense sequence  $(a_n)_{n\in\mathbb{N}}$  in the unit ball of A. After passing to a subsequence of the sequence  $(x_n)_{n\in\mathbb{N}}$ , we can assume that for some

 $\varepsilon > 0$ , for every  $\mu \in \mathbb{C}$  and every  $n \in \mathbb{N}$ ,

$$\|\pi_n((x_n-\mu)b_0)\| \ge \varepsilon$$

and

$$||[x_n, a_i]|| < 2^{-n}$$

for  $i \leq n$ . Thus, for every  $\mu \in \mathbb{C}$ ,  $n \in \mathbb{N}$ , and  $t \in \left(\frac{1}{4}, 1\right)$ ,

$$\|\pi_n((tx_n-\mu)b_0)\| \ge \frac{\varepsilon}{4}.$$

Observe that, in particular,

$$\|\pi_n(b_0)\| \ge \varepsilon$$

for every  $n \in \mathbb{N}$ . Consider  $\delta > 0$  obtained from  $\frac{\varepsilon}{8}$  as in Lemma 2.3.10 (where we set  $\eta = \varepsilon$ ). We claim that for every  $t \in (\frac{1}{4}, 1)$ ,  $n \in \mathbb{N}$ , and  $\mu \in \mathbb{C}$ ,

$$\|\pi_n((\exp{(itx_n)} - \mu)b_0)\| \ge \delta.$$

In fact suppose by contradiction that there are  $t \in (\frac{1}{4}, 1)$ ,  $n \in \mathbb{N}$ , and  $\mu \in \mathbb{C}$  such that

$$\|(\exp(it\pi_n(x_n)) - \mu)\pi_n(b_0)\| = \|\pi_n((\exp(itx_n) - \mu)b_0)\| < \delta.$$

Thus by our choice of  $\delta$  there is  $\mu \in \mathbb{C}$  such that

$$\|\pi_n \left( (itx_n - \mu) \, b_0 \right)\| = \|(it\pi_n \, (x_n) - \mu)\pi_n \, (b_0)\| \le \frac{\varepsilon}{8}.$$

Such inequality contradicts 2.3. This concludes the proof of the assertion that for every  $t \in (\frac{1}{4}, 1), n \in \mathbb{N}$ , and  $\mu \in \mathbb{C}$ ,

$$\|\pi_n((\exp{(itx_n)} - \mu)b_0)\| \ge \delta.$$

We now claim that the sequence  $(x_n)_{n \in \mathbb{N}}$  witnesses the fact that A has property AEP. Assume by contradiction that there is a nonmeager subset X of  $(0,1)^{\mathbb{N}}$  such that for every  $\mathbf{s}, \mathbf{t} \in X$ , the automorphism

$$\operatorname{Ad}\left(\exp\left(i\sum_{n\in\mathbb{N}}(t_n-s_n)x_n\right)\right)$$

of A is inner. If  $\mathbf{s}, \mathbf{t} \in X$ , then there is an element  $z_{\mathbf{t},\mathbf{s}}$  in the center of the enveloping von Neumann algebra of A such that

$$\exp\left(i\sum_{n\in\mathbb{N}}(t_n-s_n)x_n+z_{\mathbf{t},\mathbf{s}}\right)$$

multiplies A. Hence

$$y_{\mathbf{t},\mathbf{s}} = \exp\left(i\sum_{n\in\mathbb{N}}(t_n - s_n)x_n + z_{\mathbf{t},\mathbf{s}}\right)b_0$$

is an element of A. By Lemma 2.2.4, one can find an uncountable subset Y of X such that, for any  $\mathbf{t}, \mathbf{s} \in Y$ , there is  $m \in \mathbb{N}$  such that

$$|t_m - s_m| \ge \frac{1}{4}.$$

Fix  $\mathbf{s} \in Y$  and observe that, for  $\mathbf{t}, \mathbf{t}' \in Y$ ,

$$\pi_{n_0}(\exp(z_{\mathbf{t}',\mathbf{s}} - z_{\mathbf{t},\mathbf{s}})) = \mu \mathbf{1}$$

is a scalar multiple of the identity. Therefore

$$\begin{aligned} \|y_{\mathbf{t},\mathbf{s}} - y_{\mathbf{t}',\mathbf{s}}\| \\ &= \left\| \left( \exp\left(i\sum_{n\in\mathbb{N}} (t_n - t'_n)x_n\right) - \exp(z_{\mathbf{t}',\mathbf{s}} - z_{\mathbf{t},\mathbf{s}})\right)a_0\right\| \\ &\geq \left\|\pi_{n_0}\left( \left( \exp\left(i\sum_{n\in\mathbb{N}} (t_n - t'_n)x_n\right) - \exp(z_{\mathbf{t}',\mathbf{s}} - z_{\mathbf{t},\mathbf{s}})\right)a_0\right)\right\| \\ &= \left\|\pi_{n_0}((\exp((t_{n_0} - t'_{n_0})x_n) - \mu)a_0)\right\| \\ &\geq \varepsilon. \end{aligned}$$

This contradicts the separability of A.

The proofs of Lemma 2.3.12 and Lemma 2.3.13 are contained, respectively, in the proofs of Lemmas 3.6 and 3.7 of [2] and in the proof of the implication  $(i) \Rightarrow (ii)$  at page 139 of [28].

Recall that a point x of a topological space X is called *separated* if, given any point y of X distinct from x, there are disjoint open neighborhoods of x and y.

**Lemma 2.3.12.** Suppose that A is a C\*-algebra whose primitive spectrum  $\check{A}$  is  $T_1$ . Consider

a sequence  $(\xi_n)_{n\in\mathbb{N}}$  of separated points in  $\check{A}$ . Define F to be the set of limit points of the sequence  $(\xi_n)_{n\in\mathbb{N}}$  and I to be the closed self-adjoint ideal of A corresponding to F. If either the quotient A/I does not have continuous trace, or the multiplier algebra of A/I has nontrivial center, then A has Property  $AEP^+$ .

**Lemma 2.3.13.** If A is a C\*-algebra whose spectrum  $\hat{A}$  is homeomorphic to the one-point compactification of a countable discrete space, then A has Property  $AEP^+$ .

We can now prove the main result of this section that Property AEP as defined in 2.3.4 is equivalent to having an outer \*-derivation.

*Proof of Theorem 2.3.5.* We have already pointed out that Property AEP implies the existence of an outer \*-derivation. It remains only to show the converse. Suppose that A has an outer derivation. By [28, Lemma 16], either there is a quotient B of A whose spectrum B is homeomorphic to the one point compactification of a countable discrete space, or the primitive spectrum A of A is not Hausdorff. In the first case, A has Property AEP by virtue of Lemma 2.3.13 and Lemma 2.3.7. Suppose that, instead, the primitive spectrum  $\hat{A}$  of A is not Hausdorff. If A is not even  $T_1$ , the conclusion follows from Lemma 2.3.8. Suppose now that A is  $T_1$ . Since A is not Hausdorff, there are two points  $\rho_0, \rho_1$  of A that do not admit any disjoint open neighbourhoods. By [20, Proposition 1] the set of separated points is dense in A. Therefore can find a sequence  $(\xi_n)_{n\in\mathbb{N}}$  of separated points of A whose set F of limit points contains both  $\rho_0$  and  $\rho_1$ . Define I to be the closed self-adjoint ideal I of A corresponding to the closed subset F of Å. Since F contains at least two points, A/Iis nonsimple. Consider now two cases: If A/I has continuous trace then by [2, Theorem 3.9] and [2, Lemma 3.1], the multiplier algebra of A/I has nontrivial center. Therefore A has Property AEP<sup>+</sup> by Lemma 2.3.12. On the other hand if A/I does not have continuous trace, then again A has Property AEP<sup>+</sup> by Lemma 2.3.12. In either case, it follows that A has Property AEP<sup>+</sup> and, in particular, Property AEP. 

## 2.4 The case of algebras with only inner derivations

In this section we will prove that, if a C\*-algebra A with only inner derivations does not have continuous trace, then the relation of unitary equivalence of approximately inner automorphisms of A is not classifiable by countable structures. In proving this fact we will also show that any such C\*-algebra contains a central sequence that is not strict-hypercentral.

If A is a C\*-algebra, denote by  $A^{\infty}$  the quotient of the direct product  $\prod_{n \in \mathbb{N}} A$  by the direct sum  $\bigoplus_{n \in \mathbb{N}} A$ ; see [11, II.8.1.2]. Identifying as it is customary A with the algebra

of elements of  $A^{\infty}$  admitting constant representative sequence, denote by  $A_{\infty}$  the relative commutant

$$A' \cap A^{\infty} = \{ x \in A^{\infty} \mid \forall y \in A , [x, y] = 0 \}.$$

Finally define

$$\operatorname{Ann}(A, A_{\infty}) = \{ x \in A_{\infty} \mid \forall y \in A , xy = 0 \}$$

to be the annihilator ideal of A in  $A_{\infty}$ . Observe that, if A is unital, then  $\operatorname{Ann}(A, A_{\infty})$  is the trivial ideal.

A central sequence in a C\*-algebra A is a bounded sequence  $(x_n)_{n \in \mathbb{N}}$  of elements of A that asymptotically commute with any element of A. This means that for any  $a \in A$ ,

$$\lim_{n \to +\infty} \|[x_n, a]\| = 0.$$

Equivalently the image of  $(x_n)_{n\in\mathbb{N}}$  in the quotient of  $\prod_{n\in\mathbb{N}} A$  by  $\bigoplus_{n\in\mathbb{N}} A$  belongs to  $A_{\infty}$ . From the last characterization it is clear that if  $(x_n)_{n\in\mathbb{N}}$  is a central sequence of normal elements A with spectra contained in some subset D of  $\mathbb{C}$  and  $f: D \to \mathbb{C}$  is a continuous function such that f(0) = 0, then the sequence  $(f(x_n))_{n\in\mathbb{N}}$  is central. It is not difficult to verify that, if  $(x_n)_{n\in\mathbb{N}}$  is a central sequence and  $b \in M(A)$ , then the sequence  $([b, x_n])_{n\in\mathbb{N}}$  converges strictly to 0. (The strict topology on M(A) has been defined in Section 2.1.)

Let us call a central sequence  $(x_n)_{n \in \mathbb{N}}$  norm-hypercentral if it asymptotically commutes in the norm topology with any other central sequence. This amounts to say that for any other central sequence  $(y_n)_{n \in \mathbb{N}}$ 

$$\lim_{n \to +\infty} \|[x_n, y_n]\| = 0.$$

Equivalently the image of  $(x_n)_{n\in\mathbb{N}}$  in the quotient of  $\prod_{n\in\mathbb{N}} A$  by  $\bigoplus_{n\in\mathbb{N}} A$  belongs to the center of  $A_{\infty}$ . For our purposes it will be more convenient to look at central sequences that asymptotically commute *in the strict topology* with any other central sequence. This motivates the following definition:

**Definition 2.4.1.** Suppose that A is a C\*-algebra. A sequence  $(x_n)_{n \in \mathbb{N}}$  of elements of A is **strict-hypercentral** if it is central and, for any other central sequence  $(y_n)_{n \in \mathbb{N}}$ , the sequence

$$([x_n, y_n])_{n \in \mathbb{N}}$$

converges to 0 in the strict topology.

Observe that a central sequence  $(x_n)_{n \in \mathbb{N}}$  is strict-hypercentral if and only if the image of

the element of  $A_{\infty}$  having  $(x_n)_{n \in \mathbb{N}}$  as representative sequence in the quotient  $A_{\infty} / \text{Ann}(A, A_{\infty})$ belongs to the center of  $A_{\infty} / \text{Ann}(A, A_{\infty})$ . It is clear from this characterization that, if  $(x_n)_{n \in \mathbb{N}}$  is a strict-hypercentral sequence of normal elements of A with spectra contained in some subset D of  $\mathbb{C}$  and  $f: D \to \mathbb{C}$  is a complex-valued continuous function such that f(0) = 0, then the sequence  $(f(x_n))_{n \in \mathbb{N}}$  is strict-hypercentral. When A is unital the ideal Ann  $(A, A_{\infty})$  is trivial, and hence the notions of strict-hypercentral and norm-hypercentral sequence coincide. Therefore in the unital case a norm-hypercentral sequence will be simply called hypercentral.

The fact that a unital simple infinite-dimensional C\*-algebra contains a central sequence that is not hypercentral is a particular case of [112, Theorem 3.6]. We will show here how one can generalize this fact to all simple nonelementary C\*-algebras. The proof deeply relies on ideas from [112].

**Lemma 2.4.2.** If  $(x_n)_{n \in \mathbb{N}}$  is a strict-hypercentral sequence in A and  $\alpha$  is an approximately inner automorphism of A, then  $(\alpha(x_n) - x_n)_{n \in \mathbb{N}}$  converges strictly to 0.

Proof. The same proof of Kaplansky's density theorem [109, Theorem 2.3.3] shows that the unit ball of A is strictly dense in the unit ball of M(A); see [85, Proposition 1.4]. (The strict topology on the multiplier algebra of A has been defined in Section 2.1.) It follows that, if  $\varepsilon > 0$  and a is an element of A, then there is a finite subset F of the unit ball of A, a positive real number  $\delta$ , and a natural number  $n_0$  such that, for every  $n \ge n_0$  and every y in the unit ball M(A) such that  $||[y, z]|| < \delta$  for every  $z \in F$ ,

$$\max\left\{\left\|a(x_ny - yx_n)\right\|, \left\|(x_ny - yx_n)a\right\|\right\} \le \varepsilon.$$

Consider the open neighbourhood

$$U = \{ \alpha \in \operatorname{Aut}(A) \mid \|\alpha(x) - x\| < \delta \text{ for every } x \in F \}$$

of  $id_A$  in Aut(A). Observe that if  $\beta \in U$  is inner, then for every  $n \geq n_0$ 

$$\|(\beta(x_n) - x_n)a\| \le \varepsilon$$

and

$$\|a(\beta(x_n) - x_n)\| \le \varepsilon.$$

Approximating with inner automorphisms, one can see that the same is true if  $\beta \in U$  is just approximately inner. Since  $\alpha$  is approximately inner, there is a unitary multiplier u of A and an approximately inner automorphism  $\beta$  of A in U such that

$$\alpha = \beta \circ \operatorname{Ad}(u).$$

Consider a natural number  $n_1 \ge n_0$  such that, for  $n \ge n_1$ ,

$$\left\|\beta^{-1}(a)[x_n,u]\right\| \le \varepsilon$$

and

$$\left\| [x_n, u^*] \beta^{-1}(a) \right\| \le \varepsilon$$

It follows that, if  $n \ge n_1$ ,

$$\begin{aligned} \|a(\alpha(x_n) - x_n)\| &\leq \|a\beta(\operatorname{Ad}(u)(x_n) - x_n)\| + \|\beta(x_n) - x_n\| \\ &\leq \|\beta^{-1}(a)(ux_nu^* - x_n)\| + \varepsilon \\ &= \|\beta^{-1}(a)[x_n, u]\| + \varepsilon \\ &\leq 2\varepsilon \end{aligned}$$

and, analogously,

$$\|(\alpha(x_n) - x_n)a\| \le 2\varepsilon.$$

Since  $\varepsilon$  was arbitrary, this concludes the proof of the fact that

$$(a(x_n) - x_n)_{n \in \mathbb{N}}$$

converges strictly to 0.

If  $\alpha$  is an automorphism of a C\*-algebra A, then  $\alpha^{**}$  denotes the unique extension of  $\alpha$  to a  $\sigma$ -weakly continuous automorphism of the enveloping von Neumann algebra  $A^{**}$  of A (defined as in [11, Proposition III.5.2.10]).

**Lemma 2.4.3.** Suppose that A is a C\*-algebra such that every central sequence in A is strict-hypercentral. If  $\alpha$  is an approximately inner automorphism of A, then  $\alpha^{**}$  fixes pointwise the center of  $A^{**}$ , i.e.  $\alpha^{**}(z) = z$  for every central element of  $A^{**}$ .

*Proof.* Observe that z derives A, since

$$za - az = 0 \in A$$

for every  $a \in A$ . Thus, by Lemma 1.1 of [2], there is a bounded net  $(z_{\lambda})$  in A converging strongly to z such that, for every  $a \in A$ ,

$$\lim_{\lambda} \|[z_{\lambda} - z, a]\| = 0$$

Recall that strong and  $\sigma$ -strong topology agree on bounded sets, and that the  $\sigma$ -strong topology is stronger than the  $\sigma$ -weak topology; see [11, Definition I.3.1.1]. Thus the net  $(z_{\lambda})$  converges a fortiori  $\sigma$ -weakly to z. Since the  $\sigma$ -weak topology on  $A^{**}$  is the weak<sup>\*</sup> topology on  $A^{**}$  regarded as the dual space of  $A^*$ , the unit ball of  $A^{**}$  is  $\sigma$ -weakly compact by Alaoglu's theorem [110, Theorem 2.5.2]. Moreover by Kaplanski's Density Theorem [109, Theorem 2.3.3] the unit ball of A is  $\sigma$ -weakly dense in the unit ball of  $A^{**}$ . As a consequence the unit ball of  $A^{**}$  is  $\sigma$ -weakly metrizable, and the same holds for balls of arbitrary radius Thus we can find a bounded sequence  $(z_n)_{n \in \mathbb{N}}$  in A converging  $\sigma$ -weakly to z such that, for every  $a \in A$ ,

$$\lim_{n \to +\infty} \|[z_n - z, a]\| = 0.$$

Since

$$[z_n - z, a] = [z_n, a]$$

for every  $n \in \mathbb{N}$ ,  $(z_n)_{n \in \mathbb{N}}$  is a central and hence strict-hypercentral sequence (every central sequence of A is strict-hypercentral by assumption). Since  $\alpha^{**}$  is a  $\sigma$ -weakly continuous automorphism of  $A^{**}$  extending  $\alpha$ ,  $(\alpha(z_n))_{n \in \mathbb{N}}$  converges  $\sigma$ -weakly to  $\alpha^{**}(z)$ . It follows from Lemma 2.4.3 and from the facts that  $\alpha$  is approximately inner and the sequence  $(z_n)_{n \in \mathbb{N}}$  is strict-hypercentral that the bounded sequence  $(z_n - \alpha(z_n))_{n \in \mathbb{N}}$  converges strictly to 0. By [88, Lemma 1.3.1] and since weak and  $\sigma$ -weak topology agree on bounded sets, the sequence  $(z_n - \alpha(z_n))_{n \in \mathbb{N}}$  converges  $\sigma$ -weakly to 0. Therefore  $z = \alpha^{**}(z)$ .

A C\*-algebra is called *elementary* if it is \*-isomorphic to the algebra of compact operators on some Hilbert space; see [11, Definition IV.1.2.1]. By Corollary 1 of Theorem 1.4.2 in [5] any elementary C\*-algebra is simple. Moreover by Corollary 3 of Theorem 1.4.4 in [5] any automorphism of an elementary C\*-algebra is inner; in particular the group Inn(A) of inner automorphisms of an elementary C\*-algebra A is closed inside the group Aut(A) of all automorphisms. Conversely if the group of inner automorphisms of a simple C\*-algebra A is closed, then A is elementary by [111, Theorem 3.1] together with [11, Corollary IV.1.2.6 and Proposition IV.1.4.19].

Recall that in this chapter all C\*-algebras (apart from multiplier algebras and enveloping von Neumann algebras) are assumed to be norm separable. In particular separability of A is assumed in Proposition 2.4.4; however we do not know if the separability assumption is necessary there. (This is also asked in [36, Question 4.3].)

**Proposition 2.4.4.** If A is a simple  $C^*$ -algebra such that every central sequence in A is strict-hypercentral, then A is elementary.

*Proof.* It is enough to show that Inn(A) is closed in Aut(A) or, equivalently, that no outer automorphism is approximately inner. Fix an outer automorphism  $\alpha$  of A. Since A is simple, by [80, Corollary 2.3], there is an irreducible representation  $\pi$  such that  $\pi$  and  $\pi \circ \alpha$ are not unitarily equivalent; see Section 2.1. If z is the central cover of  $\pi$  in  $A^{**}$  (defined as in [109, 3.8.1]), then  $\alpha^{**}(z)$  is the central cover of  $\pi \circ \alpha$ ; moreover, being  $\pi$  and  $\pi \circ \alpha$ not equivalent,  $\alpha^{**}(z)$  is different from z by [109, Theorem 3.8.2]. Thus  $\alpha^{**}$  does not fixes pointwise the center of  $A^{**}$  and, by Lemma 2.4.3,  $\alpha$  is not approximately inner.

Proposition 2.4.4 shows that any simple nonelementary C\*-algebra contains a central sequence that is not strict-hypercentral. It is clear that the same conclusion holds for any C\*-algebra containing a simple nonelementary C\*-algebra as a direct summand. By Theorem 3.9 of [2], this class of C\*-algebras coincides with the class of C\*-algebras that have only inner derivations and do not have continuous trace. This concludes the proof of the following proposition:

**Proposition 2.4.5.** If A is a  $C^*$ -algebra that does not have continuous trace and has only inner derivations, then A contains a central sequence that is not strict-hypercentral.

In view of this result, in order to finish the proof of Theorem 2.0.17, it is enough to show that its conclusion holds for a C\*-algebra A containing a central sequence that is not strict-hypercentral.

**Proposition 2.4.6.** If A is a  $C^*$ -algebra containing a central sequence that is not stricthypercentral, then the approximately inner automorphisms of A are not classifiable by countable structures up to unitary equivalence.

*Proof.* Fix a dense sequence  $(a_n)_{n \in \mathbb{N}}$  in the unit ball of A. Suppose that  $(x_n)_{n \in \mathbb{N}}$  is a central sequence in A that is not strict-hypercentral. Thus there is a central sequence  $(y_n)_{n \in \mathbb{N}}$  in A such that the sequence

$$([x_n, y_n])_{n \in \mathbb{N}}$$

does not converge strictly to 0. This implies that, for some positive contraction b in A, then the sequence

$$(b[x_n, y_n])_{n \in \mathbb{N}}$$

does not converge to 0 is norm. Without loss of generality we can assume that, for every  $n \in \mathbb{N}$ ,  $x_n$  and  $y_n$  are positive contractions. Observe that the sequence  $(\exp(itx_n) - 1)_{n \in \mathbb{N}}$  is not strict-hypercentral for any  $t \in (0, 1)$ . After passing to subsequences, we can assume that for some strictly positive real number  $\varepsilon$ , for every  $t \in (0, 1)$ , every  $s \in (\frac{1}{2}, 1)$ , and every  $n, k \in \mathbb{N}$  such that  $k \leq n$ :

- $||[a_k, \exp(itx_n)]|| < 2^{-n};$
- $||b[x_n, y_n]|| \ge \varepsilon;$
- $||b[\exp(isx_n), y_n]|| \ge \varepsilon.$

Define  $\eta = \frac{\varepsilon}{20}$ . After passing to a further subsequence, we can assume that, for every  $t \in (0, 1)$  and every  $n, k \in \mathbb{N}$  such that  $k \leq n$ :

- $\|[\exp(itx_k), y_n]\| < 2^{-n}\eta;$
- $||[y_k, \exp(itx_n)]|| < 2^{-n}\eta;$
- $\|[\exp(itx_k), \exp(isx_n)]\| < 2^{-n}\eta.$

It is not difficult to verify that, if  $\mathbf{t} \in (0,1)^{\mathbb{N}}$ , then the sequence

$$\left(\mathrm{Ad}\left(\prod_{k=1}^{n}\exp(it_{k}x_{k})\right)\right)_{n\in\mathbb{N}}$$

is Cauchy in Aut(A). Denoting by f(t) its limit, one obtains a function

$$f: (0,1)^{\mathbb{N}} \to \overline{\mathrm{Inn}(A)}.$$

In the rest of the proof we will show that f satisfies the hypothesis of Criterion 2.2.3. Suppose that M is a Lipschitz constant for the function  $t \mapsto \exp(it)$  on [0,1]. If  $\mathbf{t}, \mathbf{s} \in (0,1)^{\mathbb{N}}$ and  $n \in \mathbb{N}$  are such that  $|t_k - s_k| < \varepsilon$  for  $k \in \{1, 2, ..., n\}$ , then it is easy to see that

$$\|f(\mathbf{t})(a_k) - f(\mathbf{s})(a_k)\| \le 2^{-n+1} + \varepsilon M$$

for  $k \leq n$ . This shows that the function f is continuous, particularly, Baire measurable. Moreover, if  $\mathbf{t}, \mathbf{s} \in (0, 1)^{\mathbb{N}}$  are such that  $\mathbf{s} - \mathbf{t} \in \ell^1$ , then the sequence

$$(\exp(it_1x_1)\cdots\exp(it_nx_n)\exp(-is_nx_n)\cdots\exp(-is_1x_1))_{n\in\mathbb{N}}$$

is Cauchy in U(A), and hence has a limit  $u \in U(A)$ . It is now readily verified that

$$f(\mathbf{t}) = \mathrm{Ad}(u) \circ f(\mathbf{s})$$

and hence  $f(\mathbf{t})$  and  $f(\mathbf{s})$  are unitarily equivalent. Finally, suppose that C is a nonmeager subset of  $(0,1)^{\mathbb{N}}$ . Thus, there are  $\mathbf{t}, \mathbf{s} \in C$  such that  $|t_n - s_n| \in (\frac{1}{2}, 1)$  for infinitely many  $n \in \mathbb{N}$ . We claim that  $f(\mathbf{t})$  and  $f(\mathbf{s})$  are not unitarily equivalent. Suppose by contradiction that this is not the case. Thus there is  $u \in U(A)$  such that

$$f(\mathbf{t}) = \operatorname{Ad}(u) \circ f(\mathbf{s}).$$

This implies that the sequence

$$(u\exp(it_1x_1)\cdots\exp(it_nx_n)\exp(-is_nx_n)\cdots\exp(-is_1x_1))_{n\in\mathbb{N}}$$

in U(A) is central, i.e. asymptotically commutes (in norm) with any element of A. Fix now any  $n_0 \in \mathbb{N}$  such that  $|t_{n_0} - s_{n_0}| \in (\frac{1}{2}, 1)$  and

$$\|b[y_n, u]\| < \eta$$

for  $n \ge n_0$ . Suppose that  $n > n_0$ . Using the fact that the elements  $\exp(it_m x_m)$  and  $\exp(it_k x_k)$  commute up to  $\eta 2^{-m}$  for  $k, m \in \mathbb{N}$ , one can show that

$$by_{n_0}u\exp(it_1x_1)\cdots\exp(it_nx_n)\exp(-is_nx_n)\cdots\exp(-is_1x_1)$$

is at distance at most  $5\eta$  from

$$buy_{n_0}\exp(i(t_{n_0}-s_{n_0})x_{n_0})\exp(it_1x_1)\cdots\exp(it_{n_0}x_{n_0})$$
  
$$\cdots\exp(it_nx_n)\exp(-is_nx_n)\cdots\exp(is_{n_0}x_{n_0})\cdots\exp(-is_1x_1),$$

where  $\exp(it_{n_0}x_{n_0})$  and  $\exp(is_{n_0}x_{n_0})$  indicate omitted terms in the product. Similarly

$$bu\exp(it_1x_1)\cdots\exp(it_nx_n)\exp(-is_nx_n)\cdots\exp(-is_1x_1)y_{n_0}$$

is at distance at most  $5\eta$  from

$$bu \exp\left(i\left(t_{n_0} - s_{n_0}\right) x_{n_0}\right) y_{n_0} \exp\left(it_1 x_1\right) \cdots \exp\left(it_{n_0} x_{n_0}\right)$$
$$\cdots \exp\left(it_n x_n\right) \exp\left(-is_n x_n\right) \cdots \exp\left(is_{n_0} x_{n_0}\right) \ldots \exp\left(-is_1 x_1\right)$$

Thus, the norm of the commutator of

$$u\exp(it_1x_1)\cdots\exp(it_nx_n)\exp(-is_nx_n)\cdots\exp(-is_1x_1)$$

and  $y_0$  is at least

$$\|b[\exp(i(t_{n_0} - s_{n_0})x_{n_0}), y_{n_0}]\| - 10\eta \ge \varepsilon - 10\eta \ge \frac{\varepsilon}{2}$$

This contradicts the fact that the sequence

$$\left(u\exp(it_1x_1)\cdots\exp(it_nx_n)\exp(-is_nx_n)\cdots\exp(-is_1x_1)\right)_{n\in\mathbb{N}}$$

is central and concludes the proof.

## 2.5 A dichotomy for derivations

If A is a C\*-algebra, then we denote as in Section 2.3 by  $\Delta_0(A)$  the separable Banach space of inner derivations of A endowed with the norm  $\|\cdot\|_{\Delta_0(A)}$  and by  $\overline{\Delta_0(A)}$  the closure of  $\Delta_0(A)$  inside the space  $\Delta(A)$  of derivations of A endowed with the operator norm. Suppose that  $E_{\Delta(A)}$  is the Borel equivalence relation on  $\overline{\Delta_0(A)}$  defined by

$$\delta_0 E_{\Delta(A)} \delta_1$$
 iff  $\delta_0 - \delta_1 \in \Delta_0(A)$ .

Observe that  $E_{\Delta(A)}$  is the orbit equivalence relation associated with the continuous action of the additive group of  $\Delta_0(A)$  on  $\overline{\Delta_0(A)}$  by translation.

**Theorem 2.5.1.** If A is a unital  $C^*$ -algebra, then the following statements are equivalent:

- 1.  $\Delta_0(A)$  is closed in  $\Delta(A)$ ;
- 2.  $E_{\Delta(A)}$  is smooth;
- 3.  $E_{\Delta(A)}$  is classifiable by countable structures;
- 4. A has continuous trace.

The equivalence of 1 and 4 follows from [67, Theorem 5.3] together with the equivalence of 1 and 3 in Theorem 2.0.18. The implication  $1 \Rightarrow 2$  follows from [44, Exercise 4.4]. Trivially 2 is stronger than 3. Finally observe that  $\Delta_0(A)$  and  $\overline{\Delta_0(A)}$  satisfy the hypothesis

of [127, Lemma 2.1]. In fact, as discussed at the beginning of Section 2.4,  $\Delta_0(A)$  endowed with the norm

$$\|\mathrm{ad}\,(ia)\|_{\Delta(A)} = \inf\left\{\|a - z\| \mid z \in A' \cap A\right\}$$

is a separable Banach space. Moreover the inclusion map of  $\Delta_0(A)$  in  $\overline{\Delta_0(A)} \subset \Delta(A)$  is bounded with norm at most 2. Thus, if  $\Delta_0(A)$  is not closed in  $\Delta(A)$ , then the continuous action of the additive group  $\Delta_0(A)$  on  $\overline{\Delta_0(A)}$  by translation is turbulent. Hjorth's turbulence theorem recalled at the beginning of Section 2.2 concludes the proof of the implication  $3 \Rightarrow 1$ .

## Chapter 3

# The Gurarij operator space

The *Gurarij space*  $\mathbb{G}$  is a Banach space first constructed by Gurarij in [53]. It has the following universal property: whenever  $X \subset Y$  are finite-dimensional Banach spaces,  $\phi : X \to \mathbb{G}$  is a linear isometry, and  $\varepsilon > 0$ , there is an injective linear map  $\psi : Y \to \mathbb{G}$  extending  $\phi$  such that  $||\psi|| ||\psi^{-1}|| < 1 + \varepsilon$ . The uniqueness of such an object was proved by Lusky [93]. A short proof was later provided by Kubis-Solecki [83]. The Gurarij space was first realized as a Fraïssé limit by Ben Yaacov in [8].

Fraïssé theory is a subject at the border between model theory and combinatorics originating from the seminar work of Fraïssé [42]. Broadly speaking, Fraïssé theory studies homogeneous structures and ways to construct them. In the discrete setting Fraïssé established in [42] a correspondence between countable homogeneous structures and what are now called Fraïssé classes. Let the *age* of a countable structure S be the collection of finitely generated substructures of S. Any Fraïssé class is the age of a countable homogeneous structure. Conversely from any Fraïssé class one can build a countable homogeneous structure that has the given class as its age. Moreover such a structure is uniquely determined up to isomorphism by this property.

This correspondence has been recently generalized in [8] by Ben Yaacov from the purely discrete setting to the setting where metric structures are considered; see also [128]. The main results of discrete Fraïssé theory are recovered in this more general framework. In particular any Fraïssé class of metric structures is the age of a separable homogeneous structure, which is unique up to isometric isomorphism. An alternative categorical-theoretic approach to Fraïssé limits in the metric setting has been developed by Kubiś [82].

The Gurarij space is the limit of the Fraïssé class of finite-dimensional Banach spaces. This has been showed in [8] building on previous work of Henson. In particular this has yielded an alternative proof of the uniqueness of the Gurarij space up to isometric isomorphism. Other naturally occurring examples of Fraïssé limits are the Urysohn universal metric space [102], the hyperfinite II<sub>1</sub> factor and the infinite type UHF C\*-algebras [22].

In this chapter we consider a noncommutative analog of the Gurarij space introduced by Oikhberg in [104] within the framework of operator spaces. Operator spaces can be regarded as noncommutative Banach spaces. In fact Banach spaces can be concretely defined as closed subspaces of C(K) spaces, where K is a compact Hausdorff space. These are precisely the *abelian* unital C\*-algebras. Replacing abelian C\*-algebras with arbitrary C\*-algebras or—equivalently—the algebra B(H) of bounded linear operators on some Hilbert space H provides the notion of an *operator space*.

An operator space  $X \subset B(H)$  is endowed with matricial norms on the algebraic tensor product  $M_n \otimes X$  obtained by the inclusion  $M_n \otimes X \subset M_n \otimes B(H) \cong B(H \oplus \cdots \oplus H)$ . A linear operator  $\phi$  between operator spaces is completely bounded with norm at most Mif all its amplifications  $id_{M_n} \otimes \phi$  are bounded with norm at most M. The notion of complete isometry is defined similarly. Operator spaces form then a category with completely bounded (or completely isometric) linear maps as morphisms. Any Banach space X has a canonical operator space structure induced by the inclusion  $X \subset C(\text{Ball}(X^*))$  where Ball  $(X^*)$  is the unit ball of the dual of X. However in this case the matricial norms do not provide any new information, and any linear map  $\phi$  between Banach spaces is automatically completely bounded with same norm. For more general operator spaces it is far from being true that any bounded linear map is completely bounded. The matricial norms play in this case a crucial role.

According to [104] an operator space is noncommutative Gurarij if it satisfies the same universal property of the Gurarij Banach space, where finite-dimensional Banach spaces are replaced with arbitrary finite-dimensional 1-exact operator spaces, and the operator norm is replaced by the completely bounded norm. The restriction to 1-exact spaces is natural since a famous result of Junge and Pisier asserts that there is no separable operator space containing all the finite-dimensional operator spaces as subspaces [66, Theorem 2.3]; see also [115, Chapter 21]. Proposition 3.1 of [104] shows that any two noncommutative Gurarij are approximately completely isometrically isomorphic. Moreover [104, Theorem 1.1] shows that separable  $\mathcal{OL}_{\infty,1+}$  spaces—as defined in [25, 65]—can be completely isometrically embedded in some noncommutative Gurarij space as completely contractively complemented subspaces. In view of [24, Theorem 4.7], Oikhberg's result implies that every separable 1-exact operator space can be completely isometrically embedded in some noncommutative Gurarij space. The main result of this chapter is that the noncommutative Gurarij space can be realized as the limit of the Fraïssé class of finite-dimensional 1-exact operator spaces. We deduce as a consequence that the noncommutative Gurarij space—which we denote by  $\mathbb{NG}$ —is unique up to complete isometry, universal among separable 1-exact operator spaces, and moreover satisfies the following homogeneity property: for any finite-dimensional subspace  $X \subset \mathbb{NG}$ , any complete isometry  $\phi : X \to \mathbb{NG}$ , and any  $\varepsilon > 0$  there is a surjective linear complete isometry  $\psi : \mathbb{NG} \to \mathbb{NG}$  such that  $\psi_{|X} - \phi$  has completely bounded norm at most  $\varepsilon$ .

This rest of this chapter is divided into three sections. Section 3.1 contains some background material on Fraïssé theory and operator spaces. We follow the presentation of Fraïssé theory for metric structures as introduced by Ben Yaacov in [8]. Similarly as [102] we adopt the slightly less general point of view—sufficient for our purposes—where one considers only structures where the interpretation of function and relation symbols are Lipschitz with a constant that does not depend on the structure. The material on operator spaces is standard and can be found for example in the monographs [115, 26, 107]. The topic of  $M_n$ -spaces is perhaps less well known and can be found in Lehrer's PhD thesis [86] as well as in [105, 104].

In Section 3.2 we show that the class of finite-dimensional  $M_n$ -spaces is a Fraïssé class. This can be seen as a first step towards proving that the class of finite-dimensional 1exact operator spaces is a Fraïssé class. Any  $M_n$ -space can be canonically endowed with a compatible operator space structure. Therefore in principle it is possible to rephrase all the arguments and results in terms of operator spaces. Nonetheless we find it more convenient and enlightening to deal with  $M_n$ -space. This allows one to recognize and use the analogy with the Banach space case.

Finally Section 3.3 contains the proof of the main result, asserting that the class of finite-dimensional 1-exact operator spaces is a Fraïssé class. Its limit is then identified as the noncommutative Gurarij space.

### 3.1 Background material

### 3.1.1 Approximate isometries

Suppose that A, B are complete metric spaces. An *approximate isometry* from A to B is a map  $\psi : A \times B \to [0, +\infty]$  satisfying the following:

$$\left|\psi\left(a,b\right)-\psi\left(a',b\right)\right| \leq d\left(a,a'\right) \leq \psi\left(a,b\right)+\psi\left(a',b\right)$$

and

$$\left|\psi\left(a,b\right)-\psi\left(a,b'\right)\right|\leq d\left(b,b'
ight)\leq\psi\left(a,b
ight)+\psi\left(a,b'
ight)$$

for every  $a, a' \in A$  and  $b, b' \in B$ . If  $\psi$  is an approximate isometry from A to B, then we write  $\psi : A \rightsquigarrow B$ . The set of all approximate isometries from A to B is denoted by Apx (A, B). This is a compact space endowed with the product topology from  $[0, +\infty]^{A \times B}$ . A partial isometry f from A to B is an isometry from a subset dom (f) of A to B.

**Remark 3.1.1.** Any partial isometry f will be identified with the approximate isometry  $\psi_f$  given by the distance function from the graph of f.

Explicitly  $\psi_f$  is defined by

$$\psi_f(a,b) = \inf_{z \in \operatorname{dom}(f)} \left( d\left(a,z\right) + d\left(f(z),b\right) \right).$$

If  $\psi : A \rightsquigarrow B$  one can consider its *pseudo-inverse*  $\psi^* : B \rightsquigarrow A$  defined by  $\psi^* (b, a) = \psi (a, b)$ . Moreover one can take *composition* of approximate isometries  $\psi : A \rightsquigarrow B$  and  $\phi : B \rightsquigarrow C$  by setting

$$\left(\phi\psi\right)\left(a,c\right) = \inf_{b\in B} \left(\psi\left(a,b\right) + \phi\left(b,c\right)\right).$$

These definitions are consistent with composition and inversion of partial isometries when regarded as approximate isometries.

If  $A_0 \subset A$ ,  $B_0 \subset B$ , and  $\psi : A \rightsquigarrow B$  then one can define the *restriction*  $\psi_{|A_0 \times B_0} = j^* \psi i$ :  $A_0 \rightsquigarrow B_0$  where *i* and *j* are the inclusion maps of  $A_0$  into *A* and  $B_0$  into *B*. Conversely if  $\phi : A_0 \rightsquigarrow B_0$  then one can consider its *trivial extension*  $j\phi i^* : A \rightsquigarrow B$ . This allows one to regard Apx  $(A_0, B_0)$  as a subset of Apx (A, B) by identifying an approximate isometry with its trivial extension.

For approximate isometries  $\phi, \psi : A \rightsquigarrow B$  we say that  $\phi$  refines  $\psi$  and  $\psi$  coarsens  $\phi$  written  $\phi \leq \psi$ —if  $\phi(a, b) \leq \psi(a, b)$  for every  $a \in A$  and  $b \in B$ . The set of approximate isometries that refine  $\psi$  is denoted by  $\operatorname{Apx}^{\leq \psi}(A, B)$ . The interior of  $\operatorname{Apx}^{\leq \psi}(A, B)$  is denote by  $\operatorname{Apx}^{<\psi}(A, B)$ . The closure under coarsening  $\mathcal{A}^{\uparrow}$  of a set  $\mathcal{A} \subset \operatorname{Apx}(A, B)$  is the collection of  $\phi \in \operatorname{Apx}(A, B)$  that coarsen some element of  $\mathcal{A}$ .

### 3.1.2 Languages and structures

A language  $\mathcal{L}$  is given by sets of predicate symbols and of function symbols. Every symbol has two natural numbers attached: its arity and its Lipschitz constant. An  $\mathcal{L}$ -structure  $\mathfrak{A}$ is given by

- a complete metric space A,
- a  $c_B$ -Lipschitz function  $B^{\mathfrak{A}} : A^{n_B} \to \mathbb{R}$  for every predicate symbol B, where  $c_B$  is the Lipschitz constant of B and  $n_B$  is the arity of B, and
- a  $c_f$ -Lipschitz function  $f^{\mathfrak{A}}: A^{n_f} \to A$  for every function symbol f, where  $c_f$  is the Lipschitz constant of f and  $n_f$  is the arity of f.

Here and in the following we assume the power  $A^n$  to be endowed with the max metric  $d(\bar{a}, \bar{b}) = \max_i d(a_i, b_i)$ . An *embedding* of  $\mathcal{L}$ -structures  $\phi : \mathfrak{A} \to \mathfrak{B}$  is a function that commutes with the interpretation of all the predicate and function symbols. An *isomorphism* is a surjective embedding. An *automorphism* of  $\mathfrak{A}$  is an isomorphism from  $\mathfrak{A}$  to  $\mathfrak{A}$ . If  $\bar{a}$  is a finite tuple in  $\mathfrak{A}$  then  $\langle \bar{a} \rangle$  denotes the smallest substructure of  $\mathfrak{A}$  containing  $\bar{a}$ . A *partial isomorphism*  $\phi : \mathfrak{A} \dashrightarrow \mathfrak{B}$  is an embedding from  $\langle \bar{a} \rangle$  to  $\mathfrak{B}$  for some finite tuple  $\bar{a}$  in  $\mathfrak{A}$ . An  $\mathcal{L}$ -structure  $\mathfrak{A}$  is *finitely generated* if  $\mathfrak{A} = \langle \bar{a} \rangle$  for some finite tuple  $\bar{a}$  in  $\mathfrak{A}$ .

We will assume that the language  $\mathcal{L}$  contains a distinguished binary predicate symbol to be interpreted as the metric. In particular this ensures that all the embeddings and (partial) isomorphisms are (partial) isometries. Therefore consistently with the convention from Remark 3.1.1 partial isomorphisms will be regarded as approximate isometries.

**Definition 3.1.2.** Suppose that C is a class of finitely-generated  $\mathcal{L}$ -structure. We say that C satisfies

- the hereditary property (HP) if  $\langle \bar{a} \rangle \in \mathcal{C}$  for every  $\mathfrak{A} \in \mathcal{C}$  and finite tuple  $\bar{a} \in \mathfrak{A}$ ,
- the joint embedding property (JEP) if for any  $\mathfrak{A}, \mathfrak{B} \in \mathcal{C}$  there is  $\mathfrak{C} \in \mathcal{C}$  and embeddings  $\phi : \mathfrak{A} \to \mathfrak{C}$  and  $\psi : \mathfrak{B} \to \mathfrak{C}$ ,
- the near amalgamation property (NAP) if, whenever  $\mathfrak{A} \subset \mathfrak{B}_0$  and  $\mathfrak{B}_1$  are elements of  $\mathcal{C}, \phi : \mathfrak{A} \to \mathfrak{B}_1$  is an embedding,  $\bar{a}$  is a finite tuple in  $\mathfrak{A}$ , and  $\varepsilon > 0$ , there exists  $\mathfrak{C} \in \mathcal{C}$  and embeddings  $\psi_0 : \mathfrak{B}_0 \to \mathfrak{C}$  and  $\psi_1 : \mathfrak{B}_1 \to \mathfrak{C}$  such that

$$d\left(\psi_0(\bar{a}), \left(\psi_1 \circ \varphi\right)(\bar{a})\right) \le \varepsilon.$$

• the amalgamation property (AP) if it satisfies (NAP) even when one takes  $\varepsilon = 0$ .

### 3.1.3 Fraïssé classes and limits

Suppose in the following that C is a class of finitely generated  $\mathcal{L}$ -structures satisfying (HP), (JEP), and (NAP).

**Definition 3.1.3.** A C-structure is an  $\mathcal{L}$ -structure  $\mathfrak{A}$  such that  $\langle \bar{a} \rangle \in C$  for every finite tuple  $\bar{a}$  in  $\mathfrak{A}$ .

Let  $\mathfrak{A}$  and  $\mathfrak{B}$  be  $\mathcal{C}$ -structures. Define  $\operatorname{Apx}_{1,\mathcal{C}}(\mathfrak{A},\mathfrak{B}) \subset \operatorname{Apx}(A,B)$  to be the set of all partial isomorphisms from  $\mathfrak{A}$  to  $\mathfrak{B}$ . Define  $\operatorname{Apx}_{2,\mathcal{C}}(\mathfrak{A},\mathfrak{B})$  to be the set of approximate isometries  $\phi: A \rightsquigarrow B$  of the form

$$\phi = g^* f$$

where  $f \in \operatorname{Apx}_{1,\mathcal{C}}(\mathfrak{A},\mathfrak{C})$  and  $g \in \operatorname{Apx}_{1,\mathcal{C}}(\mathfrak{B},\mathfrak{C})$  for some  $\mathcal{C}$ -structure  $\mathfrak{C}$ . Finally set

$$\operatorname{Apx}_{\mathcal{C}}(\mathfrak{A},\mathfrak{B}) = \operatorname{Apx}_{2,\mathcal{C}}(\mathfrak{A},\mathfrak{B})^{\uparrow}$$

Elements of  $\operatorname{Apx}_{\mathcal{C}}(\mathfrak{A},\mathfrak{B})$  are called (*C*-intrinsic) approximate morphism. A (*C*-intrinsic) strictly approximate morphism from  $\mathfrak{A}$  to  $\mathfrak{B}$  is an approximate morphism  $\phi$  such that the interior  $\operatorname{Apx}_{\mathcal{C}}^{<\phi}(\mathfrak{A},\mathfrak{B})$  of  $\operatorname{Apx}_{\mathcal{C}}^{\leq\phi}(\mathfrak{A},\mathfrak{B})$  is nonempty. The set of strictly approximate morphisms from  $\mathfrak{A}$  to  $\mathfrak{B}$  is denoted by  $\operatorname{Stx}_{\mathcal{C}}(\mathfrak{A},\mathfrak{B})$ .

Fix  $k \in \mathbb{N}$  and denote by  $\mathcal{C}(k)$  the set of pairs  $(\bar{a}, \mathfrak{A})$  where  $\mathfrak{A} \in \mathcal{C}$  and  $\bar{a}$  is a finite tuple in  $\mathfrak{A}$  such that  $\mathfrak{A} = \langle \bar{a} \rangle$ . Two such pairs  $(\bar{a}, \mathfrak{A})$  and  $(\bar{b}, \mathfrak{B})$  are identified if there is an isomorphism  $\phi : \mathfrak{A} \to \mathfrak{B}$  such that  $\phi(\bar{a}) = \bar{b}$ . By abuse of notation we will denote  $(\bar{a}, \mathfrak{A})$  simply by  $\bar{a}$ .

**Definition 3.1.4.** The Fraïssé metric  $d_{\mathcal{C}}$  on  $\mathcal{C}(k)$  is defined by

$$d_{\mathcal{C}}(\bar{a}, \bar{b}) = \inf_{\phi} \max_{i} \phi(a_i, b_i)$$

where  $\phi$  ranges in Apx<sub> $\mathcal{C}$ </sub>( $\langle \bar{a} \rangle, \langle \bar{b} \rangle$ ) or, equivalently, in Stx<sub> $\mathcal{C}$ </sub>( $\langle \bar{a} \rangle, \langle \bar{b} \rangle$ ).

Such a metric can be equivalently described in terms of embeddings:

$$d_{\mathcal{C}}(\bar{a}, \bar{b}) = \inf_{f,g} d(f(\bar{a}), g(\bar{b}))$$

where f, g range over all the embeddings of  $\langle \bar{a} \rangle$  and  $\langle \bar{b} \rangle$  into a third structure  $\mathfrak{C} \in \mathcal{C}$ .

**Definition 3.1.5.** Suppose that C is a class of finitely-generated  $\mathcal{L}$ -structures satisfying (HP), (JEP), and (NAP) from Definition 3.1.2. We say that C is a *Fraissé class* if the metric space  $(C(k), d_{\mathcal{C}})$  is complete and separable for every  $k \in \mathbb{N}$ .

**Remark 3.1.6.** In [8, Definition 2.12] a Fraïssé class is moreover required to satisfy the Continuity Property. Such a property is automatically satisfied in our more restrictive

setting, where we assume that the interpretation of any symbol from  $\mathcal{L}$  is a Lipschitz function with Lipschitz constant that does not depend from the structure.

**Definition 3.1.7.** Suppose that  $\mathcal{C}$  is a Fraïssé class. A limit of  $\mathcal{C}$  is a separable  $\mathcal{C}$ -structure  $\mathfrak{M}$  satisfying the following property: For every  $\mathfrak{A} \in \mathcal{C}$ , finite tuple  $\bar{a}$  in  $\mathfrak{A}$ , embedding  $\phi : \langle \bar{a} \rangle \to \mathfrak{M}$ , and  $\varepsilon > 0$  there is an embedding  $\psi : \mathfrak{A} \to \mathfrak{M}$  such that  $d(\psi(\bar{a}), \phi(\bar{a})) < \varepsilon$ .

The definition given above is equivalent to [8, Definition 2.14] in view of [8, Corollary 2.20].

**Definition 3.1.8.** An  $\mathcal{L}$ -structure  $\mathfrak{M}$  is *homogeneous* if for every finite tuple  $\bar{a}$  in  $\mathfrak{A}$ , embedding  $\phi : \langle \bar{a} \rangle \to \mathfrak{A}$ , and  $\varepsilon > 0$ , there is an automorphism  $\psi$  of  $\mathfrak{M}$  such that  $d(\phi(\bar{a}), \psi(\bar{a})) < \varepsilon$ .

The following theorem is a combination of the main results from [8].

**Theorem 3.1.9** (Ben Yaacov). Suppose that C is a Fraïssé class. Then C has a limit  $\mathfrak{M}$ . If  $\mathfrak{M}'$  is another limit of C then  $\mathfrak{M}$  and  $\mathfrak{M}'$  are isomorphic as  $\mathcal{L}$ -structures. Moreover  $\mathfrak{M}$  is homogeneous and contains any separable C-structure as a substructure.

**Remark 3.1.10.** Suppose that C is a Fraïssé class. Assume that A is a class of separable C-structure with the following properties:

- $\mathcal{A}$  is closed under isomorphism and countable direct limits, and
- every separable C-structure embeds into an element of A.

It is clear from the proof of [8, Lemma 2.17] that the Fraissé limit  $\mathfrak{M}$  of  $\mathcal{C}$  belongs to  $\mathcal{A}$ .

### **3.1.4** Operator spaces

An operator space is a closed subspace of B(H). Here and in the following we denote by  $M_n$  the algebra of  $n \times n$  complex matrices. If X is a complex vector space, then we denote by  $M_n \otimes X$  the algebraic tensor product. Observe that this can be canonically identified with the space  $M_n(X)$  of  $n \times n$  matrices with entries from X. If  $X \subset B(H)$  is an operator space, then  $M_n(X)$  is naturally endowed with a norm given by the inclusion  $M_n(X) \subset B(H \oplus \cdots \oplus H)$ . If  $\phi: X \to Y$  is a linear map between operator spaces, then its *n*-th amplification is the linear map  $id_{M_n} \otimes \phi: M_n \otimes X \to M_n \otimes Y$ . Under the identification of  $M_n \otimes X$  with  $M_n(X)$  and  $M_n \otimes Y$  with  $M_n(Y)$ , the map  $id_{M_n} \otimes \phi$  is defined by

$$(id_{M_n} \otimes \phi) [x_{ij}] = [\phi(x_{ij})]$$

We say that  $\phi$  is completely bounded if

$$\sup_{n} \|id_{M_n} \otimes \phi\| < +\infty.$$

In such case we define its completely bounded norm

$$\|\phi\|_{cb} = \sup_n \|id_{M_n}\otimes \phi\|$$
 .

A linear map  $\phi$  is a complete contraction if  $id_{M_n} \otimes \phi$  is a contraction for every  $n \in \mathbb{N}$ . It is a complete isometry if  $id_{M_n} \otimes \phi$  is an isometry for every  $n \in \mathbb{N}$ . Finally it is a completely isometric isomorphism if  $id_{M_n} \otimes \phi$  is an isometric isomorphism for every  $n \in \mathbb{N}$ .

Operator spaces admit an abstract characterization due to Ruan [123]. A vector space X is matrix-normed if for every  $n \in \mathbb{N}$  the space  $M_n(X)$  is endowed with a norm such that whenever  $\alpha \in M_{k,n}$ ,  $x \in M_n(X)$ , and  $\beta \in M_{n,k}$ 

$$\left\|\alpha.x.\beta\right\|_{k} \le \left\|\alpha\right\| \left\|x\right\| \left\|\beta\right\|$$

where  $\alpha . x . \beta$  denotes the matrix product, and  $\|\alpha\|$ ,  $\|\beta\|$  are the norms of  $\alpha$  and  $\beta$  regarded as operators between finite-dimensional Hilbert spaces. A matrix-normed vector space is  $L^{\infty}$ -matrix-normed provided that

$$\left\| \begin{bmatrix} x & 0 \\ 0 & y \end{bmatrix} \right\| = \max\left\{ \|x\|, \|y\| \right\}$$

for  $x \in M_n(X)$  and  $y \in M_m(X)$ . Every operator space  $X \subset B(H)$  is canonically an  $L^{\infty}$ -matrix-normed space. Ruan's theorem asserts that, conversely, any  $L^{\infty}$ -matrix-normed space is completely isometrically isomorphic to an operator space [107, Theorem 13.4].

Equivalently one can think of an operator system X as a structure on  $\mathcal{K}_0 \otimes X$ ; see [115, Section 2.2]. Suppose that H is the separable Hilbert space with a fixed orthonormal basis  $(e_k)_{k \in \mathbb{N}}$ . Let  $P_n$  be the orthogonal projection of span  $\{e_1, \ldots, e_n\}$  for  $n \in \mathbb{N}$ . We can identify  $M_n$  with the subspace of  $A \in B(H)$  such that  $AP_n = P_n A = A$ . The direct union  $\mathcal{K}_0 = \bigcup_n M_n$  is a subspace of B(H). We can identify  $\bigcup_n M_n(X)$  with  $\mathcal{K}_0[X] \cong \mathcal{K}_0 \otimes X$ . Then  $\mathcal{K}_0[X]$  is a complex vector space with a natural structure of  $\mathcal{K}_0$ -bimodule. In formulas if  $\alpha, \beta \in \mathcal{K}_0$  and  $x = \sum_i \gamma_i \otimes y_i \in \mathcal{K}_0[X]$ , then

$$\alpha.x.\beta = \sum_{i} \alpha \gamma_{i} \beta \otimes y_{i} \in \mathcal{K}_{0}[X].$$

In this framework one can reformulate Ruan's axioms as follows; see [115, Page 35]. If  $\alpha_i, \beta_i \in \mathcal{K}_0$  and  $x_i \in \mathcal{K}_0[X]$  are finite sequences then

$$\left\|\sum_{i} \alpha_{i} \cdot x_{i} \cdot \beta_{i}\right\| \leq \left\|\sum_{i} \alpha_{i}^{*} \alpha_{i}\right\|^{\frac{1}{2}} \max_{i} \left\|x_{i}\right\| \left\|\sum_{i} \beta_{i} \beta_{i}^{*}\right\|^{\frac{1}{2}}.$$

The metric on  $\mathcal{K}_0[X]$  is not complete. Nonetheless one can pass to the completion  $\mathcal{K} \otimes X$ and extend all the operations. (Here  $\mathcal{K}$  is the closure of  $\mathcal{K}_0$  inside B(H), i.e. the ideal of compact operators.)

The abstract characterization of operator spaces mentioned above allows one to regard operator spaces as  $\mathcal{L}_{OS}$ -structures for a suitable language  $\mathcal{L}_{OS}$ . Denote by

$$\mathcal{K}_0(\mathbb{Q}(i)) = \bigcup_{n \in \mathbb{N}} M_n(\mathbb{Q}(i))$$

the space of matrices with coefficients in the field of Gauss rationals  $\mathbb{Q}(i)$ . Then  $\mathcal{L}_{OS}$  contains, in addition to the special symbol d for the metric, a symbol + for the addition in  $\mathcal{K} \otimes X$ , a constant 0 for the zero vector in  $\mathcal{K} \otimes X$ , function symbols  $\sigma_{\alpha,\beta}$  for  $\alpha, \beta \in \mathcal{K}_0(\mathbb{Q}(i))$ for the bimodule operation. The Lipschitz constant for the symbol + is 2, while the Lipschitz constant of  $\sigma_{\alpha,\beta}$  is  $\|\alpha\| \|\beta\|$ . An alternative description of operator spaces as metric structures—which fits in the framework of continuous logic [9, 35]—has been provided in [30, Section 3.3] and [52, Appendix B].

It is worth noting that the space X can be described as the set of  $x \in \mathcal{K} \otimes X$  such that 1.x = x. Moreover a linear map  $\phi : \mathcal{K} \otimes X \to \mathcal{K} \otimes Y$  that respects the  $\mathcal{K}_0$ -bimodule operations satisfies

$$\phi\left(\sum_{i}\gamma_{i}\otimes x_{i}\right)=\sum_{i}\gamma_{i}\otimes\phi(x)$$

and therefore is the amplification of a linear map from X to Y. Hence when operator spaces are regarded as  $\mathcal{L}_{OS}$ -structures, embeddings and isomorphisms as defined in Subsection 3.1.2 correspond, respectively, to completely isometric linear maps and completely isometric linear isomorphisms.

If X and Y are operator spaces, then the space CB(X,Y) of completely bounded linear maps from X to Y is canonically endowed with an operator space structure obtained by identifying isometrically  $M_n(CB(X,Y))$  with  $CB(M_n(X), M_n(Y))$  with the completely bounded norm. Any linear functional  $\phi$  on an operator system X is automatically completely bounded with  $\|\phi\|_{cb} = \|\phi\|$ . Therefore the dual space  $X^*$  of X can be regarded as
the operator space  $CB(X, \mathbb{C})$ .

If X and Y are operator spaces, then their  $\infty$ -sum  $X \oplus^{\infty} Y$  is the operator system supported on the algebraic direct sum  $X \oplus Y$  endowed with norms

$$\|(x,y)\|_{M_n(X\oplus^{\infty}Y)} = \max\left\{\|x\|_{M_n(X)}, \|y\|_{M_n(Y)}\right\}.$$

The  $\infty$ -sum of a sequence of operator spaces is defined analogously.

The 1-sum  $X \oplus^1 Y$  is the operator system obtained by identifying  $X \oplus Y$  with  $(X^* \oplus^\infty Y^*)^*$ . In formulas if  $x, y \in M_n (X \oplus Y)$  then

$$\begin{aligned} \|(x,y)\|_{M_n(X\oplus^1Y)} &= \|(x,y)\|_{CB(X^*\oplus Y^*,M_n)} \\ &= \sup \|(id_{M_n}\otimes\phi)(x) + (id_{M_n}\otimes\psi)(y)\|_{M_n\otimes M_k} \end{aligned}$$

where  $\phi, \psi$  range over the unit balls of  $CB(X^*, M_k)$  and  $CB(Y^*, M_k)$  and k ranges in  $\mathbb{N}$ . Equivalently the norm on  $X \oplus^1 Y$  can be described as

$$\left\|\left(x,y\right)\right\| = \sup_{u,v} \left\|\left(id_{M_{n}}\otimes u\right)\left(x\right) + \left(id_{M_{n}}\otimes v\right)\left(y\right)\right\|$$

where u, v range over all completely contractive maps from X and Y into B(H); see [115, Section 2.6]. In analogous fashion one can define the 1-sum and the  $\infty$ -sum of a sequence of operator spaces.

We denote the sum

$$\overbrace{\mathbb{C}\oplus^1\mathbb{C}\oplus^1\dots\oplus^1\mathbb{C}}^{n \text{ times}}$$

by  $\ell^1(n)$  and the sum

$$\overbrace{\mathbb{C}\oplus^{\infty}\mathbb{C}\oplus^{\infty}\cdots\oplus^{\infty}\mathbb{C}}^{n \text{ times}}\mathbb{C}$$

by  $\ell^{\infty}(n)$ . Moreover we denote by  $\bar{e} = (e_i)$  the canonical basis of  $\ell^1(n)$  and by  $\bar{e}^*$  its dual basis of  $\ell^{\infty}(n)$ .

### 3.1.5 $M_n$ -spaces

In this subsection we recall the definition and basic properties of  $M_n$ -spaces as defined in [86, Chapter I]. A matricial *n*-norm on a space X is a norm on  $M_n(X)$  such that

$$\|\alpha.x.\beta\| \le \|\alpha\| \|x\| \|\beta\|$$

for  $\alpha, \beta \in M_n$  and  $x \in M_n(X)$ . Such a norm induces a norm on  $M_k(X)$  for  $k \leq n$  via the inclusion

$$x \mapsto \begin{bmatrix} x & 0 \\ 0 & 0 \end{bmatrix}$$

An  $L^{\infty}$ -matrix-n-norm is a matricial n-norm satisfying moreover

$$\left\| \begin{bmatrix} x & 0 \\ 0 & y \end{bmatrix} \right\| = \max\left\{ \|x\|, \|y\| \right\}$$

for  $k \leq n, x \in M_k(X)$ , and  $y \in M_{n-k}(Y)$ .

Observe that  $M_n$  has a natural *n*-norm obtained by identifying  $M_n(M_n)$  with  $M_n \otimes M_n$ (spatial tensor product). More generally if K is a compact Hausdorff space then  $C(K, M_n)$ is *n*-normed by identifying  $M_n(C(K, M_n))$  with  $C(K, M_n \otimes M_n)$ . In particular  $\ell^{\infty}(\mathbb{N}, M_n)$ has a natural *n*-norm obtained by the identification with  $C(\beta \mathbb{N}, M_n)$ .

If X, Y are *n*-normed spaces, then a linear map  $\phi : X \to Y$  is *n*-bounded if  $id_{M_n} \otimes \phi$  $\phi : M_n(X) \to M_n(Y)$  is bounded, and  $\|\phi\|_n$  is by definition  $\|id_{M_n} \otimes \phi\|$ . The notions of *n*-contraction and *n*-isometry are defined similarly. Let nB(X,Y) be the space of *n*bounded linear functions from X to Y with norm  $\|\cdot\|_n$ . Identifying  $M_n(X^*)$  with  $nB(X,\mathbb{C})$ isometrically defines an  $L^{\infty}$ -matrix-*n*-norm on the dual  $X^*$  of X. The same argument allows one to define an  $L^{\infty}$ -matrix-norm on the second dual  $X^{**}$ .

An  $L^{\infty}$ -matricially-*n*-normed space is called an  $M_n$ -space if it satisfies any of the following equivalent definitions—see [86, Théorème I.1.9]:

- 1. There is an *n*-isometry from X to B(H);
- 2. The canonical inclusion of X into  $X^{**}$  is isometric;

3. 
$$\|\sum_{i} \alpha_{i} x_{i} \beta_{i}\| \leq \|\sum_{i} \alpha_{i} \alpha_{i}^{*}\|^{\frac{1}{2}} \max_{i} \|x_{i}\| \|\sum_{i} \beta_{i}^{*} \beta_{i}\|^{\frac{1}{2}}$$
 for any  $x_{i} \in M_{n}(X), \alpha_{i}, \beta_{i} \in M_{n};$ 

4. there is an *n*-isometry from X to  $C(X, M_n)$  for some compact Hausdorff space K.

Clearly the case n = 1 gives the usual notion of Banach space. Characterization (3) allows one to show that  $M_n$ -spaces can be seen as structures in a suitable language  $\mathcal{L}_{M_n}$ . This is the same as the language for operator space described in Subsection 3.1.4 where one replaces  $\mathcal{K}_0$  with  $M_n$ . When  $M_n$ -spaces are regarded as  $\mathcal{L}_{M_n}$ -spaces, embeddings and isomorphisms as defined in Subsection 3.1.2 correspond, respectively, to *n*-isometric linear maps and *n*-isometric linear isomorphisms.

The notions of quotient and subspace of an  $M_n$ -space can be defined analogously as in the case of operator spaces. Similarly the constructions of 1-sum and  $\infty$ -sum can be performed in this context. More details can be found in [86, Section I.2]. We will use the same notations for the 1-sum and  $\infty$ -sum of operator spaces and  $M_n$ -spaces. This will be clear from the context and no confusion should arise.

For later use we recall the following observation; see [86, Remarque I.1.5]. Suppose that X is a finite-dimensional  $M_n$ -space,  $\bar{b}$  is a normalized basis of X, and  $\bar{b}'$  is its dual basis. Assume that  $\bar{b}'$  is also normalized. Then the *n*-norm on X admits the following expression:

$$\left\|\sum_{i} \alpha_{i} \otimes b_{i}\right\| = \sup\left\{\left\|\sum_{i} \alpha_{i} \otimes \beta_{i}\right\| : \beta_{i} \in M_{n}, \left\|\sum_{i} \alpha_{i} \otimes b_{i}'\right\| \le 1\right\}$$

for  $\alpha_i \in M_n$ . In particular if  $\bar{e}$  is the canonical basis of  $\ell^1(n)$  with dual basis  $\bar{e}'$  of  $\ell^{\infty}(n)$ , then we obtain

$$\left\|\sum_{i} \alpha_{i} \otimes e_{i}\right\| = \sup\left\{\left\|\sum_{i} \alpha_{i} \otimes \beta_{i}\right\| : \beta_{i} \in M_{n}, \|\beta_{i}\| \leq 1\right\}.$$

Similar expressions hold for the matrix norms in operator spaces; see [115].

In the following we will often use tacitly the fact that finite-dimensional  $M_n$ -spaces can be approximated by subspaces of finite  $\infty$ -sums of copies of  $M_n$ .

**Lemma 3.1.11.** Suppose that X is a finite-dimensional  $M_n$ -space and  $\varepsilon > 0$ . Then there is  $k \in \mathbb{N}$  and an injective linear n-contraction

$$\phi: X \to \overbrace{M_n \oplus^\infty \cdots \oplus^\infty M_n}^{k \ times}$$

such that  $||\phi^{-1}||_n \leq 1 + \varepsilon$ .

In Lemma 3.1.11 the map  $\phi$  is not assumed to be injective. The expression  $||\phi^{-1}||_n$  denotes the *n*-norm of  $\phi^{-1}$  when regarded as a map from the range of  $\phi$  to X. Similar conventions will be adopted in the rest of this dissertation. We conclude by recalling that the natural analog of the Hahn-Banach theorem holds for  $M_n$ -spaces. Such an analog asserts that  $M_n$  is an injective element in the category of  $M_n$ -spaces with *n*-contractive maps as morphisms; see [86, Proposition I.1.16].

# **3.2** The Fraissé class of finite-dimensional $M_n$ -spaces

The purpose of this section is to show that the class  $\mathcal{M}_n$  of finite-dimensional  $\mathcal{M}_n$ -spaces is a complete Fraïssé class as in Definition 3.1.5. This will allow us to consider the corresponding Fraïssé limit as in Theorem 3.1.9. The case n = 1 of these results recovers the already known fact that finite-dimensional Banach spaces form a complete Fraïssé class. This has been shown by Ben Yaacov [8, Section 3.3] building on previous works of Henson (unpublished) and Kubis-Solecki [83]. For Banach spaces the limit is the Gurarij Banach space, introduced by Gurarij in [53] and proved to be unique up to isometric isomorphism by Lusky in [93].

### 3.2.1 Amalgamation property

The properties (JEP) and (HP) as in Definition 3.1.2 are clear for  $\mathcal{M}_n$ . We now show that  $\mathcal{M}_n$  has (AP). The proof is analogous to the one for Banach spaces, and consists in showing that the category of finite-dimensional  $\mathcal{M}_n$ -spaces has pushouts; see [46, Lemma 2.1].

**Lemma 3.2.1.** Suppose that  $X_0 \subset X$  and Y are  $M_n$ -spaces,  $\delta \ge 0$ , and  $f: X \to Y$  is a linear injective map such that  $\|f\|_n \le 1+\delta$  and  $\|f^{-1}\|_n \le 1+\delta$ . Then there is an  $M_n$ -space Z and n-isometric linear maps  $i: X \to Z$  and  $j: Y \to Z$  such that  $\|j \circ f - i\|_n \le \delta$ .

*Proof.* Define  $\delta X_0$  to be the  $M_n$ -space structure on  $X_0$  given by the norm

$$\|[x_{ij}]\|_{M_n(\delta X)} = \delta \|[x_{ij}]\|_{M_n(X)}$$

Let  $\widehat{Z}$  be the 1-sum  $X \oplus^1 Y \oplus^1 \delta X_0$ , and Z be the quotient of  $\widehat{Z}$  by the subspace

$$N = \left\{ (-z, f(z), z) \in \widehat{Z} : z \in X_0 \right\}.$$

Finally let  $i: X \to Z$  and  $j: Y \to Z$  the embeddings given by

$$x \mapsto (x, 0, 0) + N$$

and

$$y \mapsto (0, y, 0) + N$$

We claim that i and j satisfy the desired conclusions. In fact it is clear that i and j are *n*-contractions such that  $||i \circ f - j||_n \leq \delta$ . We will now show that i is an *n*-isometry. The proof that j is an *n*-isometry is similar. Suppose that  $x \in M_n(X)$  consider a linear *n*-contraction  $\phi: X \to M_n$  such that  $\|\phi(x)\|_{M_n \otimes M_n} = \|x\|_{M_n(X)}$ . Observe that

$$\frac{1}{1+\delta} \left( \phi \circ f^{-1} \right) : f \left[ X \right] \to M_n$$

is an linear *n*-contraction and hence it extends to a linear *n*-contraction  $\psi : f[X] \to M_n$ . Similarly the map

$$\frac{\delta}{1+\delta}\phi:\delta X_0\to M_n$$

is a linear *n*-contraction. Therefore we have that for every  $z \in M_n(X_0)$ 

$$\begin{aligned} &\|(x-z,f(z),z)\|_{M_n(X\oplus^1Y\oplus^1\delta X_0)}\\ \geq & \left\|\phi\left(x-z\right)+\psi(z)+\frac{\delta}{1+\delta}\phi(z)\right\|_{M_n\otimes M_n}\\ &= & \left\|\phi(x)-\phi(z)+\frac{1}{1+\delta}\phi(z)+\frac{\delta}{1+\delta}\phi(z)\right\|_{M_n\otimes M_n}\\ &= &\|\phi(x)\|_{M_n\otimes M_n}=\|x\|_{M_n(X)}\,.\end{aligned}$$

This concludes the proof that i is an n-isometry.

In particular Lemma 3.2.1 for  $\delta = 0$  shows that the class  $\mathcal{M}_n$  has (AP).

### 3.2.2 The Fraïssé metric space

We fix now  $k \in \mathbb{N}$  and consider the space  $\mathcal{M}_n(k)$  of pairs  $(\bar{a}, X)$  such that X is a kdimensional  $\mathcal{M}_n$ -space and  $\bar{a}$  is a linear basis of X. Two such pairs  $(\bar{a}, X)$  and  $(\bar{b}, Y)$  are identified if there is an n-isometry  $\phi$  from X to Y such that  $\phi(\bar{a}) = \bar{b}$ . For simplicity we will write an element  $(\bar{a}, X)$  of  $\mathcal{M}_n(k)$  simply by  $\bar{a}$ , and denote X by  $\langle \bar{a} \rangle$ . Our goal is to compute the Fraïssé metric in  $\mathcal{M}_n(k)$  as in Definition 3.1.4. The following result gives an equivalent characterization of such a metric. The case n = 1 is a result of Henson (unpublished) that can be found in [8, Fact 3.2]. We denote by  $\ell^1(k)$  the k-fold 1-sum of  $\mathbb{C}$  by itself in the category of  $\mathcal{M}_n$ -spaces with canonical basis  $\bar{e}$ . An explicit formula for the corresponding norm has been recalled at the end of Section 3.1.5.

**Proposition 3.2.2.** Suppose that  $\bar{a}, \bar{b} \in \mathcal{M}_n(k)$  and M > 0. The following statements are equivalent:

1.  $d_{\mathcal{M}_n}(\bar{a}, \bar{b}) \leq M;$ 

2. For every n-contractive  $u : \langle \bar{a} \rangle \to M_n$  there is an n-contractive  $v : \langle \bar{b} \rangle \to M_n$  such that the linear function  $w : \ell^1(k) \to M_n$  defined by  $w(e_i) = u(a_i) - v(b_i)$  has n-norm at most M, and vice versa.

*Proof.* After normalizing we can assume that M = 1. We will denote  $\langle \bar{a} \rangle$  by X and  $\langle b \rangle$  by Y.

(1) $\Rightarrow$ (2) Suppose that  $d_{\mathcal{M}_n}(\bar{a}, \bar{b}) \leq 1$ . Then there are *n*-isometries  $\phi : \langle \bar{a} \rangle \to Z$  and  $\psi : \langle \bar{b} \rangle \to Z$  for some  $M_n$ -space Z such that  $\|\phi(a_i) - \psi(b_i)\| \leq 1$  for every  $i \leq k$ . Suppose that  $u : X \to M_n$  is *n*-contractive. Consider the *n*-contractive map  $u \circ \phi^{-1}$ :  $\phi[X] \to M_n$ . By injectivity of  $M_n$  there is an *n*-contractive map  $\eta : Z \to M_n$ extending  $u \circ \phi^{-1}$ . Define  $v = \eta \circ \psi : Y \to M_n$  and observe that it is *n*-contractive. Define now  $w : \ell^1(k) \to M_n$  by  $w(e_i) = u(a_i) - v(b_i)$ . We claim that w is *n*-contractive. In fact

$$\left\|\eta\left(\phi(e_i) - \psi(e_i)\right)\right\| \le 1$$

for every  $i \leq k$ . Therefore if  $\alpha_i \in M_n$ 

$$\left\| (id_{M_n} \otimes w) \left( \sum_i \alpha_i \otimes e_i \right) \right\| = \left\| \sum_i \alpha_i \otimes \eta \left( \phi(e_i) - \psi(e_i) \right) \right\|$$
$$\leq \left\| \sum_i \alpha_i \otimes e_i \right\|_{M_n(\ell^1(k))}.$$

The vice versa is proved analogously.

(2) $\Rightarrow$ (1) Conversely suppose that for every *n*-contractive  $u : X \to M_n$  there is an *n*-contractive  $v : Y \to M_n$  such that the linear function  $w : \ell^1(k) \to M_n$  defined by  $w(e_i) = u(a_i) - v(b_i)$  is *n*-contractive, and vice versa. Define  $\widehat{Z}$  to be

$$X \oplus^1 Y \oplus^1 \ell^1(k)$$

Denote by N the closed subspace

$$\left\{ \left( -\sum_{i} \lambda_{i} a_{i}, \sum_{i} \lambda_{i} b_{i}, \sum_{i} \lambda_{i} e_{i} \right) : \lambda_{i} \in \mathbb{C} \right\}$$

of  $\widehat{Z}$ . Define Z to be the quotient of  $\widehat{Z}$  by N. Let  $\phi$  be the composition of the canonical inclusion of X into  $\widehat{Z}$  with the quotient map from  $\widehat{Z}$  to Z. Similarly define  $\psi: Y \to Z$ .

By the properties of 1-sums and quotients,  $\phi$  and  $\psi$  are *n*-contractions. We claim that they are in fact *n*-isometries. We will only show that  $\phi$  is an *n*-isometry, since the proof for  $\psi$  is entirely analogous. Suppose that  $x \in M_n(X)$ . Pick an *n*-contraction  $u: X \to M_n$  such that

$$||x|| = ||(id_{M_n} \otimes u)(x)||.$$

By hypothesis there are *n*-contractions  $v : Y \to M_n$  and  $w : \ell^1(k) \to M_n$  such that  $w(e_i) = u(e_i) - v(e_i)$ . Therefore if  $\alpha_i \in M_n$  then the norm of

$$\left(x - \sum_{i} \alpha_i \otimes a_i, \sum_{i} \alpha_i \otimes b_i, \sum_{i} \alpha_i \otimes e_i\right)$$

in  $M_n(\widehat{Z})$  is bounded from below by the norm of

$$(id_{M_n} \otimes u) \left( x - \sum_i \alpha_i \otimes a_i \right) \\ + (id_{M_n} \otimes v) \left( \sum_i \alpha_i \otimes b_i \right) \\ + (id_{M_n} \otimes w) \left( \sum_i \alpha_i \otimes e_i \right) \\ = (id_{M_n} \otimes u) (x)$$

which equals ||x||. Since this is true for every  $\alpha_i \in M_n$ ,  $\phi$  is an *n*-isometry. Similarly  $\psi$  is an *n*-isometry. The proof is concluded by observing that  $||\phi(a_i) - \psi(b_i)|| \le 1$  for every  $i \le k$ .

**Corollary 3.2.3.** If  $\bar{a}, \bar{b} \in \mathcal{M}_n(k)$  and  $d_{\mathcal{M}_n}(\bar{a}, \bar{b}) \leq M$  then for every  $\alpha_i \in M$ 

$$\left\| \left\| \sum_{i} \alpha_{i} \otimes a_{i} \right\| - \left\| \sum_{i} \alpha_{i} \otimes b_{i} \right\| \right\| \leq M \left\| \sum_{i} \alpha_{i} \otimes e_{i} \right\|$$

where  $\bar{e}$  is the canonical basis of  $\ell^1(k)$ .

An Auerbach system in a Banach space is a basis  $\bar{a}$  with dual basis  $\bar{a}'$  such that  $||a_i|| = ||a_i'|| = 1$ . By analogy we say that an element  $\bar{a}$  of  $\mathcal{M}_n(k)$  is *N*-Auerbach if  $||a_i|| \leq N$  and  $||a_i'|| \leq N$  for every  $i \leq k$ . Denote by  $\mathcal{M}_n(k, N)$  the set of *N*-Auerbach  $\bar{a} \in \mathcal{M}_n(k)$ . It follows from Corollary 3.2.3 that the set  $\mathcal{M}_n(k, N)$  is closed in  $\mathcal{M}_n(k)$ . It can be easily

verified that if  $\bar{a} \in \mathcal{M}_n(k, N)$  and  $\alpha_i \in M_n$  then

$$\left\|\sum_{i} \alpha_{i} \otimes a_{i}\right\| \leq N \left\|\sum_{i} \alpha_{i} \otimes e_{i}\right\|$$

and

$$\left\|\sum_{i} \alpha_{i} \otimes e_{i}\right\| \leq kN \left\|\sum_{i} \alpha_{i} \otimes e_{i}\right\|$$

where  $\bar{e}$  is the canonical basis of  $\ell^1(k)$ .

If  $\bar{a}, \bar{b} \in \mathcal{M}_n(k)$ , denote by  $\iota_{\bar{a},\bar{b}}$  the linear isomorphism from  $\langle \bar{a} \rangle$  to  $\langle \bar{b} \rangle$  such that  $\iota_{\bar{a},\bar{b}}(a_i) = b_i$  for  $i \leq k$ . Define the *n*-bounded distance  $d_{nb}(\bar{a},\bar{b})$  to be  $||\iota_{\bar{a},\bar{b}}||_n ||\iota_{\bar{a},\bar{b}}^{-1}||_n$ . (Observe that this is not an actual metric, but  $\log(d_{nb})$  is a metric.)

In the following lemma we establish a precise relation between the *n*-bounded distance  $d_{nb}$  and the Fraïssé metric  $d_{nb}$  on  $\mathcal{M}_n(k, N)$ .

**Proposition 3.2.4.** Suppose that  $\bar{a}, \bar{b} \in \mathcal{M}_n(k, N)$ . Then

$$d_{nb}(\bar{a},\bar{b}) \le \left(1 + kNd_{\mathcal{M}_n}(\bar{a},\bar{b})\right)^2$$

and

$$d_{\mathcal{M}_n}(\bar{a}, b) \le d_{nb}(\bar{a}, b) - 1.$$

*Proof.* Suppose that  $d_{\mathcal{M}_n}(\bar{a}, \bar{b}) \leq M$ . Then by Corollary 3.2.3

$$\left\| \left\| \sum_{i} \alpha_{i} \otimes a_{i} \right\| - \left\| \sum_{i} \alpha_{i} \otimes b_{i} \right\| \right\| \leq M \left\| \sum_{i} \alpha_{i} \otimes e_{i} \right\|$$

for every  $\alpha_i \in M_n$ , where  $\bar{e}$  is the canonical basis of  $\ell^1(k)$ . Since  $\bar{a}, \bar{b}$  are N-Auerbach we have

$$\left\|\sum_{i} \alpha_{i} \otimes a_{i}\right\| \leq \left\|\sum_{i} \alpha_{i} \otimes b_{i}\right\| + M\left\|\sum_{i} \alpha_{i} \otimes e_{i}\right\| \leq (1 + kNM)\left\|\sum_{i} \alpha_{i} \otimes b_{i}\right\|$$

and similarly

$$\left\|\sum_{i} \alpha_{i} \otimes b_{i}\right\| \leq (1 + kNM) \left\|\sum_{i} \alpha_{i} \otimes b_{i}\right\|.$$

Therefore

$$d_{nb}(\bar{a},\bar{b}) \le (1+kNM)^2 \,.$$

The other inequality is an immediate consequence of Lemma 3.2.1.

We can finally show that the space  $(\mathcal{M}_n(k), d_{\mathcal{M}_n})$  is separable and complete. In view of Proposition 3.2.4 this can be proved by a standard argument; see for example [115, Theorem 21.1 and Remark 21.2]. A proof is included for the sake of completeness.

**Proposition 3.2.5.** The space  $(\mathcal{M}_n(k), d_{\mathcal{M}_n})$  is compact.

*Proof.* Suppose that  $(\bar{a}^{(m)})_{m\in\mathbb{N}}$  is a sequence in  $\mathcal{M}_n(k)$ . If  $\alpha_i \in M_n$  then

$$\left(\left\|\sum_{i}\alpha_{i}\otimes a_{i}^{(m)}\right\|\right)_{m\in\mathbb{N}}$$

is a bounded sequence of complex numbers. Therefore after passing to a subsequence we can assume that such a sequence converges for any choice of  $\alpha_i \in M_n(\mathbb{Q}(i))$ . This is easily seen to imply that in fact such a sequence convergence for any choice of  $\alpha_i \in M_n$ . Moreover the functions

$$(\alpha_1,\ldots,\alpha_k)\mapsto \left\|\sum_i \alpha_i\otimes a_i^{(m)}\right\|$$

are equiuniformly continuous on the unit ball of  $M_n$ . Therefore, by the Ascoli-Arzelá theorem, after passing to a further subsequence we can assume that the convergence is uniform on the unit ball of  $M_n$ . We can now define an element  $\bar{a}$  of  $\mathcal{M}_n(k)$  by setting

$$\left\|\sum_{i} \alpha_{i} \otimes a_{i}\right\| = \lim_{m \to +\infty} \left\|\sum_{i} \alpha_{i} \otimes a_{i}^{(m)}\right\|.$$

The abstract characterization of  $M_n$ -spaces shows that  $\bar{a}$  is indeed an element of  $\mathcal{M}_n(k)$ . By uniform convergence in the unit ball the sequence  $(\bar{a}^{(m)})_{m\in\mathbb{N}}$  is such that  $d_{nb}(\bar{a}^{(m)},\bar{a}) \to 1$ . Therefore  $d_{\mathcal{M}_n}(\bar{a}^{(m)},\bar{a}) \to 0$  by Proposition 3.2.4.

This concludes the proof that  $\mathcal{M}_n$  is a complete Fraissé class.

### 3.2.3 The Fraïssé limit

We have verified that the class  $\mathcal{M}_n$  is a Fraïssé class in the sense of Definition 3.1.5. Therefore by Theorem 3.1.9 we can consider its Fraïssé limit. Observe that the  $\mathcal{M}_n$ -structures are precisely the  $\mathcal{M}_n$ -spaces. We first provide a characterization for the Fraïssé limit of  $\mathcal{M}_n$ similar in spirit to the universal property defining the Gurarij Banach space. **Proposition 3.2.6.** Suppose that Z is a separable  $M_n$ -space. The following statements are equivalent:

- 1. Z is the Fraissé limit of the class  $\mathcal{M}_n$ ;
- 2. If  $X \subset Y$  are finite-dimensional  $M_n$ -spaces,  $\psi : X \to Z$  is a linear n-isometry, and  $\varepsilon > 0$ , then there is a linear function  $\phi : Y \to Z$  extending  $\phi$  such that  $||\phi||_n ||\phi^{-1}||_n < 1 + \varepsilon$ .

*Proof.* The proof is entirely analogous to the proof of [8, Theorem 3.3], and is presented here for convenience of the reader.

- (1) $\Rightarrow$ (2) Suppose that Z is the Fraïssé limit of the class  $\mathcal{M}_n$ . Suppose that  $X \subset Y$  are finite-dimensional  $\mathcal{M}_n$ -spaces,  $\phi: X \to Z$  is a linear *n*-isometry, and  $\varepsilon > 0$ . Fix  $\delta > 0$  small enough. Consider also a basis  $(a_1, \ldots, a_k)$  of X and a basis  $(b_1, \ldots, b_m)$  of Y such that  $b_i = a_i$  for  $i \leq k$ . Since Z is by assumption the Fraïssé limit of the class  $\mathcal{M}_n$ , there is a linear *n*-isometry  $\hat{\phi}: Y \to Z$  such that  $||\phi(a_i) \hat{\phi}(a_i)|| \leq \delta$  for every  $i \leq k$ . Define now  $\psi: Y \to Z$  by setting  $\psi(b_i) = \phi(a_i)$  for  $i \leq k$  and  $\psi(b_i) = \hat{\phi}(a_i)$  for  $k < i \leq m$ . A routine calculation shows that, for  $\delta$  small enough,  $\psi$  satisfies the desired inequality.
- (2) $\Rightarrow$ (1) Suppose now that Z satisfies condition (2). Consider  $X \in \mathcal{M}_n(k)$ , a finite *l*-tuple  $\bar{b}$  in X,  $\psi \in \operatorname{Stx}_{\mathcal{M}_n}(X, Z)$ , and  $\varepsilon > 0$ . By [8, Lemma 2.16] in order to show that Z is the Fraise limit of  $\mathcal{M}_n$  it is enough to find  $\varphi \in \operatorname{Stx}_{\mathcal{M}_n}^{<\psi}(X, Z)$  with the following property: for every  $i \leq l$  there is  $y \in Z$  such that  $\varphi(b_i, y) < \varepsilon$ . By [8, Lemma 2.8(iii)] after enlarging X, and decreasing  $\varepsilon$  we can assume that there is a finite *m*-tuple  $\bar{c}$  in X and an *n*-isometric linear map  $f : \langle \bar{c} \rangle \to Z$  such that  $\psi \geq f_{|\bar{c}} + \varepsilon$ . (Recall our convention of identifying partial isomorphisms between  $\mathcal{L}$ -structures with the corresponding approximate isomorphisms.) Denote by  $\bar{b}\bar{c}$  the concatenation of the tuples  $\bar{b}$  and  $\bar{c}$ . By assumption if  $\delta > 0$ , then we can extend f to a linear map  $f : X \to Z$  satisfying  $||f||_n ||f^{-1}||_n < 1 + \delta$ . In view of Proposition 3.2.4 by choosing  $\delta$  small enough one can ensure that

$$d_{\mathcal{M}_n}\left(\bar{b}\bar{c}, f\left(\bar{b}\bar{c}\right)\right) < \varepsilon$$

Therefore by the definition of the Fraïssé metric  $d_{\mathcal{M}_n}$  (Definition 3.1.4) there is

$$\varphi \in \operatorname{Stx}_{\mathcal{M}_n}\left(\left\langle \bar{b}\bar{c}\right\rangle, \left\langle f\left(\bar{b}\bar{c}\right)\right\rangle\right) \subset \operatorname{Stx}_{\mathcal{M}_n}(X, Z)$$

such that  $\varphi(b_i, f(b_i)) < \varepsilon$  for  $i \leq l$  and  $\varphi(c_j, f(c_j)) < \varepsilon$  for  $j \leq m$ . Observe that such a  $\varphi$  satisfies

$$\psi \ge f_{|\bar{c}} + \varepsilon > \varphi_{|\bar{c} \times f(\bar{c})} \ge \varphi.$$

This concludes the proof.

In view of Proposition 3.2.6 the following theorem is an immediate consequence of 3.1.9 and the fact that  $\mathcal{M}_n$  is a complete Fraissé class.

**Theorem 3.2.7.** There is a separable  $M_n$ -space  $\mathbb{G}_n$  with the following property: If  $X \subset Y$  are finite-dimensional  $M_n$ -spaces,  $\psi : X \to \mathbb{G}_n$  is a linear n-isometry, and  $\varepsilon > 0$ , then there is a linear function  $\phi : Y \to \mathbb{G}_n$  extending  $\phi$  such that  $||\phi||_n ||\phi^{-1}||_n < 1 + \varepsilon$ . Any two separable  $M_n$ -spaces with such a property are n-isometrically isomorphic. Moreover  $\mathbb{G}_n$  contains any separable  $M_n$ -space as a subspace, and has the following homogeneity property: If  $X \subset \mathbb{G}_n$  is finite-dimensional,  $\phi : X \to \mathbb{G}_n$  is a linear n-isometry, and  $\varepsilon > 0$ , then there is a surjective n-isometry  $\psi : \mathbb{G}_n \to \mathbb{G}_n$  such that  $||\psi|_X - \phi||_n < \varepsilon$ .

Clearly for n = 1 one obtains a Banach space which is isometrically isomorphic to the Gurarij space. The same proof as [82, Theorem 1.1] where one replaces [82, Lemma 2.1] with Lemma 3.2.1 yields the following homogeneity result for  $\mathbb{G}_n$ .

**Theorem 3.2.8.** Suppose that  $X \subset \mathbb{G}_n$  is a finite-dimensional subspace and  $\phi : X \to \mathbb{G}_n$ is an invertible linear map such that  $\|\phi\| < 1 + \varepsilon$  and  $\|\phi^{-1}\| < 1 + \varepsilon$ . Then there exists a surjective linear n-isometry  $\psi : \mathbb{G}_n \to \mathbb{G}_n$  such that  $\|\psi|_X - \phi\|_n < \varepsilon$ .

# 3.3 The noncommutative Gurarij space

### **3.3.1** MIN and MAX spaces

Clearly any operator space can be canonically regarded as an  $M_n$ -space. Conversely if X is an  $M_n$ -space, then there are two canonical ways to regard X as an operator space. It is natural to call an operator space structure  $\hat{X}$  on X compatible if the map  $X \mapsto \hat{X}$  is an n-isometry. The minimal and maximal compatible operator space structures  $MIN_n(X)$  and  $MAX_n(X)$  on an  $M_n$ -space are defined by the formulas

$$\left\|x\right\|_{M_{k}(\mathrm{MIN}_{n}(X))} = \sup_{\phi} \left\|\left(id_{M_{k}}\otimes\phi\right)(x)\right\|_{M_{k}\otimes M_{n}}$$

where  $\phi$  varies among all *n*-contractions from X to  $M_n$ , and

$$\|x\|_{M_{k}(\mathrm{MAX}_{n}(X))} = \sup_{u} \|(id_{M_{k}} \otimes u)(x)\|_{M_{k}(B(H))}$$

where u varies among all n-contractions from X to B(H). These are introduced in [86, Section I.3] as a generalization of the minimal and maximal quantization of a Banach space as in [26, Section 3.3]; see also [105, Section 2]. If X is an operator space then we define  $MIN_n(X)$  and  $MAX_n(X)$  to be the structures defined above starting from X regarded just as  $M_n$ -space. This is consistent with the terminology used in [105, 104].

The names MIN and MAX are suggestive of the following property; see [86, Proposition I.3.1]. If  $\hat{X}$  is a compatible operator space structure on X then the identity maps

$$MAX_n(X) \to \hat{X} \to MIN_n(X)$$

are completely bounded. The operator space structures MIN and MAX are characterized by the following universal property; see [86, Proposition I.3.6 and Proposition I.3.7]. If Z is an operator space and  $u: Z \to X$  is a linear map, then  $u: Z \to X$  is n-bounded if and only if  $u: Z \to MIN_n(X)$  is completely bounded, and in such case

$$||u: Z \to \operatorname{MIN}_n(X)||_{cb} = ||u: Z \to X||_n$$

Similarly if Z is an operator space and  $u: X \to Z$  is a linear map, then  $u: X \to Z$  is *n*-bounded if and only if  $u: MAX_n(X) \to Z$  is completely bounded, and in such case

$$\|u: \operatorname{MAX}(X) \to Z\|_{cb} = \|u: X \to Z\|_n$$
.

**Remark 3.3.1.** In the following we will always consider an  $M_n$ -space X as an operator system endowed with its minimal compatible operator system structure.

It is worth noting at this point that all the proofs of Section 3.2 go through without change when  $M_n$ -spaces are regarded as operator spaces with their minimal compatible operator space structure. This easily follows from the properties of the minimal quantization recalled above.

### **3.3.2** Exact and 1-exact operator spaces

Suppose that E and F are two finite-dimensional operator spaces. Define  $d_{cb}(E, F)$  to be the infimum of  $||\phi||_{cb} ||\phi^{-1}||_{cb}$  when  $\phi$  ranges over all linear isomorphisms from E to F. The exactness constant ex(E) of a finite-dimensional operator space is the infimum of  $d_{cb}(E, F)$ where F ranges among all subspaces of  $M_n$  for  $n \in \mathbb{N}$ . Equivalently one can define ex(E)to be the limit for  $n \to +\infty$  of the decreasing sequence

$$\|id_E: \operatorname{MIN}_n(E) \to E\|_{cb}$$
,

where  $id_E$  denotes the identity map of E. If X is a not necessarily finite-dimensional operator space, then its exactness constant ex(X) is the supremum of ex(E) where E ranges over all finite-dimensional subspaces of E.

An operator space is *exact* if it has finite exactness constant, and 1-*exact* if it has exactness constant 1. For C\*-algebras exactness is equivalent to 1-exactness, which is in turn equivalent to several other properties; see [11, Section IV.3.4]. Exactness is a fundamental notion in the theory of C\*-algebras and operator spaces. It is a purely noncommutative phenomenon: there is no Banach space analog of nonexactness. In fact every Banach space—and in fact every  $M_n$ -space—is 1-exact. More information and several equivalent characterizations of exactness can be found in [114] and [115, Chapter 17].

In the following we will denote by  $\mathcal{E}_1$  the class of finite-dimensional 1-exact operator spaces. Moreover we will denote by  $\mathcal{M}^0_{\infty} \subset \mathcal{E}_1$  the class of operator spaces that admit a completely isometric embedding into  $M_n$  for some  $n \in \mathbb{N}$ . Our goal is to show that  $\mathcal{E}_1$  is a Fraïssé class.

### 3.3.3 Amalgamation of 1-exact operator spaces

It is clear that  $\mathcal{E}_1$  has (HP) from Definition 3.1.2. It remains to verify that  $\mathcal{E}_1$  satisfies (AP). This will give (JEP) as consequence, since the trivial operator space  $\{0\}$  embeds in every element of  $\mathcal{E}_1$ .

We recall that if  $(Z_n)$  is a direct sequence of operator spaces with completely isometric linear maps  $\phi_n : Z_n \to Z_{n+1}$  one can define the *direct limit*  $\lim_{(\phi_n)} Z_n$  with canonical completely isometric linear maps  $\sigma_k : Z_k \to \lim_{(\phi_n)} Z_n$  in the following way. Let W = $\ell^{\infty}(\mathbb{N}, (Z_n))$  be the space of sequences  $(z_n) \in \prod_n Z_n$  with  $\sup_n ||z_n|| < +\infty$ . Define an operator seminorm structure on  $\widehat{W}$  in the sense of [12, 1.2.16] by setting

$$\rho_k\left((z_n)_{n\in\mathbb{N}}\right) = \limsup_{n\to+\infty} \left\|z_n\right\|_{M_k(Z_n)}$$

for  $k \in \mathbb{N}$  and  $z_n \in M_k(Z_n)$ . Finally define W to be the operator space associated with such an operator seminorm structure on  $\widehat{W}$ . For n, m let  $\phi_{n,n} = id_{Z_n}, \phi_{n,m} = \phi_{m-1} \circ \cdots \circ \phi_n$  if n < m, and  $\phi_{n,m} = 0$  otherwise. Define the maps  $\sigma_k : Z_k \to W$  by

$$\begin{array}{rcl} Z_k & \to & W \\ \\ x & \mapsto & \left(\phi_{k,n}(x)\right)_{n \in \mathbb{N}} \end{array}$$

Finally set  $\lim_{(\phi_n)} Z_n$  to be the closure inside W of the union of the images of  $Z_k$  under  $\sigma_k$  for  $k \in \mathbb{N}$ . It is clear that if for every  $k \in \mathbb{N}$  the space  $Z_k$  is 1-exact, then  $\lim_{(\phi_n)} Z_n$  is 1-exact.

The proof of the following proposition is inspired by [24, Theorem 4.7] and [104, Theorem 1.1].

**Proposition 3.3.2.** Suppose that  $X_0 \subset X$  and Y are finite-dimensional 1-exact operator spaces,  $\delta > 0$ , and  $f: X_0 \to Y$  is such that  $||f||_{cb} < 1 + \delta$  and  $||f^{-1}||_{cb} < 1 + \delta$ . Then there exists a 1-exact separable operator space Z and linear complete isometries  $j: Y \to Z$  and  $i: X \to Z$  such that  $||j \circ f - i|_{X_0}||_{cb} < \delta$ .

*Proof.* Fix  $\delta' < \delta$  such that  $||f||_{cb} < 1 + \delta'$  and  $||f^{-1}||_{cb} < 1 + \delta'$  and  $\varepsilon > 0$  such that  $\delta' + 4\varepsilon < \delta$ . We will construct by recursion on k sequences  $(n_k)_{k \in \mathbb{N}}, (Z_k)_{k \in \mathbb{N}}, i_k : X \to Z_k, j_k : Y \to Z_k, \phi_k : Z_k \to Z_{k+1}$  such that

- 1.  $(n_k)_{k \in \mathbb{N}}$  is nondecreasing,
- 2.  $Z_k$  is an  $M_{n_k}$ -space,
- 3.  $i_k$  and  $j_k$  are injective completely contractive linear maps,
- 4.  $\phi_k$  is a completely isometric linear map,
- 5.  $\|i_k^{-1}\|_{cb} \leq 1 + \varepsilon 2^{-k}, \|j_k^{-1}\|_{cb} \leq 1 + \varepsilon 2^{-k},$ 6.  $\|\phi_k \circ i_k - i_{k+1}\|_{cb} \leq 1 + \varepsilon 2^{-k}, \|\phi_k \circ j_k - j_{k+1}\|_{cb} \leq 1 + \varepsilon 2^{-k}, \text{ and}$ 7.  $\|j_k \circ f - (i_k)|_{X_0}\|_{cb} \leq \delta' + (2\varepsilon) \sum_{i < k} 2^{-i}.$

We can apply Lemma 3.2.1 and Lemma 3.1.11 to define  $n_1, Z_1, i_1$ , and  $j_1$ . Suppose that  $n_k, Z_k, i_k, j_k$ , and  $\phi_{k-1}$  have been defined for  $k \leq m$ . By Lemma 3.1.11 we can pick  $n_{m+1} \geq n_m$  and injective completely contractive maps  $\theta_X : X \to M_{n_{m+1}}$  and  $\theta_Y : Y \to M_{n_{m+1}}$  such that  $\|\theta_X^{-1}\|_{cb} \leq 1 + \varepsilon 2^{-2(m+1)}$  and  $\|\theta_Y^{-1}\|_{cb} \leq 1 + \varepsilon 2^{-(m+1)}$ . By injectivity of  $M_{n_{m+1}}$  there are complete contractions  $\alpha_X, \alpha_Y : Z_m \to M_{n_{m+1}}$  such that

$$\alpha_X \circ i_m = \frac{1}{1 + \varepsilon 2^{-m}} \theta_X$$
 and  $\alpha_Y \circ j_m = \frac{1}{1 + \varepsilon 2^{-m}} \theta_Y.$ 

Define W to be  $MIN_{n_{m+1}}(Z_m \oplus^{\infty} M_{n_{m+1}})$ . Define linear maps

and

$$\begin{aligned} \widehat{\alpha}_X : & Z_m & \to & W \\ & z & \mapsto & (z, \alpha_X(z)) \,, \end{aligned}$$
$$\widehat{\alpha}_Y : & Z_m & \to & W \\ & z & \mapsto & (z, \alpha_Y(z)) \,. \end{aligned}$$

Observe that  $\hat{\alpha}_X, \hat{\alpha}_Y$  are completely isometric, while  $\hat{\theta}_X$  and  $\hat{\theta}_Y$  are completely contractive with

$$\left\|\widehat{\theta}_X^{-1}\right\|_{cb} \le \left\|\theta_X^{-1}\right\|_{cb} \le 1 + \varepsilon 2^{-(m+1)}$$

and

$$\left\|\widehat{\theta}_Y^{-1}\right\|_{cb} \le \left\|\theta_Y^{-1}\right\|_{cb} \le 1 + \varepsilon 2^{-(m+1)}.$$

Note also that

$$\left\|\widehat{\theta}_X - \widehat{\alpha}_X \circ i_m\right\|_{cb} \le \varepsilon 2^{-m} \quad \text{and} \quad \left\|\widehat{\theta}_Y - \widehat{\alpha}_Y \circ i_m\right\|_{cb} \le \varepsilon 2^{-m}.$$

Define now

$$N = \{ (-(z_0 + z_1), \widehat{\alpha}_X(z_0), \widehat{\alpha}_Y(z_1)) \in Z_m \oplus W \oplus W : z_0, z_1 \in Z_m \}.$$

Let  $Z_{m+1}$  be

$$\operatorname{MIN}_{n_{m+1}}((Z_m \oplus^1 W \oplus^1 W)/N))$$

Consider the first coordinate inclusion  $\phi_m : Z_m \to Z_{m+1}$  of  $Z_m$  into  $Z_{m+1}$ . Similarly define  $\psi_X, \psi_Y : W \to Z_{m+1}$  to be the second and third coordinate inclusions. Arguing as in the proof of Lemma 3.2.1 one can verify directly that  $\phi_m, \psi_X, \psi_Y$  are complete isometries. Alternatively one can use [104, Lemma 2.4] together with the properties of MIN. Observe

that  $\widehat{\alpha}_X \circ \phi_m = \psi_X$  and  $\widehat{\alpha}_Y \circ \phi_m = \psi_Y$ . Define now linear complete contractions

$$i_{m+1} := \psi_X \circ \widehat{\theta}_X : X \to Z_{m+1}$$
 and  $j_{m+1} := \psi_Y \circ \widehat{\theta}_Y : Y \to Z_{m+1}$ .

Observe that

$$\left\| i_{m+1}^{-1} \right\|_{cb} \le \left\| \widehat{\theta}_X^{-1} \right\|_{cb} < 1 + \varepsilon 2^{-(m+1)}$$

and

$$\|j_{m+1}^{-1}\|_{cb} \le \|\widehat{\theta}_Y^{-1}\|_{cb} < 1 + \varepsilon 2^{-(m+1)}.$$

Moreover

$$\begin{aligned} \|\phi_m \circ i_m - i_{m+1}\|_{cb} &= \left\|\phi_m \circ i_m - \psi_X \circ \widehat{\theta}_X\right\|_{cb} \\ &\leq \left\|\phi_m \circ i_m - \psi_X \circ \widehat{\alpha}_X \circ i_m\right\|_{cb} + \varepsilon 2^{-m} \\ &= \varepsilon 2^{-m}. \end{aligned}$$

Similarly

$$\|\psi_m \circ j_m - j_{m+1}\|_{cb} \le \varepsilon 2^{-m}.$$

Finally we have

$$\begin{aligned} \|i_{m+1} - j_{m+1} \circ f\|_{cb} &= \left\| \psi_X \circ \widehat{\theta}_X - \psi_Y \circ \widehat{\theta}_Y \circ f \right\|_{cb} \\ &\leq \left\| \psi_X \circ \widehat{\alpha}_X \circ i_m - \psi_Y \circ \widehat{\alpha}_Y \circ j_m \circ f \right\|_{cb} + (2\varepsilon) 2^{-m} \\ &\leq \left\| \phi_m \circ i_m - \phi_m \circ j_m \circ f \right\| + (2\varepsilon) 2^{-m} \\ &\leq \left\| i_m - j_m \circ f \right\| + (2\varepsilon) 2^{-m} \\ &\leq \delta' + (2\varepsilon) \sum_{i \leq m} 2^{-i}. \end{aligned}$$

This concludes the recursive construction. Let now Z be  $\lim_{(\phi_k)} Z_k$  with canonical linear complete isometries  $\sigma_k : Z_k \to Z$ . Consider also the embeddings  $i : X \to Z$  and  $j : Y \to Z$  defined by

$$i := \lim_{k \to +\infty} \sigma_k \circ i_k$$
 and  $j := \lim_{k \to +\infty} \sigma_k \circ j_k$ .

It is easily seen as in the proof of [24, Theorem 4.7] that Z is a 1-exact separable operator space, and i, j are well defined completely isometric linear maps such that  $\|j \circ f - i|_{X_0}\|_{cb} \leq \delta + 2\varepsilon$ .

In particular Proposition 3.3.2 for  $\delta = 0$  shows that the class  $\mathcal{E}_1$  has (NAP). It is not

difficult to modify the proof above to show that the conclusions of Proposition 3.3.2 hold even when  $X_0 \subset X$  and Y are not necessarily finite-dimensional separable 1-exact operator spaces. Moreover one can obtain Z to be an  $\mathcal{OL}_{\infty,1+}$  space in the sense of [25, 65].

### 3.3.4 The Fraïssé metric space

Fix  $k \in \mathbb{N}$  and denote by  $\mathcal{E}_1(k)$  the space of pairs  $(\bar{a}, X)$  such that X is a k-dimensional 1-exact operator space and  $\bar{a}$  is a basis of X. Two such pairs  $(\bar{a}, X)$  and  $(\bar{b}, Y)$  are identified if there is a complete isometry  $\phi$  from X to Y such that  $\phi(\bar{a}) = \bar{b}$ . To simplify the notation the pair  $(\bar{a}, X)$  will be simply denoted  $\bar{a}$ , where we set  $X = \langle \bar{a} \rangle$ . Denote by  $d_{\mathcal{E}_1}$  the Fraïssé metric on  $\mathcal{E}_1(k)$  as in Definition 3.1.4. We further denote by  $\mathcal{M}^0_{\infty}(k)$  the subspace of  $\mathcal{E}_1(k)$ consisting of pairs  $(\bar{a}, X)$  such that X admits a completely isometric embedding into  $M_n$ for some  $n \in \mathbb{N}$ . Let  $\ell^1(k)$  be the k-fold 1-sum of  $\mathbb{C}$  by itself in the category of operator spaces. A similar proof as the one of Proposition 3.2.2 gives the following:

**Proposition 3.3.3.** Suppose that  $\bar{a}, \bar{b} \in \mathcal{E}_1(k)$  and M > 0. If  $d_{\mathcal{E}_1}(\bar{a}, \bar{b}) \leq M$  then for every  $n \in \mathbb{N}$  and every completely contractive  $u : X \to M_n$  there is a completely contractive  $v : Y \to M_n$  such that the linear function  $w : \ell^1(k) \to B(H)$  defined by  $w(e_i) = u(a_i) - v(b_i)$  has completely bounded norm at most M, and vice versa.

**Corollary 3.3.4.** Suppose that  $\bar{a}, \bar{b} \in \mathcal{E}_1(k)$  and M > 0. If  $d_{\mathcal{E}_1}(\bar{a}, \bar{b}) \leq M$  then for every  $n \in \mathbb{N}$  and  $\alpha_i \in M_n$ 

$$\left| \left\| \sum_{i} \alpha_{i} \otimes a_{i} \right\| - \left\| \sum_{i} \alpha_{i} \otimes b_{i} \right\| \right| \leq M \left\| \sum_{i} \alpha_{i} \otimes e_{i} \right\|$$

where  $\bar{e}$  denotes the canonical basis of  $\ell^1(k)$ .

As in Subsection 3.2.2 we define an element  $\bar{a}$  of  $\mathcal{E}_1$  to be *N*-Auerbach if  $||a_i|| \leq N$  and  $||a'_i|| \leq N$  for every  $i \leq k$ , where  $\bar{a}'$  denotes the dual basis of  $\bar{a}$ . We denote by  $\mathcal{E}_1(k, N)$  the set of *N*-Auerbach  $\bar{a} \in \mathcal{E}_1(k)$ . Observe that in view of Corollary 3.2.3 the set  $\mathcal{E}_1(k, N)$  is closed in  $\mathcal{E}_1(k)$ . Moreover it can be easily verified that if  $\bar{a} \in \mathcal{E}_1(k, N)$  and  $\alpha_i \in M_n$  then

$$\left\|\sum_{i} \alpha_{i} \otimes a_{i}\right\| \leq N \left\|\sum_{i} \alpha_{i} \otimes e_{i}\right\|$$

and

$$\left\|\sum_{i} \alpha_{i} \otimes e_{i}\right\| \leq kN \left\|\sum_{i} \alpha_{i} \otimes e_{i}\right\|.$$

If  $\bar{a}, \bar{b} \in \mathcal{E}_1(k)$ , denote as in Subsection 3.2.2 by  $\iota_{\bar{a},\bar{b}}$  the linear isomorphism from  $\langle \bar{a} \rangle$  to  $\langle \bar{b} \rangle$  such that  $\iota_{\bar{a},\bar{b}}(a_i) = b_i$  for  $i \leq k$ . Define the completely bounded distance  $d_{cb}(\bar{a},\bar{b})$  to be  $||\iota_{\bar{a},\bar{b}}||_{cb} ||\iota_{\bar{a},\bar{b}}^{-1}||_{cb}$ .

**Proposition 3.3.5.** Suppose that  $\bar{a}, \bar{b} \in \mathcal{E}_1(k, N)$ . Then

$$d_{cb}(\bar{a}, \bar{b}) \le \left(1 + kNd_{\mathcal{E}_1}(\bar{a}, \bar{b})\right)^2$$

and

$$d_{\mathcal{E}_1}(\bar{a}, \bar{b}) \le d_{cb}(\bar{a}, \bar{b}) - 1.$$

*Proof.* The first inequality can be inferred from Proposition 3.3.3; see also the proof of the first inequality in Proposition 3.2.4. The second inequality is an immediate consequence of Proposition 3.3.2.

Using Proposition 3.3.5 one can show that  $(\mathcal{E}_1(k), d_{\mathcal{E}_1})$  is a separable metric space. The proof is similar to [114, Proposition 12]. Recall that  $\mathcal{M}^0_{\infty} \subset \mathcal{E}_1$  denotes the class of operator spaces that admit a completely isometric embedding into  $M_n$  for some  $n \in \mathbb{N}$ .

**Proposition 3.3.6.** For every  $k \in \mathbb{N}$ ,  $(\mathcal{E}_1(k), d_{\mathcal{E}_1})$  is a complete metric space, and  $\mathcal{M}^0_{\infty}(k)$  is a dense subset of  $\mathcal{E}_1(k)$ .

*Proof.* Suppose that  $(\bar{a}^{(m)})_{m \in \mathbb{N}}$  is a Cauchy sequence in  $\mathcal{E}_1(k)$ . By Proposition 3.3.8 there is  $N \in \mathbb{N}$  such that, for every  $m \in \mathbb{N}$ ,  $\bar{a}^{(m)} \in \mathcal{E}_1(k, N)$ . Moreover

$$\limsup_{k,m \to +\infty} d_{cb}(\bar{a}^{(m)}, \bar{a}^{(k)}) = 1$$

Fix a nonprincipal ultrafilter  $\mathcal{U}$  over  $\mathbb{N}$ . Define X to be the ultraproduct  $\prod_{\mathcal{U}} \langle \bar{a}^{(m)} \rangle$  as in [26, Section 10.3]. Let  $a_i$  be the element of X having  $(a_i^{(m)})_{m \in \mathbb{N}}$  as representative sequence. Observe that for every  $n, m \in \mathbb{N}$  and  $\alpha_i \in M_n$ ,

$$\left\| \left\| \sum_{i} \alpha_{i} \otimes a_{i}^{(m)} \right\| - \left\| \sum_{i} \alpha_{i} \otimes a_{i} \right\| \right\|$$
$$= \lim_{k \to \mathcal{U}} \left\| \left\| \sum_{i} \alpha_{i} \otimes a_{i}^{(m)} \right\| - \left\| \sum_{i} \alpha_{i} \otimes a_{i}^{(k)} \right\|$$
$$\leq \sup_{k \ge m} d_{cb}(\bar{a}^{(m)}, \bar{a}^{(k)}).$$

Therefore  $d_{cb}(\bar{a}^{(m)}, \bar{a}) \to 1$ . Hence again by Proposition 3.3.8 we have that  $d_{\mathcal{E}_1}(\bar{a}^{(m)}, \bar{a}) \to 0$ . This shows that  $\bar{a}$  is the limit of the Cauchy sequence  $(\bar{a}^{(m)})_{m \in \mathbb{N}}$ . Observe now that by Proposition 3.3.8 and by the definition of 1-exact operator space,  $\mathcal{M}^0_{\infty}(k)$  is dense in  $(\mathcal{E}_1(k), d_{\mathcal{E}_1})$ . It follows from Proposition 3.2.5 that  $\mathcal{M}^0_{\infty}(k)$  is a separable subspace of  $\mathcal{E}_1(k)$ . Alternatively one can observe that the algebra  $\mathcal{K}$  of compact operators contains all elements of  $\mathcal{M}^0_{\infty}(k)$ , and then apply Corollary [115, Corollary 2.13.3] and Proposition 3.3.5. In any case one can conclude that  $\mathcal{E}_1(k)$  is separable as well.

### 3.3.5 The noncommutative Gurarij space as a Fraïssé limit

We have shown in Subsection 3.3.3 that  $\mathcal{E}_1$  is a Fraïssé class. Therefore we can consider its corresponding Fraïssé limit according to Theorem 3.1.9. First we observe that using Proposition 3.3.3 one can reformulate the property of being limit.

The following notion has been introduced by Oikhberg in [104].

**Definition 3.3.7.** An operator space Z is *noncommutative Gurarij* if for any completely isometric embedding  $\phi : X \to Z$  of a 1-exact operator space  $X \subset Y$ , and for any  $\varepsilon > 0$ , there is an injective linear map  $\psi : Y \to Z$  extending  $\phi$  such that  $||\psi||_{cb} ||\psi^{-1}||_{cb} < 1 + \varepsilon$ .

The following proposition characterizes noncommutative Gurarij spaces as limits of the Fraïssé class  $\mathcal{E}_1$ .

**Proposition 3.3.8.** Suppose that Z is a separable 1-exact. The following conditions are equivalent:

- 1. Z is a Fraïssé limit of the class  $\mathcal{E}_1$ ;
- 2. For every  $n \in \mathbb{N}$ , subspace X of  $M_n$ , complete isometry  $\phi : X \to Z$ , and  $\varepsilon > 0$  there is a complete isometry  $\psi : M_n \to Z$  such that  $\|\psi|_X \phi\| < \varepsilon$ ;
- 3. Z is noncommutative Gurarij;
- 4. Z satisfies the universal property of Definition 3.3.7 when  $Y = M_n$  for some  $n \in \mathbb{N}$ ;
- 5. For every  $n, k, l \in \mathbb{N}$ , for every subspace X of  $M_n$ , finite l-tuple  $\bar{b}$  in X,  $\psi \in \operatorname{Stx}_{\mathcal{E}_1}(X, Z)$ , and  $\varepsilon > 0$ , there is  $\varphi \in \operatorname{Stx}_{\mathcal{E}_1}^{<\psi}(X, Z)$  such that for every  $i \leq l$  there is  $y \in Z$  such that  $\varphi(b_i, y) < \varepsilon$ .

Proof. The implications  $(1) \Rightarrow (2)$  and  $(3) \Rightarrow (4)$  are obvious. The implications  $(2) \Rightarrow (4)$  and  $(1) \Rightarrow (3)$  can be proved using Proposition 3.3.5 similarly as the implication  $(1) \Rightarrow (2)$  in Proposition 3.2.6. The implication  $(4) \Rightarrow (5)$  can be proved as  $(2) \Rightarrow (1)$  of Proposition 3.2.6, or [8, Theorem 3.3]. Finally  $(5) \Rightarrow (1)$  is a consequence of [8, Lemma 2.16] and the fact that  $\mathcal{M}^0_{\infty}(k)$  is dense in  $(\mathcal{E}_1(k), d_{\mathcal{E}_1})$ ; see Proposition 3.3.6.

With such a characterization of a limit of the Fraissé class  $\mathcal{E}_1$  at hand, we can finally state the main result of this chapter, which is an immediate consequence of Theorem 3.1.9 and Proposition 3.3.8.

**Theorem 3.3.9.** There is a separable 1-exact operator space  $\mathbb{NG}$  which is noncommutative Gurarij. Such a space is unique up to completely isometric isomorphism. Every separable 1-exact operator space can be completely isometrically embedded into  $\mathbb{NG}$ .

Theorem 1.1 of [104] shows that any two noncommutative Gurarij spaces are *approximately* completely isometric. The uniqueness assertion in Theorem 3.3.9 improves such a result, showing that any two noncommutative Gurarij spaces are (exactly) completely isometric. Assuming uniqueness, one can also deduce universality from [104, Theorem 1.1] together with the fact that every separable 1-exact operator space embeds into a separable  $\mathcal{OL}_{\infty,1+}$  space; see [24, Theorem 4.7].

Recall that an operator space X is an  $\mathcal{OL}_{\infty,1+}$  space as defined in [25] if for every finitedimensional subspace E of X and every  $\varepsilon > 0$  there is a finite-dimensional C\*-algebra A and a subspace F of A such that  $d_{cb}(E, F) < 1 + \varepsilon$ . This notion provides the noncommutative analog of  $\mathcal{L}_{\infty,1+}$  spaces as in [87]. Clearly  $\mathcal{OL}_{\infty,1+}$  spaces are closed under direct limits. Therefore from Remark 3.1.10 and Proposition 3.3.8 one can deduce the following fact, already observed by Oikhberg in [104].

### **Proposition 3.3.10.** The noncommutative Gurarij space is an $\mathcal{OL}_{\infty,1+}$ space.

The following homogeneity property of NG follows from the homogeneity statement in Theorem 3.1.9 and Proposition 3.3.8: If  $X \subset \mathbb{NG}$  is finite-dimensional,  $\phi : X \to \mathbb{NG}$  is a complete isometry, and  $\varepsilon > 0$ , then there is a completely isometric surjection  $\psi : \mathbb{NG} \to \mathbb{NG}$ such that  $\|\psi_{|X} - \phi\| < \varepsilon$ . We now observe that one can get  $\psi$  to be close to  $\phi$  is cb-norm. The following lemma can be easily obtained from Proposition 3.3.2, as Lemma 2.2 is derived from Lemma 2.1 in [83].

**Lemma 3.3.11.** If  $X \subset \mathbb{NG}$  is finite-dimensional, Y is finite-dimensional and 1-exact, and  $f: X \to Y$  is an invertible linear map such that  $\|f\|_{cb} < 1 + \delta$  and  $\|f^{-1}\|_{cb} < 1 + \delta$  then for every  $\varepsilon > 0$  there exists  $g: Y \to \mathbb{NG}$  such that  $\|g\|_{cb} \|g^{-1}\|_{cb} < 1 + \varepsilon$  and  $\|g \circ f - id_X\|_{cb} < \delta$ .

One can then run the same argument as the proof of Theorem 1.1 in [83], where [83, Lemma 2.2] is replaced by Lemma 3.3.11, to show that  $\mathbb{NG}$  has the following homogeneity property.

**Theorem 3.3.12.** If  $X \subset \mathbb{NG}$  is a finite-dimensional subspace and  $\phi : X \to \mathbb{NG}$  is an invertible linear map such that  $\|\phi\|_{cb} < 1 + \delta$  and  $\|\phi^{-1}\|_{cb} < 1 + \delta$  then there exists a surjective complete isometry  $\psi : \mathbb{NG} \to \mathbb{NG}$  such that  $\|\psi\|_X - \phi\|_{cb} < \delta$ .

# Chapter 4

# The Gurarij operator system

A unital operator space is a closed subspace X of B(H) containing the identity operator. Unital operator spaces can also be defined abstractly as operator spaces with a distinguished unitary element; see [13, Theorem 2.1]. Particularly important among unital operator spaces are operator systems. These are the unital operator spaces  $X \subset B(H)$  that are closed by taking adjoints. Again these can be abstractly characterized as those unital operator spaces that are spanned by their hermitian elements; see [13, Proposition 3.2]. An operator system X inherits from B(H) a notion of positivity for self-adjoint elements of  $M_n(X)$ . Operator systems can be equivalently characterized in terms of the \*-vector space structure together with the unit and the matricial positive cones; see [19, Section 2] and [107, Chapter 13]. A linear map between operator systems is *positive* if it maps positive elements to positive elements, and *completely positive* if all its amplifications are positive. A unital linear map between operator systems is completely positive if and only if it is completely contractive [12, 1.3.3], and in such a case it is automatically self-adjoint. A surjective unital complete isometry between operator systems is called a *complete order isomorphism*.

To every unital operator space  $X \subset B(H)$  one can canonically assign the operator system  $X + X^* = \text{span} \{x, x^* : x \in X\}$ . Such an operator system does not depend (up to complete order isomorphism) from the unital completely isometric realization of X as a subspace of B(H); see [12, 1.3.7]. Moreover any unital completely contractive (respectively completely isometric) linear map  $\phi : X \to Y$  between unital operator systems has a unique extension to a map  $\tilde{\phi} : X + X^* \to Y + Y^*$  with the same properties. Therefore in some sense there is no real loss of generality in only considering operator systems rather than arbitrary unital operator spaces.

In this chapter we consider the natural operator system analog the notion of operator

space of universal disposition. This is obtained by replacing 1-exact operator spaces with 1-exact operator systems, and considering *unital* linear maps instead of arbitrary linear maps. (An operator system is 1-exact if it is 1-exact as an operator space; see [70, Section 5] for equivalent characterizations.) Therefore we say that a separable 1-exact operator system  $\mathbb{GS}$  to is of almost universal disposition if whenever  $E \subset F$  are finite-dimensional 1-exact operator systems,  $\phi: E \to \mathbb{GS}$  is a unital complete isometry, and  $\varepsilon > 0$ , there is an extension  $\hat{\phi}: F \to \mathbb{GS}$  of  $\phi$  such that  $\|\hat{\phi}\|_{cb} \|\hat{\phi}^{-1}\|_{cb} < 1 + \varepsilon$ . In this chapter we prove that the class  $\mathcal{E}_1^{sy}$  of finite-dimensional 1-exact operator systems

In this chapter we prove that the class  $\mathcal{E}_{1}^{sy}$  of finite-dimensional 1-exact operator systems is a Fraïssé class. Moreover an operator system is of almost universal disposition if and only if it is a limit of  $\mathcal{E}_{1}^{sy}$ . As a consequence we conclude that there exist a unique (up to complete order isomorphism) operator system GS of almost universal disposition, which we call the Gurarij operator system. Furthermore any separable 1-exact operator system admits a unital completely isometric embedding into GS. The homogeneity property of GS asserts that for any unital complete isometry  $\phi : E \to F$  between finite-dimensional subspaces of GS and any  $\varepsilon > 0$  there is a complete order automorphism  $\alpha$  of GS such that  $\|\alpha_{|E} - \phi\|_{cb} < 1 + \varepsilon$ . The Gurarij operator system is nuclear in the sense of [57, Theorem 3.1], and in fact it is an inductive limit of full matrix algebras with unital completely isometric connecting maps. In particular GS is a C\*-system in the sense of [79], i.e. the second dual GS\*\* of GS is a C\*-algebra and the canonical embedding of GS into GS\*\* is unital and completely isometric. Finally show that GS is a universal C\*-system as defined in [79, Section 3]. This means that the canonical \*-homomorphism from the universal C\*-algebra  $C_u^*$  (GS) to the C\*-envelope  $C_e^*$  (GS) is a \*-isomorphism.

The rest of the chapter is organized as follows. In Section 4.1 we recall some facts about Fraïssé limits for metric structures and about operator systems. Section 4.2 contains the proof that finite-dimensional 1-exact operator systems form a Fraïssé class. Finally Section 4.3 contains the proof of the main result, charactering the Gurarij operator systems GS as the Fraïssé limit of finite-dimensional 1-exact operator systems.

### 4.1 **Preliminary notions**

### 4.1.1 Operator systems

An operator system is a unital operator space X such that there exists a complete isometry  $\phi: X \to B(H)$  mapping the distinguished unitary to the identity operator and mapping X onto a self-adjoint subspace of B(H). Operator systems are precisely the unital operator space that are spanned by their hermitian elements; see [13, Definition 3.1 and Proposition

3.2]. This provides an intrinsic characterization of operator systems among unital operator spaces.

A unital complete isometry  $\phi: X \to B(H)$  induces an involution  $x \mapsto x^*$  on X, as well as a notion of positivity in  $M_n(X)$  for every  $n \in \mathbb{N}$ . We will regard operator systems as structures in the language  $\mathcal{L}_{OSy}$  of unital operator spaces with a unitary function symbol for the involution. The intrinsic characterization [13, Proposition 3.2] shows that an  $\mathcal{L}_{OSy}$ structure is an operator system if and only if all its finitely generated substructures are operator systems.

An earlier and more commonly used abstract description of operator spaces is due to Choi and Effros [19]. Such a description only involves the unit, the involution, and the positive cones in all the matrix amplifications. The operator space structure is completely recovered by the relation

$$\|x\|_{M_n(X)} \le 1$$
 iff  $\begin{bmatrix} I_n & x \\ x^* & I_n \end{bmatrix} \in M_{2n}(X)$  is positive.

Suppose that X and Y are operator systems. A map  $\phi: X \to Y$  is unital provided that it maps the unit to the unit, and self-adjoint if  $\phi(x^*) = \phi(x)^*$  for every  $x \in X$ . The positive cones in X and Y also define a notion of positivity for maps. Thus  $\phi: X \to Y$  is positive if it maps positive elements to positive elements, and completely positive if its amplifications  $\phi^{(n)}$  are positive for every  $n \in \mathbb{N}$ . Every positive map is automatically self-adjoint. It is well known that a unital linear map between operator systems is completely positive if and only if it is completely contractive; see [12, 1.3.3]. In the following we will abbreviate "unital completely positive" with ucp, as it is customary. We will often tacitly use Arveson's extension theorem, asserting that B(H) is an injective element in the category of operator systems and with ucp maps as morphisms; see [107, Theorem 7.5].

**Definition 4.1.1.** An operator space is *matricial* if it admits a completely isometric embedding into  $M_n$  for some  $n \in \mathbb{N}$ . An operator system is matricial if it admits a complete order embedding into  $M_n$  for some  $n \in \mathbb{N}$ .

The theory of injective envelopes of operator spaces has been developed independently by Hamana [55, 54] and Ruan [124]. It follows from the main theorem of [130] that an operator space is matricial if and only if it its injective envelope is finite-dimensional. Moreover by [12, Corollary 4.2.8] an operator system is matricial if and only if it is matricial as an operator space.

### 4.1.2 Exactness

If  $X \subset B(H)$  and  $Y \subset B(H)$  are operator spaces, then their minimal tensor product  $X \otimes_{\min} Y$  is the completion of the algebraic tensor product with respect to the norm induced by the inclusion  $X \otimes Y \subset B(H) \otimes B(H) \subset B(H \otimes H)$  where  $H \otimes H$  denotes the Hilbertian tensor product. The minimal tensor product is canonically endowed with an operator space structure induced by the inclusion  $X \otimes_{\min} Y \subset B(H \otimes H)$ .

An operator space X 1-exact whenever for every C\*-algebra B and ideal I of B the sequence

$$0 \to I \otimes_{\min} X \to B \otimes_{\min} X \to B/I \otimes_{\min} X \to 0$$

is exact, and moreover the induced isomorphisms

$$\frac{B \otimes_{\min} X}{I \otimes_{\min} X} \to B/I \otimes_{\min} X$$

is a complete isometry.

If E and F are operator spaces, the *completely bounded distance*  $d_{cb}(E, F)$  of E and F is the infimum of  $\|\phi\|_{cb} \|\phi^{-1}\|_{cb}$  when  $\phi$  ranges among all isomorphisms from E to F. Theorem 1 of [114] provides the following equivalent reformulation of 1-exactness in terms of the completely bounded distance. An operator space X is 1-exact if and only if for every finite dimensional subspace E of X and every  $\varepsilon > 0$  there is  $n \in \mathbb{N}$  and a subspace F of  $M_n$  such that  $d_{cb}(E, F) \leq 1 + \varepsilon$ .

Suppose now that X, Y are operator systems, B is a *unital* C\*-algebra, and I is an ideal of B. The minimal tensor product  $X \otimes_{\min} Y$  is an operator system obtained as before as a completion of the algebraic tensor product via the inclusion  $X \otimes Y \subset B(H \otimes H)$ . It is shown in [70] that the quotient

$$\frac{B \otimes_{\min} X}{I \otimes_{\min} X}$$

is canonically endowed with an operator system structure. An operator system X is 1-exact [70, Definition 5.4] if for every unital C\*-algebra B and every ideal I of B the sequence

$$0 \to I \otimes_{\min} X \to B \otimes_{\min} X \to B/I \otimes_{\min} X \to 0$$

is exact, and moreover the induced (unital) isomorphism

$$\frac{B \otimes_{\min} X}{I \otimes_{\min} X} \to B/I \otimes_{\min} X$$

is a complete isometry. It is then shown in [70, Proposition 5.5] that an operator system X is 1-exact if and only if it is 1-exact as an operator space. Moreover the same proof as [114, Theorem 1] shows that if X is a 1-exact operator system then for every finite subset E of X and for every  $\varepsilon > 0$  there exist  $n \in \mathbb{N}$ , a subspace F of  $M_n$ , and a ucp invertible map  $\phi : E \to F$  such that  $\|\phi^{-1}\|_{cb} < 1 + \varepsilon$ . Summarizing we can list the following equivalent characterizations of 1-exactness for operator spaces.

**Proposition 4.1.2** (Kavruk-Paulsen-Todorov-Tomforde, Pisier). Suppose that E is a finitedimensional operator system. The following statements are equivalent:

- 1. E is 1-exact;
- 2. for every  $\varepsilon > 0$  there is a matricial operator space F such that  $d_{cb}(E, F) < 1 + \varepsilon$ ;
- 3. for every  $\varepsilon > 0$  there is  $n \in \mathbb{N}$  and an injective ucp map  $\phi : E \to M_n$  such that  $\|\phi^{-1}\|_{cb} < 1 + \varepsilon;$

An operator system X is 1-exact if and only if all its finite-dimensional subspaces are 1-exact.

The implications  $3 \Rightarrow 2$  in Proposition 4.1.2 is obvious. The implication  $2 \Rightarrow 1$  follows from [114, Theorem 1] and [70, Proposition 5.5]. Finally the implication  $1 \Rightarrow 3$  follows from the proof of [114, Theorem 1].

### 4.1.3 Pushouts of operator systems

A \*-vector space is a complex vector space V endowed with a conjugate linear involutive map  $v \mapsto v^*$  from V to V. A unital \*-vector space is a \*-vector space V endowed with a distinguished element  $1_V$  such that  $(1_V)^* = 1_V$ . Clearly any operator system is in particular a unital \*-vector space. A map  $\phi : V \to B(H)$  is unital and self-adjoint if  $\phi(1_V)$  is the identity operator and  $\phi(v^*) = \phi(v)^*$  for every  $v \in V$ .

Suppose that  $\mathcal{P}$  is a collection of unital self-adjoint maps  $\phi : V \to B(H_{\phi})$  such that  $\sup_{\phi} \|\phi(x)\| < +\infty$  for every  $x \in V$ . Let  $J_{\mathcal{P}}$  be the subspace

$$\bigcap_{\phi \in \mathcal{P}} \operatorname{Ker}\left(\phi\right)$$

of V. Define on  $V/J_{\mathcal{P}}$  the norm

$$\|x + J_{\mathcal{P}}\| = \sup_{\phi \in \mathcal{P}} \|\phi(x)\|_{B(H_{\phi})}.$$

Then the completion of  $V_{\mathcal{P}}$  of  $V/J_{\mathcal{P}}$  is an operator system with unit  $1_V + J_{\mathcal{P}}$  and matrix norms

$$\left\| \left[ x_{ij} + J_{\mathcal{P}} \right] \right\| = \sup_{\phi \in \mathcal{P}} \left\| \phi \left( x_{ij} \right) \right\|_{B\left( H_{\phi} \right)}.$$

Equivalently  $V_{\mathcal{P}}$  can be defined as the closure inside  $\bigoplus_{\phi} B(H_{\phi}) \subset B(\bigoplus_{\phi} H_{\phi})$  of the image of V under the map

$$v \mapsto (\phi(v))_{\phi \in \mathcal{P}}$$

By definition any element  $\phi$  of  $\mathcal{P}$  induces a ucp map from  $V_{\mathcal{P}}$  to  $B(H_{\phi})$ .

We use this construction to define *pushouts* in the category of operator systems with ucp maps. Suppose that Z, X, Y are operator systems and  $\alpha_X : Z \to X$  and  $\alpha_Y : Z \to Y$ are ucp maps. Let V be the quotient of the algebraic direct sum  $X \oplus Y$  by the subspace  $N = \text{span} \{(-1_X, 1_V)\}$ . Consider the collection  $\mathcal{P}$  of unital self-adjoint maps  $\phi : V \to B(H)$ of the form

$$(x, y) + N \mapsto \phi_X(x) + \phi_Y(y)$$

where  $\phi_X$  and  $\phi_Y$  are ucp maps on X and Y such that  $\phi_X \circ \alpha_X = \phi_Y \circ \alpha_Y$ . Let W be the operator system associated with the collection  $\mathcal{P}$  as in the paragraph above. The canonical morphisms  $\psi_X : X \to W$  and  $\psi_Y : Y \to W$  are obtained from the first and second coordinate inclusions of X and Y into  $(X \oplus Y) / N$ . When moreover the maps  $\alpha_X : Z \to X$ and  $\alpha_Y : Z \to Y$  are unital complete isometries then  $\psi_X : X \to W$  and  $\psi_Y : Y \to W$  are unital complete isometries. It follows easily from the definition that this construction is indeed a pushout in the category of operator systems with ucp maps as morphisms.

As shown in [52, Appendix B] operator systems can be regarded as structures in a suitable language in the logic for metric structures as defined in [35, Section 2]; see also [9]. In particular if  $(X_n)_{n \in \mathbb{N}}$  is a sequence of operator systems and  $\mathcal{U}$  is an ultrafilter on  $\mathbb{N}$ , then one can define the ultraproduct  $\prod_{\mathcal{U}} X_n$  of the  $X_n$ 's in such a language with respect to the ultrafilter  $\mathcal{U}$ . The ultraproduct can also be explicitly constructed as follows. Define

$$\ell^{\infty}(X_n) := \left\{ (x_n) \in \prod_n X_n : \sup_n \|x_n\| < +\infty \right\}.$$

Such a space has a natural unital \*-vector space structure with unit  $(1_{X_n})_{n\in\mathbb{N}}$  and involution  $(x_n)^* = (x_n^*)$ . Let then  $\mathcal{P}_{\mathcal{U}}$  be the collection of ucp maps of  $\ell^{\infty}(X_n)$  of the form

$$\begin{array}{rcl} \ell^{\infty}\left(X_{n}\right) & \rightarrow & \prod_{\mathcal{U}} B\left(H_{\phi_{n}}\right) \\ (x_{n}) & \mapsto & \left[\phi_{n}\left(x_{n}\right)\right], \end{array}$$

where  $\phi_n : X_n \to B(H_{\phi_n})$  is a ucp map and  $\prod_{\mathcal{U}} B(H_{\phi_n})$  denote the C\*-algebraic ultraproduct; see [11, II.8.1.7]. The ultraproduct  $\prod_{\mathcal{U}} X_n$  is the operator system obtained from the collection ucp maps  $\mathcal{P}_{\mathcal{U}}$  on the unital \*-vector space  $\ell^{\infty}(X_n)$  as at the beginning of this subsection.

We recall that the category of operator systems also admits inductive limits, which can be defined as in [77, Section 2]; see also the proof of Proposition 16 in [79].

### 4.1.4 $M_n$ -systems

Here we recall the definition of  $M_n$ -spaces and the functor  $MIN_n$  defined and studied in [86]; see also [105, Section 2] and [104, Subsection 2.1]. Let X be an operator space and  $n \in \mathbb{N}$ . The operator space  $MIN_n(X)$  has the same underlying vector space as X with matrix norms

$$\|x\|_{M_{k}(\mathrm{MIN}_{n}(X))} = \sup_{\phi} \left\|\phi^{(k)}(x)\right\|_{M_{k}(M_{n})}$$

where  $\phi$  ranges over all completely contractive maps from X to  $M_n$ . This defines an operator space structure on X such that the identity map  $id_X : X \to \text{MIN}_n(X)$  is completely contractive. Such a space is characterized by the following property. If Z is an operator space and  $\phi : Z \to X$  is a linear map, then  $\phi : Z \to \text{MIN}_n(X)$  is completely bounded if and only if  $\phi^{(n)} : M_n(Z) \to M_n(X)$  is bounded, and in such a case

$$\left\|\phi: Z \to \mathrm{MIN}_{n}\left(X\right)\right\|_{cb} = \left\|\phi^{(n)}: M_{n}\left(Z\right) \to M_{n}\left(X\right)\right\|.$$

This is a consequence of Smith's lemma [129, Theorem 2.10].

An operator space X is called an  $M_n$ -space provided that  $id_X : MIN_n(X) \to X$  is a complete isometry. It follows easily from the above mentioned property that  $M_n$ -spaces form a full subcategory of the category of operator spaces. Moreover the inclusion functor implements an equivalence of categories with inverse the functor MIN<sub>n</sub>. Assume now that E is a finite-dimensional operator space. Then E is 1-exact if and only if for every  $\varepsilon > 0$ there is  $n \in \mathbb{N}$  such that

$$\|id_X: \mathrm{MIN}_n(E) \to E\|_{cb} \leq 1 + \varepsilon;$$

see [105, Lemma 2.2]. In particular all  $M_n$ -spaces are 1-exact.

The natural analogs of the above notions in the category of operator systems have been defined in [136]. Suppose that X is an operator system. Then  $OMIN_n(X)$  is the operator

system having the same unital \*-vector space structure as X and matrix norms

$$\|x\|_{M_{k}(\mathrm{OMIN}_{n}(X))} = \sup_{\phi} \left\|\phi^{(k)}(x)\right\|_{M_{k}(M_{n})}$$

where  $\phi$  ranges over all ucp maps from X to  $M_n$ . This is a particular case of the construction of operator systems presented in Subsection 4.1.3 where  $\mathcal{P}$  is the collection of ucp maps from X to  $M_n$ . It is shown in [136] that OMIN<sub>n</sub> has analogous properties as MIN<sub>n</sub> where one replaces operator spaces with operator systems and (completely) contractive maps with unital (completely) positive maps. In the proofs the use of Smith's lemma for operator spaces [26, Proposition 2.2.2] is replaced by [107, Theorem 6.1]. It follows from the proof of the abstract characterization of operator systems [107, Theorem 13.1] that X and OMIN<sub>n</sub> (X) have the same matricial positive cones up to the *n*-th amplification. Moreover since a unital 2-positive map is contractive, OMIN<sup>u</sup><sub>2n</sub> (X) and X are *n*-isometric.

It follows from Proposition 4.1.2 and the properties of  $\text{OMIN}_n$  that a finite-dimensional operator system E is 1-exact if and only if for every  $\varepsilon > 0$  there is  $n \in \mathbb{N}$  such that

$$\|id_E: \operatorname{OMIN}_n(E) \to E\|_{cb} \le 1 + \varepsilon.$$

**Definition 4.1.3.** An  $M_n$ -system is an operator system X such that  $id_X : OMIN_n(X) \to X$  is a complete isometry.

Equivalently X is an  $M_n$ -system if and only if it admits a complete order embedding into  $C(K, M_n)$  for some compact Hausdorff space K. As before  $M_n$ -systems form a full subcategory of the category of operator systems with ucp maps as morphisms, and the inclusion functor is an equivalence of categories with inverse OMIN<sub>n</sub>. This readily implies in view of Subsection 4.1.3 that the category of  $M_n$ -systems has pushouts. Moreover such a pushout can be obtained by applying OMIN<sub>n</sub> to the pushout in the category of operator systems. In the following we will tacitly use the fact that every  $M_n$ -system can be approximated by a matricial  $M_n$ -system. In fact if X is an  $M_n$ -system and  $\varepsilon > 0$  then there is a ucp map  $\phi: X \to M_n \oplus^{\infty} \cdots \oplus^{\infty} M_n$  such that  $\|\phi^{-1}\|_{cb} < 1+\varepsilon$ . It is a consequence of [107, Theorem 6.1] that a unital linear map between operator systems is completely contractive if and only if its *n*-th amplification is positive. Finally we observe that the ultraproduct  $\prod_{\mathcal{U}} C(K_i, M_n)$  can be identified with  $C(T, M_n)$  where T is the compact Hausdorff space  $\sum_{\mathcal{U}} K_i$  defined as in [6]. It follows that the class of  $M_n$ -systems is closed under ultraproducts.

## 4.2 Finite-dimensional 1-exact operator systems

We regard operator systems as structures in the language  $\mathcal{L}_{OSy}$  defined in Subsection 4.1.1. We denote by  $\mathcal{E}_1^{sy}$  the class of finite-dimensional 1-exact operator systems. It follows easily from the characterization of 1-exactness provided by Proposition 4.1.2 together with the abstract characterization of operator systems among unital operator spaces [13, Proposition 3.2] that the  $\mathcal{E}_1^{sy}$ -structures in the sense of Subsection 3.1.3 are precisely the 1-exact operator systems. We define also  $\mathcal{M}_{\infty}^0 \subset \mathcal{E}_1^{sy}$  to be the class of matricial operator systems in the sense of Definition 4.1.1. In this section we will show that  $\mathcal{E}_1^{sy}$  is a Fraïssé class, and  $\mathcal{M}_{\infty}^0$ is a dense subclass.

### 4.2.1 Amalgamation of 1-exact operator systems

In this subsection we will show that the class  $\mathcal{E}_1^{sy}$  of finite-dimensional 1-exact operator systems satisfies (NAP) from Definition 3.1.2. We will start by considering matricial operator systems. If X, Y are operator systems and  $f: X \to Y$  is a linear map, define  $f^*: X \to Y$  by setting

$$f^*\left(x\right) = f\left(x^*\right)^*.$$

Define then Re  $(f) = \frac{1}{2} (f + f^*)$  and Im  $(f) = \frac{1}{2i} (f - f^*)$ .

The following lemma is an approximate version of [107, Lemma 2.10].

**Lemma 4.2.1.** Suppose that X is an operator system,  $\delta \in (0, 1]$ , and  $\phi$  is a unital linear functional on X such that  $\|\phi\| \le 1 + \delta$  for  $\delta \in [0, 1]$ . Then  $|\text{Im}(\phi(x))| \le 2\sqrt{\delta} \|x\|$  whenever  $x \in X$  is self-adjoint.

Proof. Suppose that  $x \in X$  is self-adjoint and  $||x|| \leq 1$ . Denote by  $\sigma(x)$  the spectrum of x. If M > 0 then  $\sigma(x)$  is contained in the disc of center iM and radius  $(M^2 + 1)^{\frac{1}{2}}$ . Therefore  $\sigma(x - iM)$  is contained in the disc of center 0 and radius  $(M^2 + 1)^{\frac{1}{2}}$ . Thus

$$\|\phi(x) - iM\| \le (1+\delta) \|x - iM\| \le (1+\delta) (M^2 + 1)^{\frac{1}{2}}.$$

Therefore

Im 
$$(\phi(x)) \ge M - (1+\delta) (M^2 + 1)^{\frac{1}{2}}$$
.

Observe that, for  $M \ge 0$ ,

$$(1+\delta) (M^{2}+1)^{\frac{1}{2}} - x = (1+\delta) M \left(1 + \frac{1}{M^{2}}\right)^{\frac{1}{2}} - M$$
$$\leq (1+\delta) M \left(1 + \frac{1}{2M^{2}}\right) - M$$
$$\leq \frac{1}{M^{2}} + \delta M.$$

Setting  $M = \frac{1}{\sqrt{\delta}}$ , this yields  $\operatorname{Im}(\phi(x)) \geq -2\sqrt{\delta}$ . A similar argument shows that  $\operatorname{Im}(\phi(x)) \leq 2\sqrt{\delta}$ .

**Lemma 4.2.2.** Suppose that X, Y are operator systems,  $\delta \in (0, 1]$ , and  $f : X \to Y$  is a unital linear map such that  $\|f\|_{cb} \leq 1 + \delta$ . Then  $\|\operatorname{Im}(f)\|_{cb} \leq 4\sqrt{\delta}$ .

*Proof.* Suppose that  $n \in \mathbb{N}$  and  $\rho$  is a state of  $M_n(Y)$ . Applying Lemma 4.2.1 to  $\phi = \rho \circ f^{(n)}$  one obtains that, whenever  $x \in M_n(X)$  is self-adjoint,

$$\left| \left( \rho \circ \operatorname{Im} \left( f \right)^{(n)} \right) (x) \right| = \left| \operatorname{Im} \left( \left( \rho \circ f^{(n)} \right) (x) \right) \right| \le 2\sqrt{\delta}.$$
  
Therefore  $\left\| \operatorname{Im} \left( f \right)^{(n)} \right\| \le 4\sqrt{\delta}$ , and  $\left\| \operatorname{Im} \left( f \right) \right\|_{cb} \le 4\sqrt{\delta}.$ 

We will need the following perturbation lemma, which is a minor variation of [23, Theorem 2.5].

**Lemma 4.2.3.** Suppose that E is a finite-dimensional operator system, Y is an operator system, and  $\delta \in [0,1]$ . Denote by dim (E) the dimension of E as a vector space. If  $f: E \to Y$  is unital linear map such that  $\|f\|_{cb} \leq 1 + \delta$ , then there is a ucp map  $\psi: E \to Y$  such that  $\|f - \psi\|_{cb} \leq 20 (\dim(E) + 1) \sqrt{\delta}$ .

Proof. For  $\delta = 0$  this follows from [107, Lemma 2.10]. Suppose now that  $\delta$  is nonzero. Consider Y as a subsystem of B(H), and let n be dim (E). By Lemma 4.2.2 we have that  $\|\operatorname{Im}(f)\|_{cb} \leq 4\sqrt{\delta}$ . By Wittstock's decomposition theorem [135]—see also [108, Corollary 2.6]—there are completely positive maps  $\phi_1, \phi_2 : E \to B(H)$  such that  $\operatorname{Re}(f) = \phi_1 - \phi_2$  and  $\|\phi_1 + \phi_2\|_{cb} \leq \|\operatorname{Re}(f)\|_{cb}$ . Let  $a_i = \phi_i(1)$  for i = 1, 2. By [107, Proposition 3.6] we have that

$$||a_1|| \le ||a_1 + a_2|| = ||\phi_1 + \phi_2||_{cb} \le ||\operatorname{Re}(f)||_{cb} \le 1 + \delta + 4\sqrt{\delta}.$$

If  $\xi \in H$  has norm 1, then

$$\langle a_2\xi,\xi\rangle = \langle (a_1-1)\xi,\xi\rangle \le \delta + 4\sqrt{\delta} \le 5\sqrt{\delta}.$$

Therefore  $\|\phi_2\|_{cb} = \|a_2\| \leq 5\sqrt{\delta}$ . By [23, Lemma 2.4] there is a linear functional  $\theta$  on *E*—which we will regard as a function from *E* to *Y*—such that  $\|\theta\| \leq 5n\sqrt{\delta}$  and  $\theta - \phi_2$  is completely positive. Define  $\psi' = \operatorname{Re}(f) + \theta = \phi_1 - \phi_2 + \theta$  and observe that  $\psi'$  is completely positive, being sum of completely positive maps. Moreover  $\|\psi' - f\|_{cb} \leq \|\operatorname{Im}(f)\|_{cb} + \|\theta\| \leq 4\sqrt{\delta} + 5n\sqrt{\delta}$  and  $\|\psi'\|_{cb} \leq 1 + 5(n+1)\sqrt{\delta}$ . Set  $\psi'' = \frac{1}{1+5(n+1)\sqrt{\delta}}\psi'$  and observe that  $\psi''$  is completely positive and completely contractive. Moreover

$$\|\psi''(1) - 1\| \le \|\psi''(1) - \psi'(1)\| + \|\psi'(1) - f(1)\| \le 10 (n+1) \sqrt{\delta}.$$

Let  $\rho$  be a state on E and set  $\psi(x) = \psi''(x) + \rho(x)(1 - \psi''(1))$ . Observe that  $\psi$  is completely positive being sum of completely positive maps. Moreover  $\psi$  is unital and

$$\begin{aligned} \|\psi - f\|_{cb} &\leq \|1 - \psi''(1)\| + \|\psi' - f\|_{cb} + \|\psi' - \psi''\|_{cb} \\ &\leq 20 (n+1) \sqrt{\delta}. \quad \Box \end{aligned}$$

**Lemma 4.2.4.** Suppose that  $E \subset F_0$  and  $F_1$  are matricial operator systems, and  $\delta \in [0, 1]$ . If  $f: E \to F_1$  is an invertible unital map such that  $||f||_{cb} \leq 1 + \delta$  and  $||f^{-1}||_{cb} \leq 1 + \delta$ , then there exist  $d \in \mathbb{N}$  and unital completely isometric embeddings  $i: F_0 \to M_d$  and  $j: F_1 \to M_d$ such that  $||j \circ f - i|_E||_{cb} \leq 100 \dim (E) \delta^{\frac{1}{2}}$ .

Proof. Without loss of generality we can assume that  $F_0 = M_n$  and  $F_1 = M_k$  for some  $n, k \in \mathbb{N}$ . By Lemma 4.2.3 and injectivity of  $M_k$  there is a ucp map  $\phi : M_n \to M_k$  such that  $\|\phi_{|E} - f\|_{cb} \leq 50 \dim(E) \delta^{\frac{1}{2}}$ . Similarly there is a ucp map  $\psi : M_k \to M_n$  such that  $\|\psi_{|f[E]} - f^{-1}\|_{cb} \leq 50 \dim(E) \delta^{\frac{1}{2}}$  and hence  $\|\psi \circ f - id_E\| \leq 100 \dim(E) \delta^{\frac{1}{2}}$ . Set d = n + k and define maps  $i : M_n \to M_d$  and  $j : M_k \to M_d$  by

$$i(x) = \begin{bmatrix} x & 0\\ 0 & \phi(x) \end{bmatrix}$$
 and  $j(y) = \begin{bmatrix} \psi(y) & 0\\ 0 & y \end{bmatrix}$ 

for  $x \in M_n$  and  $y \in M_k$ . Observe that i and j are unital complete isometries such that

$$\|j \circ f - i_{|E}\|_{cb} = \max\left\{\|\psi \circ f - id_{E}\|_{cb}, \|f - \phi_{|E}\|_{cb}\right\} \le 100 \dim(E) \delta^{\frac{1}{2}}.$$

We are now ready to bootstrap Lemma 4.2.4 to arbitrary finite-dimensional 1-exact operator systems. The proof is analogous to the proof of [90, Proposition 4.2].

**Proposition 4.2.5.** Suppose that  $E \subset X$  and Y are finite-dimensional 1-exact operator systems, and  $\delta \in [0,1]$ . If  $f : E \to Y$  is a unital map such that  $||f||_{cb} < 1 + \delta$  and  $||f^{-1}||_{cb} < 1+\delta$ , then there exist a separable 1-exact operator system Z and unital completely isometric embeddings  $i : X \to Z$  and  $j : Y \to Z$  such that  $||j \circ f - i||_{cb} < 100 \dim (E) \delta^{\frac{1}{2}}$ .

Proof. Fix  $\varepsilon, \delta' \in (0, 1)$  such that  $||f||_{cb} < 1 + \delta', ||f^{-1}||_{cb} < 1 + \delta'$ , and  $\delta' + 2\varepsilon < \delta$ . Set  $\eta = \left(\frac{\varepsilon}{100 \dim(E)}\right)^2$ . We will define by recursion on k sequences  $(n_k)_{k \in \mathbb{N}}, (Z_k)_{k \in \mathbb{N}}, (i_k)_{k \in \mathbb{N}}, (j_k)_{k \in \mathbb{N}}$ , and  $(\phi_k)_{k \in \mathbb{N}}$  such that, for every  $k \in \mathbb{N}$ ,

- 1.  $Z_k$  is an  $M_{n_k}$ -system,
- 2.  $i_k: X \to Z_k$  is a ucp map such that  $\|i_k^{-1}\|_{cb} \leq 1 + \eta 2^{-2k}$ ,
- 3.  $j_k: Y \to Z_k$  is a ucp map such that  $\|j_k^{-1}\|_{cb} \leq 1 + \eta 2^{-2k}$  and

$$\left\| j_k \circ f - (i_k)_{|E} \right\|_{cb} \le 100 \dim (E) \,\delta^{\frac{1}{2}} + 2\varepsilon \left( \sum_{i < k} 2^{-i} \right),$$

4.  $\phi_k : Z_k \to Z_{k+1}$  is a ucp map such that  $\|\phi_k^{-1}\| \le 1 + \eta 2^{-4k}$ ,

$$\|\phi_k \circ i_k - i_{k+1}\|_{cb} \le \varepsilon 2^{-k}$$
 and  $\|\phi_k \circ j_k - j_{k+1}\|_{cb} \le \varepsilon 2^{-k}$ .

Granted the construction we can let Z be the inductive limit  $\lim_{(\phi_k)} Z_k$  and define the maps

$$i = \lim_{k} i_k : X \to Z$$
 and  $j = \lim_{k} j_k : Y \to Z$ .

It is immediate to verify that conditions (1)–(4) above guarantee that these maps have the desired properties. We will now present the recursive construction. By Lemma 4.2.4 and Proposition 4.1.2 one can find  $n_1 \in \mathbb{N}$  and ucp maps  $i_1 : X \to M_{n_1}$  and  $j_1 : Y \to M_{n_1}$  that satisfy conditions (1)–(3) above for k = 1 and  $Z_1 = M_{n_1}$ . Suppose that  $n_k, Z_k, i_k, j_k$ , and  $\phi_{k-1}$  have been defined for  $k \leq m$ . By Proposition 4.1.2 there exist  $d \in \mathbb{N}$  and ucp maps  $\theta_X : X \to M_d$  and  $\theta_Y : Y \to M_d$  such that

$$\max\left\{\left\|\theta_X^{-1}\right\|_{cb}, \left\|\theta_Y^{-1}\right\|_{cb}\right\} \le 1 + \eta 2^{-2(m+1)}.$$

Moreover by Lemma 4.2.3 there are ucp maps  $\alpha_X : i_m[X] \subset Z_m \to M_d$  and  $\alpha_Y : i_m[Y] \subset Z_m \to M_d$  such that

$$\max\left\{\left\|\alpha_X - \theta_X \circ i_m^{-1}\right\|_{cb}, \left\|\alpha_Y - \theta_Y \circ j_m^{-1}\right\|_{cb}\right\} \le \varepsilon 2^{-m}.$$

Injectivity of  $M_d$  ensures that we can extend  $\alpha_X$  and  $\alpha_Y$  to ucp maps defined on all  $Z_m$ . Set now  $n_{m+1} = n_m + d$ . Define maps

$$\widehat{\theta}_X : X \to M_{n_{m+1}} \\ x \mapsto \begin{bmatrix} i_m(x) & 0 \\ 0 & \theta_X(x) \end{bmatrix},$$

$$\widehat{\theta}_Y : Y \to M_{n_{m+1}} \\ y \mapsto \begin{bmatrix} j_m(y) & 0 \\ 0 & \theta_Y(y) \end{bmatrix},$$

and

$$\widehat{\alpha}_X : Z_m \to M_{n_{m+1}}$$

$$z \mapsto \begin{bmatrix} z & 0 \\ 0 & \alpha_X(z) \end{bmatrix},$$

$$\widehat{\alpha}_Y : Z_m \to M_{n_{m+1}}$$

$$z \mapsto \begin{bmatrix} z & 0 \\ 0 & \alpha_Y(z) \end{bmatrix}.$$

Observe that  $\widehat{\alpha}_X$  and  $\widehat{\alpha}_Y$  are unital complete isometries, while  $\widehat{\theta}_X$  and  $\widehat{\theta}_Y$  are ucp maps such that

$$\max\left\{ \left\| \widehat{\theta}_{X}^{-1} \right\|_{cb}, \left\| \widehat{\theta}_{Y}^{-1} \right\|_{cb} \right\} \le \max\left\{ \left\| \theta_{X}^{-1} \right\|_{cb}, \left\| \theta_{Y}^{-1} \right\|_{cb} \right\} \le 1 + \eta 2^{-4(m+1)}$$

Moreover

$$\max\left\{\left\|\widehat{\theta}_{X} - \widehat{\alpha}_{X} \circ i_{m}\right\|_{cb}, \left\|\widehat{\theta}_{Y} - \widehat{\alpha}_{Y} \circ j_{m}\right\|_{cb}\right\} \\ \leq \max\left\{\left\|\alpha_{X} - \theta_{X} \circ i_{m}^{-1}\right\|_{cb}, \left\|\alpha_{Y} - \theta_{Y} \circ j_{m}^{-1}\right\|_{cb}\right\} \leq \varepsilon 2^{-m}.$$

Now let  $Z_{m+1}$  be the pushout of  $\widehat{\alpha}_X : Z_m \to M_{n_{m+1}}$  and  $\widehat{\alpha}_Y : Z_m \to M_{n_{m+1}}$  in the category of  $M_{n_{m+1}}$ -systems with canonical unital complete isometries  $\psi_X : M_{n_{m+1}} \to Z_{m+1}$  and  $\psi_Y :$ 

 $M_{n_{m+1}} \to Z_{m+1}$ ; see Subsection 4.1.3 and Subsection 4.1.4. Set  $\phi_m := \psi_X \circ \widehat{\alpha}_X = \psi_Y \circ \widehat{\alpha}_Y$ ,  $i_{m+1} := \psi_X \circ \widehat{\theta}_X$ , and  $j_{m+1} := \psi_Y \circ \widehat{\theta}_Y$ . Observe that

$$\max\left\{\left\|i_{m+1}^{-1}\right\|_{cb}, \left\|j_{m+1}^{-1}\right\|_{cb}\right\} \le \max\left\{\left\|\widehat{\theta}_{X}^{-1}\right\|_{cb}, \left\|\widehat{\theta}_{Y}^{-1}\right\|_{cb}\right\} \le 1 + \eta 2^{-4(m+1)}.$$

Moreover

$$\begin{aligned} \|\phi \circ i_m - i_{m+1}\|_{cb} &= \left\|\phi \circ i_m - \psi_X \circ \widehat{\theta}_X\right\|_{cb} \\ &\leq \left\|\phi \circ i_m - \psi_X \circ \widehat{\alpha}_X \circ i_m\right\|_{cb} + \varepsilon 2^{-m} \\ &\leq \varepsilon 2^{-m} \end{aligned}$$

and similarly  $\|\phi \circ j_m - j_{m+1}\|_{cb} \leq \varepsilon 2^{-m}$ . Furthermore

$$\begin{aligned} \left\| (i_{m+1})_{|E} - j_{m+1} \circ f \right\|_{cb} &= \left\| \left( \psi_X \circ \widehat{\theta}_X \right)_{|E} - \psi_Y \circ \widehat{\theta}_Y \circ f \right\|_{cb} \\ &\leq \left\| \psi_X \circ \widehat{\alpha}_X \circ i_m - \psi_Y \circ \widehat{\alpha}_Y \circ j_m \circ f \right\| + \varepsilon 2^{-m+1} \\ &\leq \left\| \phi_m \circ i_m - \phi_m \circ j_m \circ f \right\| + \varepsilon 2^{-m+1} \\ &\leq 100 \dim (E) \, \delta^{\frac{1}{2}} + 2\varepsilon \sum_{k \leq m} 2^{-k}. \end{aligned}$$

This concludes the recursive construction.

It is not difficult to modify the proof of Proposition 4.2.5 to cover the case when X and Y are not necessarily finite-dimensional.

### 4.2.2 Embeddings of 1-exact operator systems

The goal of this section is to show that every 1-exact separable operator system admits a unital completely isometric embedding into an inductive limit of full matrix algebras with unital completely isometric connective maps. The construction of the inductive limit of an inductive sequence of operator systems can be found in [79, Section 2]; see also the proof of Proposition 16 in [77]. The proof of the following proposition is similar to the one of [24, Theorem 4.7].

**Proposition 4.2.6.** Suppose that X is a separable 1-exact operator system. Then there are a sequence natural numbers  $n_k$  and unital complete isometries  $\phi_k : M_{n_k} \to M_{n_{k+1}}$  such that X embeds unitally completely isometrically into the inductive limit  $\lim_{\phi_k} M_{n_k}$ .

Proof. Let  $(E_k)_{k\in\mathbb{N}}$  be an increasing sequence of finite-dimensional subsystems of X with dense union. Set  $\varepsilon_k = (50 \dim (E_k))^{-2}$ , where  $\dim (E_k)$  denotes the dimension of  $E_k$  as a vector space. We will define the sequences  $(n_k)$  and  $(\phi_k)$  by recursion on  $k \in \mathbb{N}$  together with ucp maps  $i_k : E_k \to M_{n_k}$  such that  $\|i_k^{-1}\|_{cb} \leq 1 + \varepsilon_k 2^{-2k}$  and  $\|(i_{k+1})|_{E_k} - \phi_k \circ i_k\| \leq 2^{-k}$ . Granted this construction we let i be the unital complete isometry

$$\lim_{k \to +\infty} i_k : X \to \lim_{(\phi_k)} M_{n_k}.$$

The map  $i_1: E_1 \to M_{n_1}$  can be defined applying the characterization of 1-exactness provided by Proposition 4.1.2. Suppose that  $n_k$ ,  $i_k$ , and  $\phi_{k-1}$  satisfying the conditions above have been defined for  $k \leq m$ . Again by Proposition 4.1.2 one can find  $d \in \mathbb{N}$  together with a ucp map  $\theta: E_{m+1} \to M_d$  such that  $\|\theta^{-1}\|_{cb} \leq 1 + \varepsilon_{m+1}2^{-2(m+1)}$ . By Lemma 4.2.3 there is a ucp map  $\alpha: i_m [E_m] \subset M_{n_m} \to M_d$  such that  $\|\alpha - \theta \circ i_m^{-1}\|_{cb} \leq 2^{-m}$ . By injectivity of  $M_n$  we can extend  $\alpha$  to a ucp map from  $M_{n_m}$  to  $M_d$ . Define now  $n_{m+1} = n_m + d$ ,  $\phi_m: M_{n_m} \to M_{n_{m+1}}$  and  $i_{m+1}: E_{m+1} \to M_{n_{m+1}}$  by setting

$$\phi_m(z) = \begin{bmatrix} z & 0 \\ 0 & \alpha(z) \end{bmatrix} \text{ and } i_{m+1}(x) = \begin{bmatrix} i_m(x) & 0 \\ 0 & \theta(x) \end{bmatrix}$$

for  $z \in M_{n_m}$  and  $x \in E_{m+1}$ . Observe that  $\phi_m$  is a unital complete isometry, and  $i_{m+1}$  is a ucp map such that

$$\|i_{m+1}^{-1}\|_{cb} \le \|\theta^{-1}\|_{cb} \le 1 + \varepsilon_{m+1} 2^{-2(m+1)}.$$

Moreover

$$\begin{aligned} \|\phi_m \circ i_m - i_{m+1}\|_{cb} &= \|\theta - \alpha \circ i_m\|_{cb} \\ &\leq \|\alpha - \theta \circ i_m^{-1}\|_{cb} \\ &< 2^{-m}. \end{aligned}$$

This concludes the recursive construction

### 4.2.3 The Fraïssé metric space

Recall that we denote by  $\mathcal{E}_1^{sy}$  the class of finite-dimensional 1-exact operator systems. It is clear that  $\mathcal{E}_1^{sy}$  satisfies (HP) from Definition 3.1.2. Proposition 4.2.5 shows that  $\mathcal{E}_1^{sy}$  has (NAP). Finally (JEP) is a consequence of (NAP) and the observation that the operator system  $\mathbb{C}$  is an initial object in the category of operator systems. In order to conclude that
$\mathcal{E}_1^{sy}$  is a Fraïssé class it is enough to show that for every  $k \in \mathbb{N}$  the space  $\mathcal{E}_1^{sy}(k)$  endowed with the Fraïssé metric  $d_{\mathcal{E}_1^{sy}}$  from Definition 3.1.4 is complete and separable. Without loss of generality we can assume that  $\mathcal{E}_1^{sy}(k)$  only contains the pairs  $(\overline{a}, E)$  where E is a finitedimensional 1-exact operator system and  $\overline{a}$  is a linear basis of E. Any two such pairs  $(\overline{a}, E)$ and  $(\overline{b}, F)$  are identified if there is a complete order isomorphism from E to F mapping  $\overline{a}$ to  $\overline{b}$ . For brevity we denote an element  $(\overline{a}, E)$  of  $\mathcal{E}_1^{sy}(k)$  simply by  $\overline{a}$  and set  $E = \langle \overline{a} \rangle$ .

Recall that a basis  $\{x_1, \ldots, x_n\}$  of a Banach space X is a Auerbach provided that  $||x_i|| = ||x'_i|| = 1$  for  $i \leq n$ , where  $\{x'_1, \ldots, x'_n\}$  denotes the dual basis; see [14, Theorem 13]. In analogy we call an element  $\overline{a}$  of  $\mathcal{E}_1^{sy}$  with dual basis  $\overline{a}'$  N-Auerbach for  $N \in \mathbb{N}$  if  $||a_i|| \leq N$  and  $||a'_i|| \leq N$  for  $i \leq n$ . The following is a standard lemma, known as the small perturbation argument; see [17, Lemma 12.3.15].

**Lemma 4.2.7.** Suppose that  $\overline{a}, \overline{b} \in \mathcal{E}_1^{sy}(k)$  are *N*-Auerbach. If  $d_{\mathcal{E}_1^{sy}}(\overline{a}, \overline{b}) \leq \frac{1}{4kN}$ , then there is a unital isomorphism  $f : \langle \overline{a} \rangle \to \langle \overline{b} \rangle$  such that  $\|f\|_{cb} \leq 1 + kNd_{\mathcal{E}_1^{sy}}(\overline{a}, \overline{b})$  and  $\|f^{-1}\|_{cb} \leq 1 + kNd_{\mathcal{E}_1^{sy}}(\overline{a}, \overline{b})$ .

*Proof.* Set  $\delta = d_{\mathcal{E}_1^{sy}}(\overline{a}, \overline{b})$ . We can assume that  $\langle \overline{a} \rangle$  and  $\langle \overline{b} \rangle$  are unitally contained in a unital C\*-algebra A in such a way that  $\max_i ||a_i - b_i|| \leq \delta$ . Define the map

$$\begin{array}{rccc} \theta : & A & \to & A \\ & z & \mapsto & z + \sum\limits_{i \leq n} b'_i\left(z\right) \left(b_i - a_i\right). \end{array}$$

Observe that  $\theta(a_i) = b_i$  for  $i \leq k$  and

$$\|id_A - \theta\|_{cb} \le \sum_{i \le n} \|b'_i\| \le kN\delta.$$

Reversing the roles of  $\overline{a}$  and  $\overline{b}$  allows one to define a map  $\tilde{\theta} : A \to A$  such that  $\tilde{\theta}(b_i) = a_i$ and  $\left\| id_A - \tilde{\theta} \right\|_{cb} \leq kN\delta$ . Let now  $f : \langle \overline{a} \rangle \to \langle \overline{b} \rangle$  be  $\theta_{|\langle \overline{a} \rangle}$ , and observe that f satisfies the desired conclusions.

The following lemma is an immediate consequence of Proposition 4.2.5.

**Lemma 4.2.8.** Suppose that  $\delta \in [0, \frac{1}{20}]$  and  $\overline{a}, \overline{b} \in \mathcal{E}_1^{sy}(k)$ . If there is an isomorphism  $f : \langle \overline{a} \rangle \to \langle \overline{b} \rangle$  such that  $||f(1) - 1|| \leq \delta$ ,  $||f||_{cb} \leq 1 + \delta$  and  $||f^{-1}||_{cb} \leq 1 + \delta$ , then  $d_{\mathcal{E}_1^{sy}}(\overline{a}, \overline{b}) \leq 100k\delta^{\frac{1}{2}}$ .

*Proof.* Let  $\rho$  be a state on  $\langle \overline{a} \rangle$ . Define  $g(x) = f(x) + \rho(x)(1 - f(1))$ . Observe that  $\|g - f\|_{cb} \leq \delta$  and hence  $\|g\|_{cb} \leq 1 + 2\delta$ . Furthermore g is unital, and an easy calculation

shows that g is invertible with  $\|g^{-1}\|_{cb} \leq 1+10\delta$ . By Proposition 4.2.5 we can now conclude that  $d_{\mathcal{E}_1^{sy}}(\overline{a},\overline{b}) \leq 100k (10\delta)^{\frac{1}{2}} \leq 1000k\delta^{\frac{1}{2}}$ .

We are now ready to show that  $\left(\mathcal{E}_{1}^{sy}(k), d_{\mathcal{E}_{1}^{sy}}\right)$  is a complete metric space. The proof of this fact involves a standard argument; see also [114, Proposition 12].

**Proposition 4.2.9.** Suppose that  $k \in \mathbb{N}$ . Then  $\mathcal{E}_1^{sy}(k)$  is a complete metric space with respect to the metric  $d_{\mathcal{C}}$ , and  $\mathcal{M}^0_{\infty}(k)$  is a dense subspace.

*Proof.* Let us first show that  $\mathcal{E}_1^{sy}(k)$  is complete. Suppose that  $(\overline{a}^{(m)})_{m\in\mathbb{N}}$  is a Cauchy sequence in  $\mathcal{E}_1^{sy}(k)$ . It is not difficult to verify that there is  $N \in \mathbb{N}$  such that  $\overline{a}^{(m)}$  is N-Auerbach for every  $m \in \mathbb{N}$ . Fix a nonprincipal ultrafilter  $\mathcal{U}$  over  $\mathbb{N}$  and let  $E := \prod_{\mathcal{U}} \langle \overline{a}^{(m)} \rangle$  be the corresponding ultraproduct. Let  $a_i$  for  $i \leq k$  be the element of X with representative sequence  $(a_i^{(m)})_{m\in\mathbb{N}}$ . Let  $\phi_m : \langle \overline{a}^{(m)} \rangle \to E$  be the linear map such that

$$\phi_m\left(a_i^{(m)}\right) = a_i$$

By Lemma 4.2.7 and Los' theorem for ultraproducts [35, Proposition 4.3] we have that

$$\lim_{m \to +\infty} \left\| \phi_m \right\|_{cb} \left\| \phi_m^{-1} \right\|_{cb} \to 1.$$

Therefore X is 1-exact by Proposition 4.1.2. Moreover  $d_{\mathcal{E}_1^{sy}}(\overline{a}, \overline{a}^{(m)}) \to 0$  by Lemma 4.2.8. The fact that  $\mathcal{M}_{\infty}^0(k)$  is dense in  $\mathcal{E}_1^{sy}(k)$  is an immediate consequence of Proposition 4.2.6. To prove separability it is enough to show that  $\mathcal{M}_{\infty}^0(k)$  is separable. For every  $n \in \mathbb{N}$  let  $D_n \subset M_n$  be a self-adjoint countable dense subset containing the identity. Then the set of  $\overline{a} \in \mathcal{M}_{\infty}^0$  such that  $\overline{a} \subset D_n$  for some  $n \in \mathbb{N}$  is a countable dense subset of  $\mathcal{M}_{\infty}^0(k)$ .

This concludes the proof that  $\mathcal{E}_1^{sy}$  is a Fraïssé class. In view of Theorem 3.1.9 we can consider the corresponding limit. Equivalent characterizations of such a limit will be obtained in Subsection 4.3.1

## 4.3 The Gurarij operator system

## 4.3.1 The limit of the class $\mathcal{E}_1^{sy}$

**Definition 4.3.1.** A separable 1-exact operator system  $\mathbb{GS}$  is *Gurarij* if whenever  $E \subset F$  are finite-dimensional 1-exact operator systems,  $\phi : E \to \mathbb{GS}$  is a unital complete isometry, and  $\varepsilon > 0$ , there is a linear map  $\hat{\phi} : F \to \mathbb{GS}$  extending  $\phi$  such that  $\|\hat{\phi}\|_{cb} \|\hat{\phi}^{-1}\|_{cb} \leq 1 + \varepsilon$ .

The following characterization of the limit of  $\mathcal{E}_1^{sy}$  follows easily from our results above. The proof is analogous to the ones of [8, Theorem 3.3] and [90, Proposition 4.8].

**Proposition 4.3.2.** Suppose that Z is a separable 1-exact operator system. The following statements are equivalent:

- 1. Z is the Fraissé limit of the class  $\mathcal{E}_1^{sy}$  in the sense of Definition 3.1.7;
- 2. Whenever  $E \subset F$  are finite-dimensional 1-exact operator systems,  $\phi : E \to Z$  is a unital complete isometry, and  $\varepsilon > 0$ , there is a unital complete isometry  $\psi : F \to Z$  such that  $\|\psi_{|E} \phi\| \leq \varepsilon$ ;
- 3. Z satisfies the same condition as (2) when  $F = M_n$  for some  $n \in \mathbb{N}$ ;
- 4. Z is Gurarij;
- 5. Z satisfies the same conditions as in Definition 4.3.1 when  $F = M_n$  for some  $n \in \mathbb{N}$ ;
- 6. For every matricial operator space  $E, \psi \in \text{Stx}_{\mathcal{E}_1^{sy}}(E, Z)$ , finite subset B of E, and  $\varepsilon > 0$ , there is  $\phi \in \text{Apx}_{\mathcal{E}_1^{sy}}(E, Z)$  such that  $\phi \le \psi$  and for every  $b \in B$  there is  $z \in Z$  such that  $\phi(b, z) \le \varepsilon$ .

*Proof.* We present a proof of the nontrivial implications below.

- (1) $\Rightarrow$ (2) By definition of Fraïssé limit Z is a separable  $\mathcal{E}_1^{sy}$ -structure, and hence a separable 1-exact operator system by [13, Proposition 3.2]. Observe that the embeddings between structures in the language  $\mathcal{L}_{OSy}$  for operator systems are precisely the unital complete isometries. Therefore (2) is a reformulation of being limit as in Definition 3.1.7;
- (3) $\Rightarrow$ (5) Suppose that  $E \subset M_n$  and  $\phi : E \to Z$  is a unital complete isometry. Fix a strictly positive real number  $\delta$ . By hypothesis there is a unital complete isometry  $\psi: M_n \to Z$  such that  $\|\psi_{|E} \phi\| < 1 + \delta$ . If  $\delta$  is small enough, then by Lemma 4.2.7 one can perturb  $\psi$  to a unital linear map  $\hat{\phi}: M_n \to Z$  such that  $\|\hat{\phi}\|_{cb} \|\hat{\phi}^{-1}\|_{cb} \leq 1 + \varepsilon$  and  $\hat{\phi}_{|E} = \phi$ .
- (2) $\Rightarrow$ (4) Analogous to the implication (3) $\Rightarrow$ (5).
- (5) $\Rightarrow$ (6) Suppose that *E* is a matricial operator system,  $B \subset E$  is a finite subset,  $\psi \in$ Stx<sub> $\mathcal{E}_1^{sy}$ </sub> (*E*, *Z*), and  $\varepsilon > 0$ . Lemma 4.2.4 shows that the class  $\mathcal{M}^0_{\infty}$  of matricial operator

systems has (HP), (JEP), and (NAP) of Definition 3.1.2. Therefore by [8, Lemma 2.8(iii)], possibly after enlarging E and decreasing  $\varepsilon$ , we can assume that there is a finite subset A of E and a unital complete isometry

$$f: \operatorname{span}\left(\{1\} \cup A \cup A^*\right) \to Z$$

such that  $\psi \ge f_{|A} + \varepsilon$ . (Here as in Subsection 3.1.1 we identify a partial morphisms with the approximate morphism given by the distance function from its graph.) Let  $\bar{c}$  be linear basis for

$$\operatorname{span}\left(\{1\} \cup A \cup B \cup A^* \cup B^*\right),$$

and k be the length of  $\overline{c}$ . Fix  $\delta \in (0, 1]$ . By assumption, since E is matricial we can extend f to a unital linear map  $f: E \to Z$  such that  $||f||_{cb} \leq 1+\delta$  and  $||f^{-1}||_{cb} \leq 1+\delta$ . This implies that  $d_{\mathcal{E}_1^{sy}}(\overline{c}, f(\overline{c})) \leq 1000k\delta^{\frac{1}{2}}$  by Lemma 4.2.8. Therefore there are a finite-dimensional 1-exact operator system F and unital complete isometries  $\phi_0: \langle \overline{c} \rangle \to$ F and  $\phi_1: \langle f(\overline{c}) \rangle \to F$  such that  $||\phi_0(c_i) - (\phi_1 \circ f)(c_i)|| \leq 1000k\delta^{\frac{1}{2}}$  for every *i*. Let  $\phi: \langle \overline{c} \rangle \to \langle f(\overline{c}) \rangle$  be the composition of  $\phi_1^{-1}$  and  $\phi_0$  as approximate morphisms. It is clear that by choosing  $\delta$  small enough we can ensure that  $\phi(x, f(x)) \leq \varepsilon$  for every  $x \in A \cup B$ . Observe finally that

$$\phi \in \operatorname{Apx}\left(\langle \overline{a} \rangle, \langle f(\overline{a}) \rangle\right) \subset \operatorname{Apx}\left(E, Z\right)$$

and

$$\psi \ge f_{|A} + \varepsilon \ge \phi.$$

This concludes the proof.

(6) $\Rightarrow$ (1) Since by Proposition 4.2.9  $\mathcal{M}^0_{\infty}(k)$  of k-generated matricial operator systems is dense in  $\mathcal{E}^{sy}_1(k)$  for every  $k \in \mathbb{N}$ , the conclusion follows from the characterization of the Fraïssé limit provided by [8, Lemma 2.16].

#### 4.3.2 Existence and uniqueness of the Gurarij operator system

Recall that we have shown in Subsection 4.2.1 and Subsection 4.2.3 that  $\mathcal{E}_1^{sy}$  is a Fraïssé class. We can therefore consider its limit. Proposition 4.3.2 shows that being limit of  $\mathcal{E}_1^{sy}$  is equivalent to being a Gurarij operator system.

The following theorem is now an immediate consequence of Ben Yaacov's main results on Fraïssé limits of metric structures; see Theorem 3.1.9. The following lemma can be obtained from Proposition 4.2.5, similarly as Lemma 2.2 is obtained from Lemma 2.1 in [83].

**Lemma 4.3.3.** Suppose that  $E \subset \mathbb{GS}$  is a finite-dimensional subsystem, Y is a finitedimensional 1-exact operator system,  $\delta \in (0,1]$ , and  $f: E \to Y$  is a unital invertible linear map such that  $||f||_{cb} < 1 + \delta$  and  $||f^{-1}||_{cb} < 1 + \delta$ . Then for every  $\eta > 0$  there exists a  $g: Y \to \mathbb{NG}$  such that  $||g||_{cb} < 1 + \eta$ ,  $||g^{-1}||_{cb} < 1 + \eta$ , and  $||g \circ f - id_X||_{cb} < 100 \dim(E) \delta^{\frac{1}{2}}$ .

**Theorem 4.3.4.** There is a Gurarij operator system GS as in Definition 4.3.1. Such an operator system is unique up to complete order isomorphism. Any separable 1-exact operator system embeds unitally and completely isometrically into GS. Furthermore GS has the following homogeneity property: if E is a finite-dimensional subsystem of GS,  $\delta \in (0, 1]$ , and  $\phi : X \to \mathbb{GS}$  is a unital invertible linear map such that  $\|\phi\|_{cb} < 1 + \delta$  and  $\|\phi^{-1}\|_{cb} < 1 + \delta$ , then there exists a complete order automorphism  $\alpha$  of GS such that  $\|\phi - \alpha_{|E}\|_{cb} < 100 \dim (X) \delta^{\frac{1}{2}}$ .

*Proof.* The existence, uniqueness, and universality statements follow from Theorem 3.1.9. The proof of the homogeneity property is analogous to the proof of [83, Theorem 1.1] where [83, Lemma 2.2] is replaced by Lemma 4.3.3.  $\Box$ 

## 4.3.3 An explicit construction of $\mathbb{GS}$

Lemma 4.2.4 shows that the class of matricial operator systems is Fraïssé. It follows from this observation and the proof of the existence statement for Fraïssé limits [8, Lemma 2.7] that  $\mathbb{GS}$  can be written as a direct limit  $\lim_{(\phi_k)} X_k$ , where  $X_k$  is an operator system completely order isomorphic to  $M_{n_k}$  for some  $n_k \in \mathbb{N}$  and  $\phi_k : X_k \to X_{k+1}$  is a complete order embedding. We will present in this subsection an explicit construction of  $\mathbb{GS}$  that makes this fact apparent.

Let us say that a subset D of a metric space X is  $\varepsilon$ -dense for some  $\varepsilon > 0$  if every element of X is at distance at most  $\varepsilon$  from some element of D. For  $m, k \in \mathbb{N}$ , let  $D_{m,k}$ be a finite  $2^{-k}$ -dense subset of the unit ball of  $M_m$ , and  $(\overline{a}_{m,k,i})$  be an enumeration of the finite tuples from  $D_{m,k}$ . Set  $E_{m,k,i} = \langle \overline{a}_{m,k,i} \rangle \subset M_m$ . We define by recursion on k sequences  $(n_k), (X_k), (D_k^X), (\phi_k), (f_{m,k,i,j})$  such that

- 1.  $X_k$  is an operator system completely order isomorphic to  $M_{n_k}$ ;
- 2.  $D_k^X$  is a  $2^{-k}$ -dense subset of the ball of radius 2 of  $X_k$ ;

- 3.  $\phi_k : X_k \to X_{k+1}$  is a complete order embedding;
- 4.  $(f_{m,k,i,j})$  enumerates all the unital linear maps f from  $E_{m,k,i}$  to  $X_k$  such that  $f(\overline{a}_{m,k,i}) \subset D_k^X$ ;
- 5. for every m, k, i, j such that  $m \leq k$  there exists a complete order embedding  $g_{m,k,i,j}$ :  $M_m \to X_{k+1}$  such that

$$\|g_{m,k,i,j} - \phi_k \circ f_{m,k,i,j}\|_{cb}$$

$$\leq 100 \dim (E_{m,k,i}) \left( \max \left\{ \|f_{m,k,i,j}\|_{cb}, \left\|f_{m,k,i,j}^{-1}\right\|_{cb} \right\} - 1 \right)^{\frac{1}{2}}.$$

Granted the construction define  $X = \lim_k X_k$  and identify  $X_k$  with its image inside X. Suppose that  $m \in \mathbb{N}$ ,  $E \subset M_m$  is a subsystem,  $\varepsilon > 0$ , and  $f : E \to X$  is a complete order embedding. Fix  $\eta \in (0, 1)$  small enough. By the small perturbation argument [17, Lemma 12.3.15] there exist  $k, i, j \in \mathbb{N}$  such that  $m \leq k$ ,  $\|f_{m,k,i,j} - f\|_{cb} \leq \eta$ ,  $\|f_{m,k,i,j}\|_{cb} \leq 1 + \eta$ , and  $\|f_{m,k,i,j}^{-1}\|_{cb} \leq 1 + \eta$ . Therefore

$$\begin{aligned} \|g_{m,k,i,j} - f\|_{cb} &\leq \|g_{m,k,i,j} - f_{m,k,i,j}\|_{cb} + \|f_{m,k,i,j} - f\|_{cb} \\ &\leq 100m\eta^{\frac{1}{2}} + \eta \leq \varepsilon \end{aligned}$$

for  $\eta$  small enough. This proves that X satisfies Condition (3) of Proposition 4.3.2. Since X is 1-exact, it follows that X is completely order isomorphic to GS. We now show how to recursively define sequences as above. Let  $n_1 = 1$ ,  $X_1 = \mathbb{C}$ . Suppose that the sequences  $(n_k), (X_k), (D_k^X), (\phi_{k-1}), (f_{m,k,i,j})$  have been defined for  $k \leq l$ . Observe that condition (5) only concerns finitely many functions  $f_{m,k,i,j}$ . Therefore one can obtain  $X_{k+1}, \phi_k : X_k \to X_{k+1}$ , and  $(g_{m,k,i,j})$  by repeatedly applying Lemma 4.2.4. This concludes the construction.

#### 4.3.4 Unital operator spaces

In this subsection we want to observe that  $\mathbb{GS}$  is of universal disposition for 1-exact unital operator spaces. Recall that a unital operator space is an operator space with a distinguished unitary element in the sense of [13]. A unital operator space can be concretely represented as a unital subspace of B(H) with the inherited matrix norms and the identity operator as distinguished unitary. If  $X \subset B(H)$  is a unital operator space, define the operator system  $X + X^* = \{x_1 + x_2^* : x_1, x_2 \in X\}$ . Recall that if  $\phi : X \to Y$  is a unital completely contractive (resp. completely isometric) map from X into an operator system Y, then  $x_1 + x_2^* \mapsto \phi(x_1) + \phi(x_2)^*$  is a unital completely contractive (resp. completely isometric) map; see [12, Lemma 1.3.6]. This implies that the complete order isomorphism class of  $X + X^*$  does not depend from the concrete representation of X as a unital subspace of B(H).

A unital operator space is 1-exact if it is 1-exact as an operator space. We now want to observe that a unital operator space X is 1-exact if and only if the operator system  $X + X^*$  is 1-exact. The proof of [114] shows that a unital operator space X is 1-exact if and only if for every finite-dimensional unital operator space  $E \subset X$  and for every  $\delta > 0$  there exist  $n \in \mathbb{N}$  and a unital completely contractive map  $\phi : E \to M_n$  such that  $\|\phi^{-1}\|_{cb} \leq 1+\delta$ . The proof the following lemma is analogous to the proof of Proposition 4.2.6 with the additional ingredient of [12, Lemma 1.3.6].

**Lemma 4.3.5.** Suppose that X is a 1-exact unital operator space. Then X admits a unital completely isometric embedding into an inductive limit of full matrix algebras with unital completely isometric connective maps.

Suppose now that E is a finite-dimensional unital 1-exact operator space. By Lemma 4.3.5 we can assume that  $E \subset Z$ , where Z is an inductive limit of full matrix algebras with unital completely isometric connective maps. Hence

$$E + E^{\star} = \{x_1 + x_2^* : x_1, x_2 \in E\} \subset Z.$$

Let  $\overline{a}$  be a basis for E and  $\varepsilon > 0$ . Pick  $\delta > 0$  small enough. There exist a subsystem  $Z_0 \subset Z$  completely order isomorphic to a full matrix algebra and a tuple  $\overline{b}$  in  $Z_0$  such that  $\max_i ||a_i - b_i|| \leq \delta$  and, hence,  $\max_i ||a_i^* - b_i^*|| \leq \delta$ . The proof of Lemma 4.2.7 shows that for  $\delta$  small enough one has that  $d_{cb}(E + E^*, F) \leq \varepsilon$  for some subsystem F of  $Z_0$ . This shows that  $E + E^*$  is a 1-exact operator system. The fact that a unital operator space X is 1-exact if and only if  $X + X^*$  is 1-exact now follows immediately.

**Proposition 4.3.6.** If  $E \subset F$  are unital 1-exact operator spaces,  $\phi : E \to \mathbb{GS}$  is a unital complete isometry, and  $\varepsilon > 0$ , then there exists a unital complete isometry  $\psi : F \to \mathbb{GS}$  such that  $\|\psi_{|E} - \phi\|_{cb} \leq 1 + \varepsilon$ .

Proof. Since F is 1-exact, also  $F + F^*$  is 1-exact. Moreover by [12, Lemma 1.3.6] the map  $\phi$ :  $E + E^* \to \mathbb{GS}$  defined by  $x_1 + x_2^* \mapsto \phi(x_1) + \phi(x_2)^*$  is a unital complete isometry. Therefore there is a unital complete isometry  $\tilde{\psi} : F + F^* \to \mathbb{GS}$  such that  $\left\| \tilde{\psi}_{|E+E^*} - \tilde{\phi} \right\|_{cb} \leq 1 + \varepsilon$ . Setting  $\psi := \tilde{\psi}_{|E}$  yields a unital complete isometry from F to  $\mathbb{GS}$  such that  $\left\| \psi - \phi \right\|_{cb} \leq 1 + \varepsilon$ . Let us now consider  $\mathbb{GS}$  as a unital operator space, i.e. a structure in the language  $\mathcal{L}_{uOS}$  of unital operator spaces. Proposition 4.3.6 shows that  $\mathbb{GS}$  is a homogeneous  $\mathcal{L}_{uOS}$ -structure. Therefore the age of  $\mathbb{GS}$  in the language  $\mathcal{L}_{uOS}$ , which is the collection of finite-dimensional 1-exact unital operator spaces, is a Fraïssé class with limit  $\mathbb{GS}$ .

#### 4.3.5 A unital function system of almost universal disposition

We will now consider the class of  $M_n$ -systems defined in Subsection 4.1.4. Let us say that a unital  $M_n$ -space is a subspace of an  $M_n$ -system containing the unit. Observe that a unital  $M_n$ -space is in particular an  $M_n$ -space in the sense of [12, 4.1.1]. We can regard unital  $M_n$ -spaces as structures in a language  $\mathcal{L}_{uM_n}$ , which is the language of  $M_n$ -spaces described in [90, Subsection 2.5] with the addition of a constant symbol for the unit. Similarly  $M_n$ systems can be regarded as structures in the language  $\mathcal{L}_{M_n^{sy}}$  obtained from  $\mathcal{L}_{uM_n}$  adding a symbol for the involution.

The proofs of Lemma 4.3.7 and Proposition 4.3.8 is analogous to the proofs of Lemma 4.2.4 and Lemma 4.2.5. We denote the  $\infty$ -sum of k copies of  $M_n$  by  $\ell_k^{\infty}(M_n)$ .

**Lemma 4.3.7.** Suppose that  $F_0, F_1 \subset \ell_k^{\infty}(M_n), E \subset F_0$ , and  $\delta \in [0,1]$ . If  $f: E \to F_1$  is an invertible unital map such that  $||f||_{cb} \leq 1 + \delta$  and  $||f^{-1}||_{cb} \leq 1 + \delta$ , then there exist  $d \in \mathbb{N}$  and unital completely isometric embeddings  $i: F_0 \to \ell_k^{\infty}(M_n)$  and  $j: F_1 \to \ell_k^{\infty}(M_n)$  such that  $||j \circ f - i_{|E}||_{cb} \leq 100 \dim (E) \delta^{\frac{1}{2}}$ .

**Proposition 4.3.8.** Suppose that X and Y are  $M_q$ -systems,  $E \subset X$  is a finite-dimensional subsystem,  $\varepsilon > 0$ , and  $\delta \in [0, 1]$ . If  $f : E \to Y$  is a unital map such that  $||f||_{cb} \leq 1 + \delta$  and  $||f^{-1}||_{cb} \leq 1 + \delta$ , then there exist a separable  $M_q$ -system Z and unital completely isometric embeddings  $i : X \to Z$  and  $j : Y \to Z$  such that  $||j \circ f - i||_{cb} \leq 100 \dim (E) \delta^{\frac{1}{2}} + \varepsilon$ .

Proposition 4.3.8 shows that  $M_n$ -systems form a Fraïssé class. Arguing as before, one can obtain the following characterization for the limit  $\mathbb{G}_n^u$  of the Fraïssé class of  $M_n$ -systems.

**Proposition 4.3.9.** Suppose that Z is a separable  $M_n$ -system. The following conditions are equivalent:

- 1. Z is completely order isomorphic to  $\mathbb{G}_n^u$ ;

3. whenever  $E \subset F$  are finite-dimensional  $M_n$ -systems,  $\phi : E \to Z$  is a unital complete isometry, and  $\varepsilon > 0$ , there is a unital linear map  $\psi : F \to Z$  such that  $\left\| \widehat{\psi} \right\|_{cb} \left\| \widehat{\psi}^{-1} \right\|_{cb} \leq 1 + \varepsilon$ .

Similar characterizations hold when E and F in (2) and (3) are only assumed to be finite-dimensional unital  $M_n$ -spaces. Observe that unital  $M_1$ -spaces are precisely unital function spaces, i.e. subspaces of C(K) for some compact Hausdorff space K containing the function  $1_K$  constantly equal to 1. Similarly  $M_1$ -systems are function systems, i.e. subspaces of C(K) for some compact Hausdorff space K containing  $1_K$  and closed under taking adjoints. We can therefore conclude from Proposition 4.3.9 that  $\mathbb{G}_1^u$  is a separable unital function system of almost universal disposition for unital function spaces. In other words whenever  $E \subset F$  are unital function spaces,  $\phi : E \to \mathbb{G}_1^u$  is a unital isometry, and  $\varepsilon \geq 0$ , there is a unital isometry  $\psi : F \to \mathbb{G}_1^u$  such that  $\|\psi|_E - \phi\| \leq 1 + \varepsilon$ .

## 4.3.6 The C\*-envelope of the Gurarij operator system

We now want to show that the canonical \*-homomorphism from the universal C\*-algebra of  $\mathbb{GS}$  to the C\*-envelope of  $\mathbb{GS}$  is a \*-isomorphism.

Suppose that X is an operator system. A C\*-cover of X is a unital completely isometric embedding  $\phi$  of X into a C\*-algebra A such that the image of X under  $\phi$  generates A as a C\*-algebra. It was shown by Hamana [55, 54] and, independently, Ruan [124] that there always exists a (projectively) minimal C\*-cover  $i_X : X \to C_e^*(X)$ , called the C\*-envelope of X (or the regular C\*-algebra in [79]). The C\*-envelope has the property that whenever  $\phi : X \to A$  is a C\*-cover, there is a (necessarily unique and surjective) \*-homomorphism  $\pi : A \to C_e^*(X)$  such that  $\pi \circ \phi = i_X$ .

Similarly it is shown in [79, Section 3] that there always exists a (projectively) maximal C\*-cover  $u_X : X \to C_u^*(X)$ , called the *universal C\*-algebra* of X. This has the property that whenever  $\phi : X \to A$  is a C\*-cover, there is a (necessarily unique and surjective) \*-homomorphism  $\pi : C_u^*(X) \to A$  such that  $\pi \circ u_X = \phi$ . In particular there is a \*-homomorphism  $\sigma_X : C_u^*(X) \to C_e^*(X)$ .

An operator system X for which  $\sigma_X$  is injective (or, equivalently, a \*-isomorphism) is called *universal* in [79]. In particular this property implies that if  $\phi : X \to A$  is any C\*-cover of X, then  $A \cong C_u^*(X) \cong C_e^*(X)$ . It is proved in [79, Theorem 15 and Corollary 18] that if X is a universal operator system of dimension at least 2, then X does not embed unitally completely isometrically into any exact C\*-algebra. An example of a nuclear separable universal operator system is constructed in [79, Theorem 17]. In the rest of this subsection we will show that the Gurarij operator system is universal in the sense of [79]. The argument is similar to the one in the proof of [79, Theorem 17]. In the following we identify an operator system X with its image inside the C<sup>\*</sup>envelope  $C_e^*(X)$ , and we denote by  $u_X : X \to C_u^*(X)$  the canonical embedding of X into its universal C<sup>\*</sup>-algebra. It is shown in [79, Proposition 9] that a unital complete isometry  $\phi : X \to Y$  between operator system has a unique "lift" to an injective \*-homomorphism  $\overline{\phi} : C_u^*(X) \to C_u^*(Y)$  such that  $\overline{\phi} \circ u_X = u_Y \circ \phi$ . In particular this allows one to identify the universal C<sup>\*</sup>-algebra of an inductive limit with the inductive limit of the universal C<sup>\*</sup>-algebras of the building blocks.

**Theorem 4.3.10.** The canonical \*-homomorphism  $\sigma_{\mathbb{GS}} : C_u^*(\mathbb{GS}) \to C_e^*(\mathbb{GS})$  is a \*isomorphism.

Proof. As observed in Subsection 4.3.3,  $\mathbb{GS}$  is the limit of an inductive sequence  $(X_k)_{k\in\mathbb{N}}$  with unital completely isometric connective maps  $\phi_k : X_k \to X_{k+1}$ , where  $X_k$  is completely order isomorphic to a full matrix algebra. Denote by  $\iota_k : X_k \to \mathbb{GS}$  the canonical inclusion, and observe that the set of elements of the form  $\bar{\iota}_k(z)$  for  $k \in \mathbb{N}$  and  $z \in C_u^*(X_k)$  are dense in  $\mathbb{GS}$ . Suppose that  $\bar{\iota}_k(z)$  is such an element. Fix  $\delta > 0$  small enough. By [79, Corollary 7] there exist  $d \in \mathbb{N}$  and a unital \*-homomorphism  $\pi : C_u^*(X_k) \to M_d$  such that  $\pi \circ u_{X_k}$  is a unital complete isometry and  $\|\pi(z)\| \ge (1-\varepsilon) \|z\|$ . Set  $\theta := \pi \circ u_{X_k} : X_k \to M_d$ , and observe that  $\pi$  is the unique \*-homomorphism from  $C_u^*(X_k)$  to  $M_d$  such that  $\pi \circ u_{X_k} = \theta$ . By the approximate injectivity property of  $\mathbb{GS}$ —see Proposition 4.3.2—there is a unital complete isometry  $\eta : M_d \to \mathbb{GS}$  such that  $\|\eta \circ \theta - \iota_k\|_{cb} \le \delta$ . Observe that, for  $\delta$  small enough, we have

$$\left\| \left( \overline{\eta \circ \theta} \right)(z) - \overline{\iota}_k(z) \right\| \le \varepsilon.$$

By [18, Theorem 4.1]—see also [79, Lemma 6]—there is a \*-homomorphism  $\mu : C^*(\eta[M_d]) \subset C_e^*(\mathbb{GS}) \to M_d$  such that  $\mu \circ \eta = id_{M_d}$ . Observe now that

$$\mu \circ \sigma_{\mathbb{GS}} \circ \overline{\eta \circ \theta} : C_u^*(X_k) \to Y$$

is a \*-homomorphism such that

$$\mu \circ \sigma_{\mathbb{GS}} \circ \overline{\eta \circ \theta} \circ u_{X_k} = \mu \circ \sigma_{\mathbb{GS}} \circ u_{\mathbb{GS}} \circ \eta \circ \theta$$
$$= \mu \circ \eta \circ \theta$$
$$= \theta.$$

Therefore  $\mu \circ \sigma_{\mathbb{GS}} \circ \overline{\eta \circ \theta} = \pi$ . Hence we have that

$$\begin{aligned} \|\sigma_{\mathbb{GS}}\left(\bar{\iota}_{k}\left(z\right)\right)\| &\geq \|\left(\sigma_{\mathbb{GS}}\circ\overline{\eta\circ\theta}\right)\left(z\right)\|-\varepsilon\\ &\geq \|\left(\mu\circ\sigma_{\mathbb{GS}}\circ\overline{\eta\circ\theta}\right)\left(z\right)\|-\varepsilon\\ &= \|\pi\left(z\right)\|-\varepsilon\\ &\geq \|\bar{\iota}_{k}\left(z\right)\|-2\varepsilon. \end{aligned}$$

This concludes the proof that  $\sigma$  is injective.

**Corollary 4.3.11.** The Gurarij operator system  $\mathbb{GS}$  does not admit any complete order embedding into an exact C\*-algebra. Moreover if  $\mathbb{GS} \subset B(H)$  is a unital completely isometric representation of  $\mathbb{GS}$ , then the C\*-algebra generated by  $\mathbb{GS}$  inside B(H) is \*-isomorphic to  $C_e^*(\mathbb{GS})$ .

In particular it follows from Corollary 4.3.11 that  $\mathbb{GS}$  is not completely order isomorphic to a unital C\*-algebra. (Recall that a C\*-algebra is exact if and only if it is 1-exact as an operator system.) In fact by [13, Proposition 4.2] one can also conclude that  $\mathbb{GS}$  is not completely isometric to a unital operator algebra (even without assuming that the complete isometry preserves the unit).

## 4.3.7 The triple envelope of the Gurarij operator space

The (noncommutative) Gurarij operator space NG has been defined by Oikhberg in [104] and proven to be unique and universal among separable 1-exact operator spaces in [90]. A 1-exact operator space Z is completely isometric to NG if and only if for every  $n \in \mathbb{N}$ ,  $E \subset M_n$ , complete isometry  $\phi : E \to \mathbb{NG}$ , there exists a complete isometry  $\psi : F \to \mathbb{NG}$  such that  $\|\psi_{|E} - \phi\|_{cb} \leq \varepsilon$ . A similar argument as the one in Subsection 4.3.3 shows that  $\mathbb{NG}$  is completely isometric to a direct limit  $\lim_{(\phi_k)} X_k$  where  $X_k$  is an operator space completely isometric to  $M_{n_k}$  for some  $n_k \in \mathbb{N}$  and  $\phi_k : X_k \to X_{k+1}$  is a complete isometry. Here one needs to replace Lemma 4.2.4 with the following lemma, which can be proved in the same way using injectivity of  $M_d$  in the category of operator spaces with completely contractive maps.

**Lemma 4.3.12.** Suppose that  $k, m \in \mathbb{N}$ ,  $F_0$ ,  $F_1$  are matricial operator spaces,  $E \subset F_0$ , and  $\delta > 0$ . If  $f : E \to F_1$  is an invertible linear map such that  $||f||_{cb} \leq 1 + \delta$  and  $||f^{-1}||_{cb} \leq 1 + \delta$ , then there exist  $d \in \mathbb{N}$  and unital completely isometric embeddings  $i : F_0 \to M_d$  and  $j : F_1 \to M_d$  such that  $||j \circ f - i|_E||_{cb} \leq 2\delta$ .

Here we want to point out that the methods employed in Subsection 4.3.6 can used to show that  $\mathbb{NG}$  is not completely isometric to a C\*-algebra. This gives an answer to [104, Remark 3.3]. The key idea is to replace the C\*-envelope with the triple envelope, and the universal C\*-algebras with the universal TRO. Recall that a *ternary ring of operators* (TRO) is a subspace V of B(H, K) for some Hilbert spaces H, K such that  $xy^*z \in V$  for any  $x, y, z \in V$ . The operation  $(x, y, z) \mapsto xy^*z$  on V is called *triple product*. A *triple morphism* between TROs is a linear map that preserves the triple product. Observe that (the restriction of) a \*-homomorphism is in particular a triple morphism. A TRO has a canonical operator space structure, where the matrix norms are uniquely determined by the triple product [69, Proposition 2.1]. A triple morphism between TROs is automatically completely contractive, and it is injective if and only if is completely isometric [12, Lemma 8.3.2].

Suppose that X is an operator space. A triple cover of X is a pair  $(\phi, V)$  where V is a TRO and  $\phi : X \to V$  is a linear complete isometry. Triple covers naturally form a category, where a morphism from  $(\phi, V)$  to  $(\phi', V')$  is a triple morphism  $\psi : V \to V'$  such that  $\psi \circ \phi = \phi'$ . The (unique up to isomorphism) terminal object in such a category is called the triple envelope  $\mathcal{T}_e(X)$  of X. The existence of such an object has been proved independently by Ruan [124] and Hamana [56]. The (unique up to isomorphism) initial object in the category of triple covers is the universal TRO  $\mathcal{T}_u(X)$  of X. We could not find the existence of such an object explicitly stated in the literature. However this can be easily established with a minor modification of the proof of the existence of the universal C\*-algebra of an operator system; see [79, Proposition 8]. Moreover the same proof as [79, Proposition 9]—where one replaces Arveson's extension theorem with Wittstock's extension theorem [107, Theorem 8.2]—shows that a complete isometry between TROS "lifts" to an injective triple morphism between the corresponding universal TROs.

The universal property of  $\mathcal{T}_u(X)$  yields a (necessarily surjective) canonical triple morphism  $\sigma_X : \mathcal{T}_u(X) \to \mathcal{T}_e(X)$ . Our goal is to show that such a triple morphism is injective (and hence a triple isomorphism) in the case of the Gurarij operator space NG. First we need an adaptation to TROs of a well known result of Choi and Effros; see [18, Theorem 4.1].

**Proposition 4.3.13.** Suppose that V, W are TROs and  $\theta : V \to W$  is a linear complete isometry. Then there is a triple morphism  $\eta : \mathcal{T}(\theta[V]) \to V$  such that  $\eta \circ \theta = id_V$ , where  $\mathcal{T}(\theta[V])$  is the subTRO of W generated by the image of V under  $\theta$ .

*Proof.* The proof consists in reducing the problem to the unital case via the Paulsen trick;

see [12, 1.3.14]. Represent V as a subspace of  $B(H_1, K_1)$  and define the Paulsen system

$$\mathcal{S}(V) = \begin{bmatrix} \mathbb{C}I_{H_1} & V \\ V^* & \mathbb{C}I_{K_1} \end{bmatrix} = \left\{ \begin{bmatrix} \lambda & x \\ y^* & \mu \end{bmatrix} : x, y \in V, \lambda, \mu \in \mathbb{C} \right\} \subset B(H_1 \oplus K_1).$$

One can similarly define the Paulsen system  $\mathcal{S}(W) \subset B(H_2 \oplus K_2)$  of W. The linear complete isometry  $\theta: V \to W$  yields a linear map  $\tilde{\theta}: \mathcal{S}(V) \to \mathcal{S}(W)$  defined by

$$\begin{bmatrix} \lambda & x \\ y^* & \mu \end{bmatrix} \mapsto \begin{bmatrix} \lambda & \theta(x) \\ \theta(y)^* & \mu \end{bmatrix}.$$

By [12, Lemma 1.3.15]  $\tilde{\theta}$  is a unital complete isometry. Let  $\tilde{\theta}^{-1} : \tilde{\theta}[\mathcal{S}(V)] \to \mathcal{S}(V) \subset B(H_1 \oplus K_1)$  be the inverse of  $\tilde{\theta}$ . By the Arveson extension theorem [107, Theorem 7.5] there is a unital completely positive map

$$\eta: C^*\left(\widetilde{\theta}\left[\mathcal{S}\left(V\right)\right]\right) \to B\left(H_1 \oplus K_1\right)$$

extending  $\tilde{\theta}$ . Here  $C^*\left(\tilde{\theta}[\mathcal{S}(V)]\right)$  is the C\*-algebra generated by  $\tilde{\theta}[\mathcal{S}(V)]$  inside  $B(H_2 \oplus K_2)$ . We claim that  $\eta$  is a \*-homomorphism. By Choi's multiplicative domain theorem [107, Theorem 3.18] it is enough to show that for every self-adjoint element a of  $\mathcal{S}(V)$ , one has that  $\eta\left(\theta(a)^2\right) = \eta\left(\theta(a)\right)^2$ . This easily follow from the Kadison-Schwartz inequality [12, Proposition 1.3.9] applied to  $\eta$  and  $\theta$ . Therefore  $\eta$  is a \*-homomorphism, and its restriction to the subTRO  $\mathcal{T}(\theta[V]) \subset B(H_2, K_2)$  of W generated by the image of V under  $\theta$  is a triple morphism. Since moreover  $\eta \circ \theta$  is the identity of V and V is a TRO, it follows that  $\eta$  maps  $\mathcal{T}(\theta[V])$  into V. This concludes the proof.

**Theorem 4.3.14.** The canonical triple morphism  $\sigma_{\mathbb{NG}} : \mathcal{T}_u(\mathbb{NG}) \to \mathcal{T}_e(\mathbb{NG})$  is a triple isomorphism.

Proof. The proof is entirely similar to the one of Theorem 4.3.10. Here one needs to use the fact that NG is the inductive limit of an inductive sequence  $(X_k)$  of operator spaces with completely isometric connective maps  $\phi_k : X_k \to X_{k+1}$ , as observed above. Moreover by [90, Theorem 4.12] NG satisfies the following approximate injectivity property: whenever  $E \subset F$  are 1-exact finite-dimensional operator spaces,  $\phi : E \to \mathbb{NG}$  is a linear complete isometry, and  $\varepsilon > 0$ , then there is a linear complete isometry  $\psi : F \to \mathbb{NG}$  such that  $\|\psi_{|E} - \phi\|_{cb} < \varepsilon$ . Finally one needs to replace the use of [18, Theorem 4.1] with Proposition 4.3.13.

We now discuss how Theorem 4.3.14 implies that NG does not admit any completely isometric embedding into an exact TRO. One can canonically assign to a TRO a C\*-algebra called its *linking algebra*. The local properties of a TRO are closely reflected by the local properties of its linking algebra. In particular a TRO is exact if and only if it is 1-exact if and only if its linking algebra is exact [69, Theorem 4.4]. Moreover a (surjective) triple morphism between TROs induces a (surjective) \*-homomorphism between the corresponding linking algebras. Since the class of exact C\*-algebras is closed under quotients [17, Corollary 9.4.3], it follows that the image of an exact TRO under a triple morphism is exact.

Observe now that if X is an operator system,  $C_u^*(X)$  is its universal C\*-algebra, and  $\mathcal{T}_u(X)$  is the triple envelope of X, then the universal property of  $\mathcal{T}_u(X)$  implies the existence of a surjective triple morphism from  $\mathcal{T}_u(X)$  to  $C_u^*(X)$ . If  $C_u^*(X)$  is not exact, then  $\mathcal{T}_u(X)$  is not exact as well. Since the universal C\*-algebra of  $M_2(\mathbb{C})$  is not exact [79, Section 5], it follows that the universal TRO of  $M_2(\mathbb{C})$  is not exact. Recall that, as observed above, completely isometric embeddings between operator spaces "lift" to injective ternary morphisms between the corresponding universal TROs. Since  $M_2(\mathbb{C})$  embeds completely isometrically into NG, it follows that  $\mathcal{T}_u(M_2(\mathbb{C}))$  embeds as a subTRO of  $\mathcal{T}_u(\mathbb{NG}) \cong \mathcal{T}_e(\mathbb{NG})$ . Therefore  $\mathcal{T}_e(\mathbb{NG})$  is not exact. Now, if NG embeds completely isometrically into a TRO V, then the universal property of  $\mathcal{T}_e(\mathbb{NG})$  implies that  $\mathcal{T}_e(\mathbb{NG})$  is the image of under a triple morphism of the subTRO of V generated by (the image of) NG. Hence V is not exact as well.

**Corollary 4.3.15.** The Gurarij operator space  $\mathbb{NG}$  does not admit any unital completely isometric embedding into an exact C\*-algebra. Moreover if  $\mathbb{NG} \subset B(H)$  is a unital completely isometric representation of  $\mathbb{NG}$  then the TRO generated by  $\mathbb{NG}$  inside B(H) is \*-isomorphic to  $\mathcal{T}_e(\mathbb{NG})$ .

Since  $\mathbb{NG}$  is 1-exact, it follows in particular that  $\mathbb{NG}$  is not completely isometric to a TRO or a C\*-algebra.

# Bibliography

- Charles A. Akemann, George A. Elliott, Gert K. Pedersen, and Jun Tomiyama, Derivations and multipliers of C\*-algebras, American Journal of Mathematics 98 (1976), no. 3, 679–708.
- [2] Charles A. Akemann and Gert K. Pedersen, Central sequences and inner derivations of separable C\*-algebras, American Journal of Mathematics 101 (1979), no. 5, 1047– 1061.
- [3] Claire Anantharaman-Delaroche, Old and new about treeability and the Haagerup property for measured groupoids, arXiv:1105.5961 (2011).
- [4] Martn Argerami, Samuel Coskey, Mehrdad Kalantar, Matthew Kennedy, Martino Lupini, and Marcin Sabok, The classification problem for finitely generated operator systems and spaces, arXiv:1411.0512 (2014).
- [5] William Arveson, An invitation to C\*-algebras, Springer-Verlag, New York-Heidelberg, 1976, Graduate Texts in Mathematics, No. 39.
- [6] Paul Bankston, Reduced coproducts of compact Hausdorff spaces, The Journal of Symbolic Logic 52 (1987), no. 2, 404–424.
- [7] Howard Becker and Alexander S. Kechris, *The descriptive set theory of polish group actions*, London Mathematical Society Lecture Note Series, vol. 232, Cambridge University Press, Cambridge, 1996.
- [8] Itaï Ben Yaacov, *Fraïssé limits of metric structures*, Journal of Symbolic Logic, to appear.
- [9] Itaï Ben Yaacov, Alexander Berenstein, C. Ward Henson, and Alexander Usvyatsov, Model theory for metric structures, Model theory with applications to algebra and

analysis. Vol. 2, London Mathematical Society Lecture Note Series, vol. 350, Cambridge University Press, 2008, p. 315–427.

- [10] Itaï Ben Yaacov, Andre Nies, and Todor Tsankov, A Lopez-Escobar theorem for continuous logic, arXiv:1407.7102 (2014).
- [11] Bruce Blackadar, Operator algebras, Encyclopaedia of Mathematical Sciences, vol. 122, Springer-Verlag, Berlin, 2006.
- [12] David P. Blecher and Christian Le Merdy, Operator algebras and their modules—an operator space approach, London Mathematical Society Monographs. New Series, vol. 30, Oxford University Press, Oxford, 2004.
- [13] David P. Blecher and Matthew Neal, Metric characterizations of isometries and of unital operator spaces and systems, Proceedings of the American Mathematical Society 139 (2011), no. 3, 985–998.
- [14] Béla Bollobás, *Linear analysis*, second ed., Cambridge University Press, Cambridge, 1999.
- [15] Joan Bosa, Nathanial P. Brown, Yasuhiko Sako, Aaron Tikuisis, Stuart White, and Wilhelm Winter, Covering dimension of C\*-algebras and the classification of maps by traces, in preparation.
- [16] Ola Bratteli, Inductive limits of finite dimensional C\*-algebras, Transactions of the American Mathematical Society 171 (1972), 195–234.
- [17] Nathanial P. Brown and Narutaka Ozawa, C\*-algebras and finite-dimensional approximations, Graduate Studies in Mathematics, vol. 88, American Mathematical Society, 2008.
- [18] Man-Duen Choi and Edward G. Effros, The completely positive lifting problem for C\*-algebras, The Annals of Mathematics 104 (1976), no. 3, 585.
- [19] \_\_\_\_\_, Injectivity and operator spaces, Journal of Functional Analysis 24 (1977), no. 2, 156–209.
- [20] J. Dixmier, Points séparés dans le spectre d'une C\*-algèbre, Acta Universitatis Szegediensis. Acta Scientiarum Mathematicarum 22 (1961), 115–128.

- [21] Jacques Dixmier, Traces sur les C\*-algbres, Universit de Grenoble. Annales de l'Institut Fourier 13 (1963), no. fasc. 1, 219–262.
- [22] Christopher J. Eagle, Ilijas Farah, Bradd Hart, Boris Kadets, Vladyslav Kalashnyk, and Martino Lupini, *Fraïssé limits of C\*-algebras*, arXiv:1411.4066 (2014).
- [23] Edward G. Effros and Uffe Haagerup, Lifting problems and local reflexivity for C\*algebras, Duke Mathematical Journal 52 (1985), no. 1, 103–128.
- [24] Edward G. Effros, Narutaka Ozawa, and Zhong-Jin Ruan, On injectivity and nuclearity for operator spaces, Duke Mathematical Journal 110 (2001), no. 3, 489–521.
- [25] Edward G. Effros and Zhong-Jin Ruan, OL<sub>p</sub> spaces, Operator algebras and operator theory (Shanghai, 1997), Contemp. Math., vol. 228, American Mathematica Society, Providence, RI, 1998, pp. 51–77.
- [26] \_\_\_\_\_, Operator spaces, London Mathematical Society Monographs. New Series, vol. 23, The Clarendon Press, Oxford University Press, New York, 2000.
- [27] George A Elliott, On the classification of inductive limits of sequences of semisimple finite-dimensional algebras, Journal of Algebra 38 (1976), no. 1, 29–44.
- [28] George A. Elliott, Some C\*-algebras with outer derivations, III, Annals of Mathematics 106 (1977), no. 1, 121–143.
- [29] \_\_\_\_\_, Towards a theory of classification, Advances in Mathematics 223 (2010), no. 1, 30–48.
- [30] George A. Elliott, Ilijas Farah, Vern Paulsen, Christian Rosendal, Andrew S. Toms, and Asger Törnquist, *The isomorphism relation for separable C\*-algebras*, Mathematical Research Letters **20** (2013), no. 6, 1071–1080.
- [31] Ruy Exel, Inverse semigroups and combinatorial C\*-algebras, Bulletin of the Brazilian Mathematical Society, New Series 39 (2008), no. 2, 191–313.
- [32] Ilijas Farah, All automorphisms of the Calkin algebra are inner, Annals of Mathematics 173 (2011), no. 2, 619–661.
- [33] \_\_\_\_\_, A dichotomy for the Mackey Borel structure, Proceedings of the 11th Asian Logic Conference, World Sci. Publ., Hackensack, NJ, 2012, p. 86–93.

- [34] \_\_\_\_\_, Logic and operator algebras, Proceedings of the International Congress of Mathematicians (Seoul, South Corea), 2014.
- [35] Ilijas Farah, Bradd Hart, and David Sherman, Model theory of operator algebras II: Model theory, Israel Journal of Mathematics, to appear.
- [36] Ilijas Farah, Dan Hathaway, Takeshi Katsura, and Aaron Tikuisis, A simple C\*algebra with finite nuclear dimension which is not Z-stable, Münster Journal of Mathematics, to appear.
- [37] Ilijas Farah, Andrew Toms, and Asger Törnquist, *The descriptive set theory of C\*-algebra invariants*, International Mathematics Research Notices (2012), 5196–5226.
- [38] Ilijas Farah, Andrew S. Toms, and Asger Törnquist, Turbulence, orbit equivalence, and the classification of nuclear C\*-algebras, Journal für die reine und angewandte Mathematik 688 (2014), 101–146.
- [39] J. M. G. Fell, The structure of algebras of operator fields, Acta Mathematica 106 (1961), 233–280.
- [40] Valentin Ferenczi, Alain Louveau, and Christian Rosendal, The complexity of classifying separable Banach spaces up to isomorphism, Journal of the London Mathematical Society 79 (2009), no. 2, 323–345.
- [41] Matthew Foreman and Benjamin Weiss, An anti-classification theorem for ergodic measure preserving transformations, Journal of the European Mathematical Society 6 (2004), no. 3, 277–292.
- [42] Roland Fraïssé, Sur l'extension aux relations de quelques propriétés des ordres, Annales Scientifiques de l'École Normale Supérieure. Troisième Série 71 (1954), 363–388.
- [43] Harvey Friedman and Lee Stanley, A Borel reductibility theory for classes of countable structures, Journal of Symbolic Logic 54 (1989), no. 03, 894–914.
- [44] Su Gao, Invariant descriptive set theory, Pure and Applied Mathematics (Boca Raton), vol. 293, CRC Press, Boca Raton, FL, 2009.
- [45] Su Gao and Alexander S. Kechris, On the classification of Polish metric spaces up to isometry, Memoirs of the American Mathematical Society 161 (2003), no. 766.

- [46] Joanna Garbulińska-Węgrzyn and Wiesław Kubiś, A universal operator on the Gurarii space, Journal of Operator Theory, to appear.
- [47] Eusebio Gardella and Martino Lupini, Conjugacy and cocycle conjugacy of automorphisms of  $\mathcal{O}_2$  are not Borel, arXiv:1404.3617 (2014).
- [48] Izrail M. Gelfand and Mark A. Neumark, On the imbedding of normed rings into the ring of operators in Hilbert space, Matematicheskii Sbornik Novaya Seriya 12(54) (1943), 197–213.
- [49] James Glimm, Type I C\*-algebras, Annals of Mathematics. Second Series 73 (1961), 572–612.
- [50] James G. Glimm, On a certain class of operator algebras, Transactions of the American Mathematical Society 95 (1960), no. 2, 318–340.
- [51] Isaac Goldbring and Martino Lupini, Model-theoretic aspects of the Gurarij operator space, arXiv:1501.04332 (2015).
- [52] Isaac Goldbring and Thomas Sinclair, On Kirchberg's embedding problem, Journal of Functional Analysis, to appear.
- [53] Vladimir I. Gurariĭ, Spaces of universal placement, isotropic spaces and a problem of Mazur on rotations of Banach spaces, Siberian Mathematical Journal 7 (1966), 1002–1013.
- [54] Masamichi Hamana, Injective envelopes of C\*-algebras, Journal of the Mathematical Society of Japan 31 (1979), no. 1, 181–197.
- [55] \_\_\_\_\_, Injective envelopes of operator systems, Publications of the Research Institute for Mathematical Sciences 15 (1979), no. 3, 773–785.
- [56] \_\_\_\_\_, Triple envelopes and Šilov boundaries of operator spaces, Mathematics Journal of Toyama University 22 (1999), 77–93.
- [57] Kyung Hoon Han and Vern I. Paulsen, An approximation theorem for nuclear operator systems, Journal of Functional Analysis 261 (2011), no. 4, 999–1009.
- [58] Greg Hjorth, Non-smooth infinite-dimensional group representations, Manuscript, 1997.

- [59] \_\_\_\_\_, Classification and orbit equivalence relations, Mathematical Surveys and Monographs, vol. 75, American Mathematical Society, Providence, RI, 2000.
- [60] Greg Hjorth and Alexander S. Kechris, Borel equivalence relations and classifications of countable models, Annals of Pure and Applied Logic 82 (1996), no. 3, 221–272.
- [61] Masaki Izumi, Finite group actions on C\*-algebras with the Rohlin property, I, Duke Mathematical Journal 122 (2004), no. 2, 233–280.
- [62] \_\_\_\_\_, Finite group actions on C\*-algebras with the rohlin property—II, Advances in Mathematics **184** (2004), no. 1, 119–160.
- [63] Stephen C. Jackson, Alexander S. Kechris, and Alain Louveau, Countable Borel equivalence relations, J. Math. Logic 2 (2002), 1–80.
- [64] David L. Johnson, Presentations of groups, second ed., London Mathematical Society Student Texts, vol. 15, Cambridge University Press, Cambridge, 1997.
- [65] Marius Junge, Narutaka Ozawa, and Zhong-Jin Ruan, On  $\mathcal{OL}_{\infty}$  structures of nuclear  $C^*$ -algebras, Mathematische Annalen **325** (2003), no. 3, 449–483.
- [66] Marius Junge and Gilles Pisier, Bilinear forms on exact operator spaces and  $B(H) \otimes B(H)$ , Geometric and Functional Analysis 5 (1995), no. 2, 329–363.
- [67] Richard V. Kadison, E. Christopher Lance, and John R. Ringrose, Derivations and automorphisms of operator algebras. II, Journal of Functional Analysis (1967), 204–221.
- [68] Takeshi Katsura and Hiroki Matui, Classification of uniformly outer actions of on UHF algebras, Advances in Mathematics 218 (2008), no. 3, 940–968.
- [69] Manmohan Kaur and Zhong-Jin Ruan, Local properties of ternary rings of operators and their linking C\*-algebras, Journal of Functional Analysis 195 (2002), no. 2, 262– 305.
- [70] Ali S. Kavruk, Vern I. Paulsen, Ivan G. Todorov, and Mark Tomforde, Quotients, exactness, and nuclearity in the operator system category, Advances in Mathematics 235 (2013), 321–360.
- [71] Alexander Kechris and Alain Louveau, The classification of hypersmooth Borel equivalence relations, Journal of the American Mathematical Society 10 (1997), no. 1, 215–242.

- [72] Alexander S. Kechris, Countable sections for locally compact group actions, Ergodic Theory and Dynamical Systems 12 (1992), no. 2, 283–295.
- [73] \_\_\_\_\_, Classical descriptive set theory, Graduate Texts in Mathematics, vol. 156, Springer-Verlag, New York, 1995.
- [74] Alexander S. Kechris and Nikolaos E. Sofronidis, A strong generic ergodicity property of unitary and self-adjoint operators, Ergodic Theory and Dynamical Systems 21 (2001), no. 5, 1459–1479.
- [75] David Kerr, Hanfeng Li, and Mikaël Pichot, Turbulence, representations, and tracepreserving actions, Proceedings of the London Mathematical Society 100 (2010), no. 2, 459–484.
- [76] David Kerr, Martino Lupini, and N. Christopher Phillips, Borel complexity and automorphisms of C\*-algebras, accepted for publication by the Journal of Functional Analysis.
- [77] E. Kirchberg, On subalgebras of the Car-algebra, Journal of Functional Analysis 129 (1995), no. 1, 35–63.
- [78] Eberhard Kirchberg and N. Christopher Phillips, Embedding of exact C\*-algebras in the Cuntz algebra O<sub>2</sub>, Journal f
  ür die reine und angewandte Mathematik 525 (2000), 17–53.
- [79] Eberhard Kirchberg and Simon Wassermann, C\*-algebras generated by operator systems, Journal of Functional Analysis 155 (1998), no. 2, 324–351.
- [80] Akitaka Kishimoto, Outer automorphisms and reduced crossed products of simple C\*algebras, Communications in Mathematical Physics 81 (1981), no. 3, 429–435.
- [81] \_\_\_\_\_, The Rohlin property for automorphisms of UHF algebras, Journal für die reine und angewandte Mathematik **1995** (1995), no. 465, 183–196.
- [82] Wiesław Kubiś, Fraïssé sequences: category-theoretic approach to universal homogeneous structures, Annals of Pure and Applied Logic 165 (2014), no. 11, 1755–1811.
- [83] Wiesław Kubiś and Sławomir Solecki, A proof of uniqueness of the Gurariĭ space, Israel Journal of Mathematics 195 (2013), no. 1, 449–456.

- [84] Alex Kumjian, David Pask, Iain Raeburn, and Jean Renault, Graphs, groupoids, and cuntz-krieger algebras, Journal of Functional Analysis 144 (1997), no. 2, 505–541.
- [85] E. C. Lance, *Hilbert C\*-modules*, London Mathematical Society Lecture Note Series, vol. 210, Cambridge University Press, Cambridge, 1995.
- [86] Franz Lehner,  $M_n$ -espaces, sommes d'unitaires et analyse harmonique sur le groupe libre, Ph.D. thesis, Université de Paris 6, 1997.
- [87] Joram Lindenstrauss and Aleksander Pełczyński, Absolutely summing operators in  $L_p$ -spaces and their applications, Studia Mathematica **29** (1968), 275–326.
- [88] Terry A. Loring, Lifting solutions to perturbing problems in C\*-algebras, Fields Institute Monographs, vol. 8, American Mathematical Society, Providence, RI, 1997.
- [89] Martino Lupini, Polish groupoids and functorial complexity, arXiv:1407.6671 (2014).
- [90] \_\_\_\_\_, Uniqueness, universality, and homogeneity of the noncommutative Gurarij space, arXiv:1410.3345 (2014).
- [91] \_\_\_\_\_, Unitary equivalence of automorphisms of separable C\*-algebras, Advances in Mathematics 262 (2014), 1002–1034.
- [92] \_\_\_\_\_, A universal nuclear operator system, arXiv:1412.0281 (2014).
- [93] Wolfgang Lusky, The Gurarij spaces are unique, Archiv der Mathematik 27 (1976), no. 6, 627–635.
- [94] George W. Mackey, Borel structure in groups and their duals, Transactions of the American Mathematical Society 85 (1957), no. 1, 134–165.
- [95] \_\_\_\_\_, Ergodic theory, group theory, and differential geometry, Proceedings of the National Academy of Sciences of the United States of America 50 (1963), 1184—1191.
- [96] Hiroki Matui, Classification of outer actions of  $\mathbb{Z}^N$  on  $\mathcal{O}_2$ , Advances in Mathematics **217** (2008), no. 6, 2872–2896.
- [97] \_\_\_\_\_, Z<sup>N</sup>-actions on UHF algebras of infinite type, Journal fr die Reine und Angewandte Mathematik. 657 (2011), 225–244. MR 2824789
- [98] Hiroki Matui and Yasuhiko Sato, Z-stability of crossed products by strongly outer actions, Communications in Mathematical Physics 314 (2012), no. 1, 193–228.

- [99] Julien Melleray, Computing the complexity of the relation of isometry between separable Banach spaces, Mathematical Logic Quarterly 53 (2007), no. 2, 128–131.
- [100] \_\_\_\_\_, On the geometry of Urysohn's universal metric space, Topology and its Applications 154 (2007), no. 2, 384–403.
- [101] Julien Melleray and Todor Tsankov, Generic representations of abelian groups and extreme amenability, Israel Journal of Mathematics 198 (2013), no. 1, 129–167.
- [102] \_\_\_\_\_, Extremely amenable groups via continuous logic, arXiv:1404.4590 (2014), arXiv: 1404.4590.
- [103] Hideki Nakamura, Aperiodic automorphisms of nuclear purely infinite simple C\*algebras, Ergodic Theory and Dynamical Systems 20 (2000), no. 06, 1749–1765.
- [104] Timur Oikhberg, The non-commutative Gurarii space, Archiv der Mathematik 86 (2006), no. 4, 356–364.
- [105] Timur Oikhberg and Éric Ricard, Operator spaces with few completely bounded maps, Mathematische Annalen 328 (2004), no. 1-2, 229–259.
- [106] Alan L. T. Paterson, Groupoids, inverse semigroups, and their operator algebras, Progress in Mathematics, vol. 170, Birkhäuser, Boston, MA, 1999.
- [107] Vern Paulsen, Completely bounded maps and operator algebras, Cambridge Studies in Advanced Mathematics, vol. 78, Cambridge University Press, Cambridge, 2002.
- [108] Vern I. Paulsen, Completely bounded maps on C\*-algebras and invariant operator ranges, Proceedings of the American Mathematical Society 86 (1982), no. 1, 91–96.
- [109] Gert K. Pedersen, C\*-algebras and their automorphism groups, London Mathematical Society Monographs, vol. 14, Academic Press Inc., London, 1979.
- [110] \_\_\_\_\_, Analysis now, Graduate Texts in Mathematics, vol. 118, Springer-Verlag, New York, 1989.
- [111] John Phillips, Outer automorphisms of separable C\*-algebras, Journal of Functional Analysis 70 (1987), no. 1, 111–116.
- [112] \_\_\_\_\_, Central sequences and automorphisms of C\*-algebras, American Journal of Mathematics 110 (1988), no. 6, 1095–1117.

- [113] John Phillips and Iain Raeburn, Automorphisms of C\*-algebras and second Čech cohomology, Indiana University Mathematics Journal 29 (1980), no. 6, 799–822.
- [114] Gilles Pisier, Exact operator spaces, Astérisque (1995), no. 232, 159–186, Recent advances in operator algebras (Orléans, 1992).
- [115] \_\_\_\_\_, Introduction to operator space theory, London Mathematical Society Lecture Note Series, vol. 294, Cambridge University Press, Cambridge, 2003.
- [116] Iain Raeburn and Jonathan Rosenberg, Crossed products of continuous-trace C\*algebras by smooth actions, Transactions of the American Mathematical Society 305 (1988), no. 1, 1–45.
- [117] Iain Raeburn and Dana P. Williams, Morita equivalence and continuous-trace C\*algebras, Mathematical Surveys and Monographs, vol. 60, American Mathematical Society, Providence, RI, 1998.
- [118] Arlan Ramsay, Virtual groups and group actions, Advances in Mathematics 6 (1971), no. 3, 253–322.
- [119] \_\_\_\_\_, The Mackey-Glimm dichotomy for foliations and other Polish groupoids, Journal of Functional Analysis 94 (1990), no. 2, 358–374.
- [120] Arlan B. Ramsay, *Polish groupoids*, Descriptive set theory and dynamical systems, London Math. Soc. Lecture Note Ser., vol. 277, Cambridge University Press, Cambridge, 2000, p. 259–271.
- [121] Pedro Resende, Lectures on étale groupoids, inverse semigroups and quantales, (2006).
- [122] Mikael Rørdam, A simple C\*-algebra with a finite and an infinite projection, Acta Mathematica 191 (2003), no. 1, 109–142.
- [123] Zhong-Jin Ruan, Subspaces of C\*-algebras, Journal of Functional Analysis 76 (1988), no. 1, 217–230.
- [124] \_\_\_\_\_, Injectivity of operator spaces, Transactions of the American Mathematical Society **315** (1989), no. 1, 89–104.
- [125] Marcin Sabok, Completeness of the isomorphism problem for separable C\*-algebras, arXiv:1306.1049 (2013).

- [126] Ramez L. Sami, Polish group actions and the Vaught conjecture, Transactions of the American Mathematical Society 341 (1994), no. 1, 335–353.
- [127] Román Sasyk and Asger Törnquist, Turbulence and Araki–Woods factors, Journal of Functional Analysis 259 (2010), no. 9, 2238–2252.
- [128] Konstantinos Schoretsanitis, Fraïssé theory for metric structures, Ph.D. thesis, University of Illinois at Urbana-Champaign, 2007.
- [129] Roger R. Smith, Completely bounded module maps and the Haagerup tensor product, Journal of Functional Analysis 102 (1991), no. 1, 156–175.
- [130] \_\_\_\_\_, Finite dimensional injective operator spaces, Proceedings of the American Mathematical Society 128 (2000), no. 11, 3461–3462.
- [131] Andrew Toms, On the independence of K-theory and stable rank for simple C\*algebras, Journal fur die reine und angewandte Mathematik 578 (2005), 185–199.
- [132] \_\_\_\_\_, On the classification problem for nuclear C\*-algebras, Annals of Mathematics 167 (2008), no. 3, 1029–1044.
- [133] Vladimir V. Uspenskij, On the group of isometries of the Urysohn universal metric space, Commentationes Mathematicae Universitatis Carolinae 31 (1990), no. 1, 181–182.
- [134] Robert Vaught, Invariant sets in topology and logic, Fundamenta Mathematicae 82 (1974), 269–294.
- [135] Gerd Wittstock, Ein operatorwertiger Hahn-Banach Satz, Journal of Functional Analysis 40 (1981), no. 2, 127–150.
- [136] Blerina Xhabli, The super operator system structures and their applications in quantum entanglement theory, Journal of Functional Analysis 262 (2012), no. 4, 1466– 1497.
- [137] Joseph Zielinski, The complexity of the homeomorphism relation between compact metric spaces, arXiv:1409.5523 (2014).